

Modular Multilevel Converters in Hybrid Multi-Terminal HVDC Systems

Author: Sun, Pingyang

Publication Date: 2021

DOI: https://doi.org/10.26190/unsworks/22715

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/71097 in https:// unsworks.unsw.edu.au on 2024-04-30

Modular Multilevel Converters in Hybrid Multi-Terminal HVDC Systems

by Pingyang Sun



A thesis submitted in fulfilment of the requirements for the degree of

Master of Philosophy

School of Electrical Engineering and Telecommunications

The University of New South Wales

© Pingyang Sun

June 2021 Sydney, NSW, Australia



Thesis/Dissertation Sheet

Surname/Family Name	:	Sun
Given Name/s	:	Pingyang
Abbreviation for degree as give in the University calendar	:	MPhil
Faculty	:	Engineering
School	:	Electrical Engineering and Telecommunications
Thesis Title	:	Modular Multilevel Converters in Hybrid Multi-Terminal HVDC System

Abstract 350 words maximum: (PLEASE TYPE)

High-voltage direct current (HVDC) systems are becoming commonplace in modern power systems. Line commutated converters (LCCs) are suitable for bulk power and ultra-HVDC (UHVDC) transmission, while with inflexible power reversal capability and possible commutation failures. However, voltage source converters (VSCs) possess flexible power reversal capability and provide immunity to commutation failures. Modular VSC topologies offer improved performance compared to conventional 2 level/3 level VSC-based HVDC. The family of modular VSCs includes the well-established modular multilevel converter (MMC) and other emerging modular VSC topologies such as the DC-fault tolerant alternate arm converter (AAC) that share topological and operational similarities with the MMC. It is noteworthy that the integration of LCC and modular VSCs leads to unique benefits despite the challenges of different HVDC configurations. Hence, it is necessary to explore the system performance of different HVDC converter topologies, especially more complex hybrid multiterminal HVDC (MTDC) systems and DC-grids combining different converters. This thesis focuses on the combination of the LCC, MMC and AAC to constitute different hybrid HVDC transmission systems.

It is of significance to provide a common platform where the proper comparison and evaluation of different HVDC systems and control methods can be completed and independently validated. Therefore, this thesis also provides an overview of current HVDC benchmark models available in the existing literature. In addition, the detailed modeling methods of HVDC systems are discussed in this thesis. For ensuring the static security of HVDC systems especially the future DC-grids, this thesis proposes a generalized expression of DC power flow under mixed power/voltage (P/V) and current/voltage (I/V) droop control, considering the DC power flow for normal operation and after converter outage.

Detailed simulation models are established in PLECS-Blockset and Simulink to study the hybrid HVDC/MTDC systems and DC grid combining the LCC with the MMC and (or) AAC. The detailed sets of results demonstrate the functionalities of developed hybrid HVDC systems and validate the performance of systems complying with widely accepted HVDC operating standards. The developed LCC/AAC-based HVDC/MTDC systems and LCC/MMC/AAC-based DC grid in this thesis are prime steps towards the study of more complex MTDC systems and a key element in the development of future DC super grids.

Declaration relating to disposition of project thesis/dissertation

I hereby grant to the University of New South Wales or its agents a non-exclusive licence to archive and to make available (including to members of the public) my thesis or dissertation in whole or in part in the University libraries in all forms of media, now or here after known. I acknowledge that I retain all intellectual property rights which subsist in my thesis or dissertation, such as copyright and patent rights, subject to applicable law. I also retain the right to use all or part of my thesis or dissertation in future works (such as articles or books).

22/09/2021

Signature

Date

The University recognises that there may be exceptional circumstances requiring restrictions on copying or conditions on use. Requests for restriction for a period of up to 2 years can be made when submitting the final copies of your thesis to the UNSW Library. Requests for a longer period of restriction may be considered in exceptional circumstances and require the approval of the Dean of Graduate Research.

ORIGINALITY STATEMENT

'I hereby declare that this submission is my own work and to the best of my knowledge it contains no materials previously published or written by another person, or substantial proportions of material which have been accepted for the award of any other degree or diploma at UNSW or any other educational institution, except where due acknowledgement is made in the thesis. Any contribution made to the research by others, with whom I have worked at UNSW or elsewhere, is explicitly acknowledged in the thesis. I also declare that the intellectual content of this thesis is the product of my own work, except to the extent that assistance from others in the project's design and conception or in style, presentation and linguistic expression is acknowledged.'

Signed

22/09/2021 Date

COPYRIGHT STATEMENT

'I hereby grant the University of New South Wales or its agents a non-exclusive licence to archive and to make available (including to members of the public) my thesis or dissertation in whole or part in the University libraries in all forms of media, now or here after known. I acknowledge that I retain all intellectual property rights which subsist in my thesis or dissertation, such as copyright and patent rights, subject to applicable law. I also retain the right to use all or part of my thesis or dissertation in future works (such as articles or books).'

'For any substantial portions of copyright material used in this thesis, written permission for use has been obtained, or the copyright material is removed from the final public version of the thesis.'

Signed

22/09/2021

Date

AUTHENTICITY STATEMENT

'I certify that the Library deposit digital copy is a direct equivalent of the final officially approved version of my thesis.'

Signed

22/09/2021

Date



INCLUSION OF PUBLICATIONS STATEMENT

UNSW is supportive of candidates publishing their research results during their candidature as detailed in the UNSW Thesis Examination Procedure.

Publications can be used in their thesis in lieu of a Chapter if:

- The candidate contributed greater than 50% of the content in the publication and is the "primary author", ie. the candidate was responsible primarily for the planning, execution and preparation of the work for publication
- The candidate has approval to include the publication in their thesis in lieu of a Chapter from their supervisor and Postgraduate Coordinator.
- The publication is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in the thesis

Please indicate whether this thesis contains published material or not:



This thesis contains no publications, either published or submitted for publication *(if this box is checked, you may delete all the material on page 2)*



Some of the work described in this thesis has been published and it has been documented in the relevant Chapters with acknowledgement *(if this box is checked, you may delete all the material on page 2)*



This thesis has publications (either published or submitted for publication) incorporated into it in lieu of a chapter and the details are presented below

CANDIDATE'S DECLARATION

I declare that:

- I have complied with the UNSW Thesis Examination Procedure
- where I have used a publication in lieu of a Chapter, the listed publication(s) below meet(s) the requirements to be included in the thesis.

Candidate's Name	Signature	Date (dd/mm/yy)
Pingyang Sun		22/09/2021

Abstract

High-voltage direct current (HVDC) systems are becoming commonplace in modern power systems. On the one hand, thyristor-based line commutated converter (LCC) is a mature technology and suitable for bulk power and ultra-HVDC (UHVDC) transmission, while with inflexible power reversal capability and possible commutation failures. On the other hand, voltage source converters (VSCs) offer flexible power reversal capability and provide immunity to commutation failures. Hence, VSCs are more suitable for multiterminal HVDC (MTDC) systems and future DC grids. Moreover, hybrid MTDC systems and DC-grids combining LCCs and VSCs are options for delivering power from remote energy zones to main load centers or dealing with commutation failures by replacing LCCs in the inverter side with VSCs.

Modular VSC topologies offer improved performance compared to conventional two level or three level VSC-based HVDC. The family of modular VSCs includes the wellestablished modular multilevel converter (MMC) and other emerging modular VSC topologies that share topological and operational similarities with the MMC such as the DC-fault tolerant alternate arm converter (AAC). Although modular VSCs offer unique benefits over LCCs and conventional VSCs, there are still challenges that need to be further addressed, specifically in modular VSCs other than the MMC. It is noteworthy that the integration of LCC and modular VSCs leads to unique benefits despite the challenges of different HVDC configurations. Thus, it is necessary to explore the system performance of different HVDC converter topologies, especially more complex hybrid MTDC systems and DC-grids combining different converters. This thesis focuses on the combination of the LCC, MMC and AAC to constitute different hybrid HVDC transmission systems and hybrid DC grids.

The advantages of HVDC transmission in the current electricity grid environment have sparked an increased interest on HVDC system studies. A common challenge is the proper comparison and evaluation of different systems and control methods. Benchmark models facilitate this approach as they provide a common platform where such comparisons can be completed and independently validated. This thesis also provides an overview of current HVDC benchmark models available in the existing literature. The analysis and discussion based on the review identifies the current research gaps and opportunities for development of HVDC benchmark models, the importance of openly available data across all benchmark models covering a wide range of simulations and applications as well as the potential for development of extended models to facilitate large-scale network analysis and DC grids. Different modeling approaches can be used in HVDC systems for specific studies. The detailed modeling approaches of HVDC systems are discussed in this thesis, including the electromagnetic transient (EMT), electromechanical transient, dynamic frequency capturing and co-simulation modeling methods. For ensuring the static security of HVDC systems especially that of future hybrid DC grids, this thesis proposes a generalized expression of DC power flow under mixed power/voltage (P/V) and current/voltage (I/V) droop control. The initial DC power flow for normal operation and the DC power flow after converter outage are discussed considering the maximum power limitation of converters. Also, the accuracy of proposed generalized expression of DC power flow is verified in an MMC-based DC grid and developed hybrid multi-converter DC grid.

Detailed simulation models are established in PLECS-Blockset and Simulink to study the hybrid point-to-point (PTP), multi-terminal HVDC systems and DC grid combining the LCC with the MMC and (or) AAC. A control hierarchy is developed and detailed control schemes under AC and DC faults are proposed for hybrid HVDC systems.

The detailed sets of results, including the steady-state operation, reference tracking and system performance under AC and DC faults, demonstrate the functionalities of developed hybrid HVDC systems and validate the performance of systems complying with widely accepted HVDC operating standards. The developed LCC and AAC-based PTP HVDC system, MTDC system and LCC, MMC and AAC-based DC grid in this thesis are prime steps towards the study of more complex MTDC systems and a key element in the development of future DC super grids.

Acknowledgements

Upon the completion of this thesis, I would like to express my sincere gratitude to my supervisor, Dr. Georgios Konstantinou and co-supervisor, Dr. Harith Wickramasinghe for their continuous support, enthusiastic encouragement and useful critiques of my master study and research. They still provide patient guidance and share their immense knowledge via online meetings, although the ongoing relentless pandemic from 2020 leads to work inconvenience.

Besides my supervisors, my sincere thanks also goes to all my colleagues, Felipe Arraño-Vargas, Yumeng Tian, Zhiwei Shen and Xinyuan Liu, in UNSW Sydney and all my friends in Sydney. I will always value these enjoyable moments with you.

Last but not the least, I am extremely grateful to my parents for giving birth to me, educating and preparing me for my future. I also wish to thank them for their spiritual and financial support throughout my study.

Contents

Ał	ostrac	zt	iv
Ac	knov	vledgements	vi
Li	st of l	Figures	xii
Li	st of '	Tables	xv
Li	st of A	Abbreviations x	vii
1	Intr	oduction	1
	 1.1 1.2 1.3 1.4 1.5 1.6 	Line-Commutated Converter (LCC) and Voltage Source Converter (VSC)- based Hybrid High-Voltage Direct Current (HVDC) SystemsLiterature Review1.2.1Background1.2.2LCC and Conventional VSC-based Hybrid HVDC Systems1.2.3LCC and Modular Multilevel Converter (MMC)-based Hybrid Point- to-Point (PTP) HVDC Systems1.2.4LCC and MMC-based Hybrid Multiterminal HVDC (MTDC) SystemsThesis ObjectivesSimulation Models and ToolsList of Publications1.6.1Journal Papers	1 7 8 9 12 15 15 15 16 17 17
2	1.7 Ben	1.6.2 International Conference Papers 1.6.2 International Conference Papers 1.6.3 Papers Under Revision or Review 1.6.4 Other Publications 1.6.4 Other Publications 1.6.4 Other Publications Thesis Outline 1.6.4 Other Publications chmark Models for HVDC Converters, HVDC Systems and DC Grid Studies	17 18 18 19 21
	2.1 2.2	Introduction	21 22 22 22

		2.2.3	HVDC Systems	3
	2.3	Types	of HVDC Computational Models	4
	2.4	Review	v of HVDC Benchmark Models	5
	2.5	Discus	sion and Conclusion	9
		2.5.1	Research Gaps in Benchmark Models	9
		2.5.2	Data Availability	9
		2.5.3	Expansion of Types and Complex Models	0
3	Торо	ologies	for Hybrid HVDC Systems 33	2
	3.1	Introd	uction	2
	3.2	Line-C	ommutated Converter (LCC)	2
		3.2.1	Basic Topology and Operating Principles of the LCC	3
		3.2.2	Control Schemes of the LCC	5
			3.2.2.1 Conventional Control Schemes	5
			3.2.2.2 Improved Control Schemes	6
		3.2.3	AC and DC Filters of the LCC	7
	3.3	Modul	ar Multilevel Converter (MMC)	9
		3.3.1	Basic Topology and Operating Principles of the MMC	9
		3.3.2	Control Schemes of the MMC	2
			3.3.2.1 Control Hierarchy	2
			3.3.2.2 Modulation Techniques	3
			3.3.2.3 SM Capacitor Voltage Balancing 43	3
			3.3.2.4 Circulating Current Control	4
	3.4	Alterna	ate Arm Converter (AAC)	7
		3.4.1	Basic Topology and Operating Principles of the AAC	7
		3.4.2	Control Scheme of the AAC	9
			3.4.2.1 Control Hierarchy	9
			3.4.2.2 Energy Balancing Strategy 50	0
			3.4.2.3 Modulation and Direct Switch (DS) Control	0
			3.4.2.4 Overlap Period Control	1
			3.4.2.5 Circulating Current Control	2
			3.4.2.6 Zero-Current Switch (ZCS) of the DSs	2
		3.4.3	Extended Overlap AAC (EO-AAC)	3
	3.5	Conclu	ision	4
4	Mod	leling A	approaches and Power Flow Analysis of HVDC systems 55	5
	4.1	Introd	uction	5
	4.2	Model	ing Approaches for HVDC Systems	5
		4.2.1	Electromagnetic Transient (EMT) Modeling 50	6
			4.2.1.1 Analysis Methods in EMT Programs	6
			4.2.1.2 Numerical Integration Theory	8
			4.2.1.3 Parallel Computation Method	2
		4.2.2	Electromechanical Transient Modeling 64	4
		4.2.3	Other Modeling Approaches and Co-Simulation Studies 65	5

			4.2.2.1 Dynamia Fraguancy Conturing Modeling 66	
			4.2.3.1 Dynamic Frequency Capturing Modeling	
	12	Dotail	4.2.3.2 CO-simulation Studies	
	4.5		Pasic Theory 70	
		4.3.1	Static Characteristic 70	
	11	Gener	plized Expression of DC Power Flow 72	
	4.4		Initial DC Power Flow for Normal Operation 72	
		4.4.1	A A 1 1 Single DAV or LAV Droop Control 72	
			4.4.1.1 Single P/V of I/V Droop Control	
		449	4.4.1.2 Mixed P/V and I/V Droop Control	
		4.4.4	4.4.2.1 Single D/V or L/V Dreep Control	
			4.4.2.1 Single P/V of I/V Droop Control	,
		112	4.4.2.2 Mixed P/V and I/V Droop Control	
		4.4.3	4.4.2.1 Varification in MMC based DC Crid under Single D/UDroop	
			4.4.3.1 Verification in MMC-based DC Grid under Single P/ v Droop	
			4432 Verification in MMC-based DC Grid under Mixed Droop	
			Control	
	4.5	Conclu	1sion 86	
	110	Gomene		
5	LCC	and A/	AC-Based Hybrid PTP HVDC Transmission System 87	
	5.1	Introd	uction	
	5.2	Conve	rter Description and Control Hierarchy	
		5.2.1	Converter Description	
		5.2.2	Control Hierarchy	
	5.3	System	n Topology and Control Scheme	
		5.3.1	System Topology	
		5.3.2	Control Scheme	
			5.3.2.1 Control Scheme under Normal Operation 90	
			5.3.2.2 Control Scheme under AC and DC Faults 91	
	5.4	System	n Parameter Selection	
		5.4.1	Main Parameters of the LCC	
		5.4.2	Main Parameters of the AAC	
		5.4.3	DC Transmission Line and Control Parameters	
		5.4.4	AC and DC Link Filters	
		5.4.5	Comparison between the MMC and the AAC	
	5.5	Simula	ation Results)
		5.5.1	Steady-State Operation	
		5.5.2	Reference Tracking / Steady-State	5
			5.5.2.1 DC Current Reference Change	5
			5.5.2.2 Reactive Power Reference Change	ŀ
		5.5.3	Performance under AC Faults)
			5.5.3.1 AC Faults at the AC-Grid of the LCC)
			5.5.3.2 AC Faults at the AC-Grid of the AAC)
		5.5.4	Permanent Pole-to-Pole DC Fault	'

	5.6	Conclu	ision	111
6	LCC	and A/	AC-Based Hybrid MTDC Transmission System	113
	6.1	Introd	uction	113
	6.2	Detaile	ed System Description	113
		6.2.1	System Topology	113
		6.2.2	System Parameter Selection	116
			6.2.2.1 DC Output Voltage of the LCC	116
			6.2.2.2 Power Calculation of the LCC	116
			6.2.2.3 Operating Point and Overlap Period of the AAC	116
			6.2.2.4 Number of SMs and Switching Devices of the AAC	117
			6.2.2.5 SM Capacitance of the AAC	117
			6.2.2.6 Filters and DC Transmission Line	117
	6.3	P/V D	roop Control and DC Fault-Ride-Through (FRT) Scheme	118
		6.3.1	P/V Droop Control	118
		6.3.2	DC FRT Scheme	119
	6.4	Simula	ation Results	120
		6.4.1	Steady-State Operation	120
		6.4.2	Reference Tracking	121
			6.4.2.1 Power Adjustment of the LCC	121
			6.4.2.2 Active Power Reference Change of AAC2	123
			6.4.2.3 Reactive Power Reference Change of both AACs	123
		6.4.3	System Performance under AC Faults	125
			6.4.3.1 AC Faults at Rectifier Side (LCC)	125
			6.4.3.2 AC Faults at Inverter Side (AAC1)	125
		6.4.4	Temporary Pole-to-Pole DC Fault	127
	6.5	Conclu	ision	130
7	Deve	elopme	ent of Hybrid Multi-Converter DC Grids	131
	7.1	Introd	uction	131
	7.2	Config	uration of Hybrid Multi-Converter DC Grid	131
		7.2.1	Description of the LCC	132
		7.2.2	Description of the MMC and the AAC	133
	7.3	AC and	d DC Fault Handling Scheme	133
		7.3.1	AC Fault Handling Scheme	134
		7.3.2	DC Fault Handling Scheme	134
	7.4	DC Por	wer Flow in the Hybrid Multi-Converter DC Grid	135
		7.4.1	DC Power Flow under Normal Operation	136
		7.4.2	DC Power Flow after MMC Outage	137
	7.5	Simula	ation Results	138
		7.5.1	Power Flow Verification	138
		7.5.2	AC FRT Capability	140
		7.5.3	DC FRT Capability	141
		7.5.4	Discussion	144

7.6 Conclusion	144
8 Conclusions and Future Work 8.1 Conclusions	145 145
Bibliography	14/ 148

List of Figures

1.1	LCC-based HVDC station made by Siemens.	2
1.2	Conventional VSC-based HVDC station made by ABB	3
1.3	Modular VSC-based HVDC station made by Siemens	3
1.4	Changji-Guquan ± 1100 kV LCC-based UHVDC transmission system	4
1.5	Zhangbei meshed four-terminal DC grid	5
1.6	Hybrid HVDC PTP system.	7
1.7	Hybrid MTDC system.	7
1.8	Available configurations of hybrid HVDC systems	10
1.9	One-line diagram of the Skagerrak hybrid HVDC system.	10
1.10	Configurations of hybrid UHVDC systems.	11
1.11	One-line diagram of the Baihetan hybrid HVDC systems.	12
1.12	Available configurations of Hybrid MTDC systems.	13
1.13	One-line diagram of the Kunliulong hybrid MTDC project.	13
1.14	Operation modes of the three-termianl hybrid UHVDC system	14
21	HVDC converter topologies	J J
2.1	HVDC station configurations	20 22
<u>ح.ح</u> ງງ	HVDC station configurations	23 24
⊿.ວ ງ_∕	CICDE P4 72 comprehensive DC grid henchmark model	24
2.4	CIGKE D4-72 comprehensive DC grid benchmark model	20
3.1	Converter topology of the LCC.	33
3.2	Conventional control schemes of the LCC.	36
3.3	Improved control schemes of the LCC.	37
3.4	AC and DC filters of the LCC.	38
3.5	Three-phase topology of the MMC.	40
3.6	Operation of the MMC arms.	41
3.7	Control hierarchy of the MMC.	42
3.8	Conventional SM capacitor voltage balancing algorithm.	44
3.9	Modified SM capacitor voltage balancing algorithm.	45
3.10	Circulating current controller.	46
3.11	Three-phase topology of the AAC.	47
3.12	Operation of the AAC arms.	48
3.13	Control hierarchy of the AAC.	49
3.14	Modulation and direct switch control of the AAC.	51
3.15	Overlap period configuration of the AAC	51

3.16	Extended overlap period configuration of the AAC.	53
4.1 4.2	Discrete companion model of R-L series branch and inductance branch Discrete companion model of capacitance branch	59 59
4.3	Parallel computation method of LCC and MMC.	63
4.4	Discrete model of HBSMs and equivalent arm structure	64
4.5	Equivalent function of DC systems in electromechanical modeling	65
4.6	Time-varing Fourier transform in DP modeling approach	66
4.7	Spectra of original signal and complex signal.	67
4.8	Spectra of complex signal under different transforms	68
4.9	Spectra of complex signal and complex envolope signal	68
4.10	Static characteristic of droop control in HVDC systems	71
4.11	Flowchart of initial DC power flow calculation.	74
4.12	Flowchart of DC power flow under converter outage	78
4.13	MMC-based four terminal DC grid.	79
4.14	Detailed current directions of MMC-based DC grid under single P/V droop control in the MMC-based DC grid	80
4.15	Detailed current directions of MMC-based DC grid under mixed droop control.	83
5.1	Control Hierarchy of the LCC and AAC-based hybrid PTP HVDC transmis-	
	sion system.	89
5.2	Topology of the LCC and AAC-based hybrid PTP HVDC transmission system.	90
5.3	Circuit of the LCC and AAC-based hybrid PTP HVDC transmission system under DC fault.	90
5.4	Control Structure of the LCC and AAC-based hybrid PTP HVDC transmission system.	91
5.5	Control scheme for riding through AC fault in the LCC and AAC-based hy- brid PTP HVDC transmission system	92
5.6	Control scheme for handling permanent DC fault in the LCC and AAC-based	2
	AC file on the LCC station	92
5./	AC Inters in the LCC station	97
5.0 5.0	Stondy state waveforms of the LCC side in the LCC and AAC based hybrid	97
5.9	DTD HVDC transmission system	101
5 10	Steady state waveforms of the AAC side in the ICC and AAC based hybrid	101
5.10	PTP HVDC transmission system	102
5 11	Harmonics of the LCC station in the LCC and AAC-based hybrid PTP HVDC	102
5.11	transmission system.	102
5.12	Harmonics of the AAC station in the LCC and AAC-based hybrid PTP HVDC	102
0.12	transmission system.	103
5.13	Capacitor voltages and arm currents of the AAC in the LCC and AAC-based	- 0
	hybrid PTP HVDC transmission system.	104
5.14	Performance under DC current and reactive power reference changes in the LCC and AAC-based hybrid PTP HVDC transmission system.	105

5.15 Single-lin the LCC a	ne to ground and short-circuit faults at the AC-grid of the LCC in and AAC-based hybrid PTP HVDC transimission system 10	07
5.16 Single-lin the LCC a	te to ground and short-circuit faults at the AC-grid of the AAC in and AAC-based hybrid PTP HVDC transimission system	08
5.17 DC fault system ba	at the midpoint of the line in the hybrid PTP HVDC transmission ased on the LCC and the AAC	10
5.18 DC fault system ba	at the midpoint of the line in the hybrid PTP HVDC transmission ased on the LCC and the HB-MMC.	11
6.1 Configura6.2 Control s	ation of the LCC and AAC-based hybrid MTDC transmission system. 11 structure of the LCC and AAC-based hybrid MTDC transmission	14
system. 6.3 DC fault	ride through scheme in the LCC and AAC-based hybrid MTDC	14 19
6.4 Capacitor	r voltage and arm current of AAC1 in the LCC and AAC-based hy-	
6.5 Capacitor	r voltage and arm current of AAC2 in the LCC and AAC-based hy-	20
brid MTE 6.6 Power ad	OC transmission system	21
in the LC 6.7 Reactive	C and AAC-based hybrid MTDC transmission system	22
hybrid M 6.8 Single lir	TDC transmission system. 12 ne to ground and three-phase short-circuit faults of the LCC in the	24
LCC and 6.9 Single lir	AAC-based hybrid MTDC transmission system	26
LCC and	AAC-based hybrid MTDC transmission system	28
transmiss	sion system	29
7.1 Topology	of the hybrid multi-converter DC grid	32
7.2 AC fault	ride through scheme in the hybrid multi-converter DC grid 13	34
7.3 DC fault	ride through scheme in the hybrid multi-converter DC grid. \ldots 13	35
7.4 Detailed	current directions of initial state (black) and after MMC outage	
(red) in t	he hybrid multi-converter DC grid	37
7.5 Steady-st C1 in the	ate and transient waveforms before and after MMC outage with hybrid multi-converter DC grid.	40
7.6 Temporal	ry three-phase short-circuit fault at MMC in the hybrid multi-converter	12
7.7 Tempora	ry pole to pole DC fault in the hybrid multi-converter DC grid 14	43

List of Tables

2.1	Summary of HVDC benchmark models	27
3.1 3.2	Basic control schemes of LCC	36 39
4.1	Parameters of four MMCs in the DC grid.	79
4.2 4.3	References of outer control loop for single P/V droop control in the MMC-	79
4.4	based DC grid	80
45	MMC-based DC grid	80
т.5	the MMC-based DC grid	81
4.6	Calculation of DC power flow for single P/V droop control after MMC4 outage in the MMC-based DC grid.	82
4.7	Simulation results of DC power flow for single P/V droop control after MMC4 outage in the MMC-based DC grid.	82
4.8	References of outer control loop for mixed droop control in the MMC-based	01
4.9	Calculation of initial DC power flow of mixed droop control in the MMC-	04
4.10	based DC grid	84
4 1 1	MMC-based DC grid	84
1.11	in the MMC-based DC grid.	85
4.12	outage in the MMC-based DC grid.	85
5.1	Rectifier parameters (LCC) in the LCC and AAC-based hybrid PTP HVDC	
5.2	Inverter parameters (AAC) in the LCC and AAC-based hybrid PTP HVDC	93
53	transmission system	94
-	hybrid PTP HVDC transmission system.	96
5.4	Detailed parameters of AC filters in the LCC station.	98
5.5	The limitation of DC-link filters in the LCC and AAC stations	99

5.6	Detailed parameters of DC-link filters in the LCC and AAC stations 99
5.7	transmission system.
6.1	Parameters of the LCC and AAC-based hybrid MTDC transmission system 115
6.2	Comparison of main modular VSCs in hybrid MTDC systems
6.3	Parameters of DC transmission line in the LCC and AAC-based hybrid MTDC
	transmission system
6.4	Control parameters in the LCC and AAC-based hybrid MTDC transmission
	system
6.5	DC Voltages and active power flow for the LCC and AAC-based hybrid
	MTDC transmission system
7.1	Parameters of DC transmission lines in the hybrid multi-converter DC grid 132
7.2	Parameters of the LCC in the hybrid multi-converter DC grid
7.3	Parameters of MMC and AACs in the hybrid multi-converter DC grid 133
7.4	References for high-level control in the hybrid multi-converter DC grid 136
7.5	Calculation of initial DC power flow for normal operation in the hybrid
	multi-converter DC grid
7.6	Calculation of DC power flow after MMC outage in the hybrid multi-converter
	DC grid
7.7	Simulation results of initial DC power flow for normal operation in the hy-
	brid multi-converter DC grid
7.8	Simulation results of DC power flow after MMC outage in the hybrid multi-
	converter DC grid

List of Abbreviations

Abbreviation	Definition
AAC	Alternate Arm Converter
AC	Alternating Current
AVM	Average Value Model
BTB	Back to Back
CB-PWM	Carrier-based Pulse Width Modulation
CCC	Capacitor Commutated Converter
CCSC	Circulating Current Suppression Control
CD-MMC	Clamped-Double Modular Multilevel Converter
CDSM	Clamped-Double Submodule
CSC	Current Source Converter
DAE	Differential-Algebraic Equation
DC	Direct Current
DE	Detailed Equivalent
DFSRF	Double-Frequency Synchronous Reference Frame
DP	Dynamic Phasor
DS	Direct Switch
EHV	Extra High Voltage
EMT	Electromagnetic Transient
EO-AAC	Extended Overlap AAC
FBSM	Full-Bridge Submodule
FB-MMC	Full-Bridge Modular Multilevel Converter
FCCC	Forced Circulating Current Control

FH-MMC	Full-Bridge and Half-Bridge Modular Multilevel Converter
FRT	Fault-Ride-Through
GTO	Gate Turn-Off
HB-MMC	Half-Bridge Modular Multilevel Converter
HBSM	Half-Bridge Submodule
HiL	Hardware-in-the-Loop
HVDC	High Voltage Direct Current
IGBT	Insulated gate bipolar transistor
LCC	Line Commutated Converter
LCC-D-MMC	Line Commutated Converter-Diode-Modular Multilevel
	Converter
LS-PWM	Level-Shift Pulse Width Modulation
MMC	Modular Multilevel Converter
ММСВ	Bank of Modular Multilevel Converter
MTDC	Multiterminal High Voltage Direct Current
MV	Medium Voltage
NLM	Nearest Level Modulation
ODE	Ordinary Differential Equation
OHL	Overhead Line
OLTC	On-load Tap Changer
OOC	Overlap Onset Control
PDE	Partial Differential Equation
PI	Proportional Integral
PLL	Phase Locked Loop
PM	Phasor Model
PR	Proportional Resonant
PS-PWM	Phase-Shift Pulse Width Modulation
PTP	Point-to-Point
PWM	Pulse Width Modulation
RK	Runge-Kutta
RMS	Root Mean Square

RT	Real-Time
SCR	Short-Circuit-Ratio
SFP	Shifted Frequency Phasor
SM	Submodule
SO-AAC	Short Overlap Alternate Arm Converter
SSN	State-Space Nodal
SV-PWM	Space Vector Pulse Width Modulation
THD	Total Harmonic Distortion
UGC	Underground Cable
UHVDC	Ultra High Voltage Direct Current
USC	Undersea Cable
VDCOL	Voltage Dependent Current Order Limiter
VSC	Voltage Source Converter
ZCS	Zero Current Switching

Abbreviation (Institution)	Definition
CIGRE	Conseil International Des Grands Reseaux Elecctriques
SGRI	State Grid Research Institution of China

Chapter 1

Introduction

1.1 Line-Commutated Converter (LCC) and Voltage Source Converter (VSC)-based Hybrid High-Voltage Direct Current (HVDC) Systems

High voltage direct current (HVDC) transmission has become an irreplacable part in the current electricity grid due to its unique advantages [1–3]. For long distance power transmission with the same voltage, HVDC is more efficient and economical than HV alternating current (AC), since the distance of DC transmission is not limited by the reactive power consumption and the cost of DC lines is lower than AC lines, while investments in DC stations are higher than AC stations [4]. Owing to the charging currents of the line capacitance, it is necessary to provide periodic reactive power compensation for AC transmission systems with the increase of transmission distance. Also, another limiting factor is the skin and proximity effect in AC lines, which will limit the current capacity of AC transmission. Hence, HVDC is more efficient and can play significant role in future long distance power transmission [5, 6].

From line-commutated converter (LCC) or current source converter (CSC)-based highvoltage direct current (HVDC) transmission system (Fig. 1.1) to conventional voltage source converter (VSC)- and modular VSC-based HVDC transmission systems (Fig. 1.2 and Fig. 1.3), hundreds of HVDC projects have been commissioned and are now under construction in the world [7]. LCC-based HVDC system is the most mature technology and it is still the best choice for bulk power transmission due to the high voltage and current rating of thyristors such as the Changji-Guquan ± 1100 kV UHVDC transmission project (Fig. 1.4). However, there are some issues with LCC topologies:

- i) Limited power reversal capability as the restriction of DC current direction,
- ii) possible commutation failures mainly caused by severe AC voltage drop, and
- iii) requirement of reactive power compensation due to the reactive power absorption from the AC system.
- iv) Investments in additional filter devices because of the characteristic and uncharacteristic harmonics in AC-side and DC-link generated by the natural commutation process.



FIGURE 1.1: LCC-based HVDC station made by Siemens: (a) main station, (b) thyristor module and (c) high power thyristors (4 inch, 5 inch and 6 inch) [8].

By comparison, VSC-based HVDC systems with fully-controlled devices have many benifits over LCC-based HVDC systems:

- i) independent control of active and reactive power,
- ii) reduced or no filter requirement,
- iii) no issue of commutation failures,
- iv) potential capability of connecting passive networks, and
- v) more feasible for the formation of multiterminal systems [1–3].



FIGURE 1.2: Conventional VSC-based HVDC station made by ABB [9].



(a)





(b)

(c)

FIGURE 1.3: Modular VSC-based HVDC station made by Siemens: (a) main station, (b) converter arm segment and (c) submodule [10].







FIGURE 1.4: Changji-Guquan ± 1100 kV LCC-based UHVDC transmission system: (a) project map view, (b) aerial view of Guquan station and (c) 1100 kV transformer [11].

It is noted that modular VSC topologies provide improved performance compared to conventional VSC-based HVDC. The main features are:

- i) modularity and scalability,
- ii) capability to handle high power ratings,
- iii) no filter requirement as they generate nearly sinusoidal output voltage,
- iv) low switching losses, and
- v) improved fault-ride-through (FRT) ability [4–6].

In addition, there is a range of modular VSC topologies that can be considered for hybrid HVDC system applications which include:

- i) MMC with half-bridge sub-modules (HBSMs) [4, 12, 13],
- ii) MMC with full-bridge SMs (FBSMs) or other bipolar SMs [14-19],
- iii) MMC with hybrid configurations of SMs [20, 21],
- iv) the alternate arm converter (AAC) [22-54], and
- v) other hybrid converter configurations [55, 56].

Fig. 1.5 shows the recently commissioned Zhangbei meshed four-terminal DC grid project based on the MMC with HBSMs.



FIGURE 1.5: Zhangbei meshed four-terminal DC grid [57].

There are still many challenges for these modular VSC topologies, although the stateof-the-art modular VSC-based HVDC possesses incomparable benefits compared with LCCbased HVDC transmission system. The DC fault tolerance level and DC FRT capability are the key concerns. The main approaches of dealing with DC fault in actual project is to i) open AC-breakers, ii) install DC-breakers and iii) use bipolar SMs. Opening AC breakers can isolate the AC-side and DC-link while the response of AC breakers is slow. DC breakers can interrupt the fault current at a short time but it requires high investment. In addition, bipolar SMs adopted in modular VSC topologies can limit the increase of fault current via the discharging of SM capacitors in blocking state, at the cost of losses and increased investment. Besides, the other main challenges of modular VSCs include:

- i) high cost and low power density of converters,
- ii) unbalanced energy and losses distribution amongst SMs,
- iii) circulating current suppression,
- iv) internal SM faults during long-term operation, and
- v) increased number of sensors and measurements.

Despite the issues of LCC, conventional VSC and modular VSC topologies, the commissioned and constructed HVDC projects have confirmed the feasiblity of these basic topologies [7]. In addition, some projects combine different topologies together taking the respective advantages. The two-level VSC (2L-VSC) is selected to combine with LCC for the initial study on hybrid HVDC systems [58–76]. With the appearance of MMC, current study of hybrid HVDC systems focuses on the combination of LCC and MMC or other modular VSCs [6, 21, 77–94]. In general, the benefits of LCC and VSC-based hybrid HVDC transmission system are summarized below:

- i) cost reduction and low losses,
- ii) no commutation failures,
- iii) potential connnection with passive networks and weak grids,
- iv) reduced requirement of filter devices , and
- v) improved DC fault clearing capability by force-retard of firing angle.

It is a paramount requirement for future development and extension of DC grids to integrate converters from multiple vendors [49]. Hence, the thesis studies the combination of LCC and modular VSC topologies, including the state-of-the-art MMC and the emerging DC-fault tolerant AAC, based on the current available benchmark models. The detailed parameter selection, steady-state and transient performance are considered demonstrating the feasibility of developed different hybrid HVDC systems in this thesis.

1.2 Literature Review

1.2.1 Background

The combination of LCC and VSC has long been attractive from the end of 20th century. However, the self-commutated converter of early hybrid HVDC system was gate turn-off (GTO) devices and mainly arranged at inverter side [58, 59]. Based on the hybrid PTP HVDC systems, the initial idea of constructing hybrid MTDC systems, was to replace the AC substation with rectifier and inverter station. The extra-HV (EHV) /HV AC substation is replaced by an LCC-based rectifier station, and the HV/medium voltage (MV) substation is replaced by VSC-based inverter station since most AC systems connected with HV/MV substations are passive networks [60, 61].

Due to the rapid development of power electronic devices, IGBT has become the main device for voltage source converters. In general, a hybrid HVDC system can be constituted by an LCC and two-level (2L), three-level (3L) or multilevel VSC. However, considering the requirements of modern projects, the combination scheme of LCC and modular VSC is more practical than the hybrid LCC-2L-VSC or LCC-3L-VSC scheme due to the increased handling ability of high power and voltage ratings [73].







FIGURE 1.7: Hybrid MTDC system: (a) configuration 1 and (b) configuration 2.

For a hybrid PTP HVDC system, LCC and VSC in most cases function as the rectifier and the inverter, respectively (Fig. 1.6(a)). The goal of this configuration is to avoid commutation failures on the inverter side. An alternative option is that of a VSC set up at the rectifier side for the scenario of connecting passive networks such as wind farms (Fig. 1.6(b)), while consequent commutation failures of LCC in the inverter side need to be further considered and tolerated. For hybrid MTDC transmission system, LCCs can transmit bulk power from energy zones which can be absorbed by VSCs separately (Fig. 1.7) [63–66]. The study of hybrid MTDC system is similar as hybrid PTP HVDC system, while the control scheme is more complicated considering the coordination of multi-converters. The following sections will explore the current study on hybrid HVDC systems based on LCC and VSC.

1.2.2 LCC and Conventional VSC-based Hybrid HVDC Systems

LCC-2L-VSC and LCC-3L-VSC are the two main candidates of LCC and conventional VSCbased hybrid HVDC systems. Based on the description of previous section, the commutation failures should be considered when 2L (3L)-VSC is located at rectifier side and LCC is in the inverter side. An alternative approach is that capacitor commutated converter (CCC) can be applied to replace LCC for avoiding commutation failures [71–73]. In addition, a line-commutated converter based on pulse width modulation (PWM)-CSC can also be applied for connecting with wind farms, which also possesses the capability of inherent short-circuit protection [66]. From the control logic, LCC in the inverter side can adopt conventional voltage dependent current order limiter (VDCOL), which is also appropriate to the hybrid HVDC systems with LCC and other modular VSCs.

The study of LCC and 2L (3L)-VSC-based three-terminal hybrid HVDC system is the first exploration to hybrid MTDC systems. Such a configuration is applied to study different conditions, such as system start-up, steady state operation and AC voltage sag in the inverter side [75]. The basic configuration of three-terminal hybrid HVDC system is that LCC and VSC are arranged in the rectifier side and inverter side, respectively, the third terminal can be either LCC or VSC operating on rectifier mode or inverter mode depending on the actual requirement. Likewise, the rectifier side should be VSCs when connecting with passive networks or weak grids [68, 76]. In general, the study of hybrid HVDC systems based on conventional VSCs paves the way for the combination of LCC and modular VSCs due to the similar external characteristics of conventional and modular VSCs.

1.2.3 LCC and Modular Multilevel Converter (MMC)-based Hybrid Point-to-Point (PTP) HVDC Systems

The MMC, as the state-of-the-art VSC topology is the favorable choice for hybrid HVDC systems, and the current hybrid HVDC projects all adopt the LCC-MMC hybrid topology [6, 95–97]. Given the different SMs in hybrid LCC-MMC HVDC system, MMC side can adopt different arrangements of SMs to satisfy specific demand. The HB-MMC is the most economic choice for a hybrid system but it lacks the capability of clearing DC fault. FB-MMC and other MMCs with bipolar SMs have DC fault ride-through capabilities and the voltage polarity can be changed without stopping operation, as is the case with HB-MMC, at the cost of higher losses and investment [77, 78, 80].

A possible method, based on installing two sets of high power diodes between two converters creating an LCC-Diode-MMC (LCC-D-MMC) system, can handle DC line faults when adopting HB-SMs, but the power flow of such a system cannot be reversed due to the unidirectional continuity of the diode [88–91]. Besides, the combination scheme of full-bridge and half-bridge MMC (FH-MMC) is favorable for hybrid systems considering the investment costs, which can use either symmetric FH-MMC or asymmetric FH-MMC, while the detailed ratio of different SMs needs to be further explored. It is noted that asymmetric FH-MMC can effectively balance SM capacitor voltages under DC-bus voltage deviations [21, 79].

The available configurations of hybrid PTP HVDC systems are shown in Fig. 1.8. The configuration of hybrid monopole system with 12-pulse LCC and MMC follows with the Fig. 1.8(a). In addition, there are four possible configurations for bipole hybrid HVDC system. Two sets of 12-pulse LCCs and an MMC can be employed (Fig. 1.8(b)), but the system can be in monopole operation if with two MMCs (Fig. 1.8(d)). Also, each station can employ hybrid topology which is a 12-pulse LCC and an MMC are respectively distributed at positive or negative pole (Fig. 1.8(c), (e)). The topology of Skagerrak 3 and Skagerrak 4 between Denmark and Norway follows the configuration of Fig. 1.8(e), where Skagerrak 3 comprises two 12-pulse LCCs in negative pole while Skagerrak 4 comprises two MMCs in positive pole (Fig. 1.9) [95].

For hybrid UHVDC transmission systems, more converters in one station should be arranged (Fig. 1.10). Fig. 1.10 (a) is the conventional configuration that two sets of 12-pulse LCCs and MMCs are respectively series-connected arranging at their individual station of each pole. Based on the conventional configuration, one MMC or 12-pulse LCC in each pole can be replaced by each other constituting different topologies (Fig. 1.10 (b), (c), (d)). In addition, the capacity issue of MMC can be resolved by parallel connection of converters which is also called bank of MMCs (MMCB) [98, 99]. Fig. 1.10 (b) is an improved configuration of MMC-MMC PTP system which offers better DC FRT capability



FIGURE 1.8: Available configurations of hybrid HVDC systems: (a) monopole configuration, (b) bipole configuration 1, (c) bipole configuration 2, (d) bipole configuration 3, and (e) bipole configuration 4.



FIGURE 1.9: One-line diagram of the Skagerrak hybrid HVDC system.



FIGURE 1.10: configurations of hybrid UHVDC systems: (a) configuration 1, (b) configuration 2, (c) configuration 3 and (d) configuration 4.

since the 12-pulse LCC can block the fault current of MMC during DC fault. In addition, the MMCB in the topology of Fig. 1.10 (c), as an extension of Fig. 1.10 (a), can provide reactive power support for 12-pulse LCC avoiding commutation failures to some extent. It is worth mentioning that the Baihetan-Jiangsu ± 800 kV UHVDC project plans to adopt the configuration of Fig. 1.10 (c) and three MMCs in parallel connection forms an MMCB (Fig. 1.11) [97]. Although the configuration of Fig. 1.10 (d) has worse efficiency than the configuration of Fig. 1.10 (c), it can be considered a combination of Fig. 1.10 (b), (c) and also has many benefits compared to the conventional configuration of Fig. 1.10 (a):

- i) potential capability of connecting with weak grid or passive network,
- ii) independent control of active and reactive power,
- iii) risk reduction of commutation failures, and
- iv) improved DC fault handling capability via force-retard of firing angle and converter blocking.



FIGURE 1.11: One-line diagram of the Baihetan hybrid HVDC systems.

1.2.4 LCC and MMC-based Hybrid Multiterminal HVDC (MTDC) Systems

The SM arrangements of LCC and MMC-based hybrid MTDC systems are similar as LCC and MMC-based hybrid PTP HVDC systems which have been described in Chapter 1.2.3.1. By setting up additional MMCs in the inverter side, the hybrid PTP HVDC system can be extended to the most commonly employed hybrid three-terminal HVDC system (Fig. 1.12(a)). Besides, multiple LCCs and MMCs in rectifier or inverter side can constitute more complex hybrid MTDC systems or simple DC grids (Fig. 1.12(b),(c)). Also, LCC can be as a sending end from long distance transmitting bulk power to a DC grid consisted by MMCs, as shown in Fig. 1.12 (d). Some current LCC PTP or LCC-MMC PTP sytems have the potential to further develop to a hybrid MTDC system, which now has been developed into a four-terminal HVDC system with two FB-MMCs [86].

For ultra long distance transmision, the hybrid UHVDC transmission systems will play significant role in the future [93]. The Kunliulong three-terminal hybrid UHVDC project (\pm 800kV, 8000MW) with hybrid MMC SMs was put into operation on December, 2020 (Fig. 1.13) [6]. Taking the three-terminal hybrid HVDC system as an example (Fig. 1.14), there are six basic operation modes for conventional hybrid multiterminal UHVDC system:

- i) bipole operation mode (Fig. 1.14 (a)),
- ii) monopole operation mode (ground return) (Fig. 1.14 (b)),
- iii) monopole operation mode (metallic return) (Fig. 1.14 (c)),

- iv) two-terminal operation mode (Fig. 1.14 (d)),
- v) bipole-monopole hybrid operation mode (Fig. 1.14 (e)) and
- vi) single valve group operation mode (Fig. 1.14 (f)).



FIGURE 1.12: Available configurations of hybrid MTDC systems: (a) configuration 1, (b) configuration 2, (c) configuration 3 and (d) configuration 4.



FIGURE 1.13: One-line diagram of the Kunliulong hybrid MTDC project.



FIGURE 1.14: Operation modes of the three-termianl hybrid UHVDC system: (a) bipole operation mode, (b) monopole operation mode (ground return), (c) monopole operation mode (metallic return), (d) two-terminal operation mode, (e) bipole-monopole hybrid operation mode, and (f) single valve group operation mode.

The three-terminal HVDC hybrid system can transfer into monople operation taking advantage of the bipole operation flexibility. The system can automatically run into two-terminal operation mode if one receiving terminal is out of operation due to faults. In addition, the bipole-monopole hybrid operation mode or single valve group operation mode is possible in hybrid MTDC system when one pole of receiving terminal or one valve group is isolated. In general, hybrid MTDC systems possess potential capability handling different conditions, whether from faults or actual requirements, due to the various operating modes [100].

Based on the above description, there have been important advances in hybrid HVDC
systems based on the LCC and the VSC. However, there are still many challenges of hybrid HVDC systems, especially the MTDC systems and future DC-grids, mainly reflecting in:

- i) improved DC fault tolerant topologies and efficient DC-breakers ensuring the reliability and resiliency of the hybrid HVDC systems,
- ii) fast FRT capability and effective protection schemes under different faults,
- iii) flexible power flow control and complete power flow analysis of hybrid MTDC systems and DC-grids,
- iv) coordinated control of LCC with different modular VSCs from multiple vendors, and detailed system performance of hybrid multi-converter DC power systems.

1.3 Thesis Objectives

The main objectives of the thesis are:

- To review the available benchmark models for HVDC converters, HVDC systems and DC grid stuides.
- To summarize modeling approaches of HVDC systems from the aspect of computational complexity and simulation accuracy.
- To propose generalized expressions of DC power flow considering mixed droop control.
- To develop LCC, state-of-the-art and emerging modular VSC topologies-based hybrid HVDC and MTDC systems filling the research gaps on more complex HVDC systems combining different converters from multiple vendors.
- To study hybrid DC grids based on LCC, state-of-the-art and emerging modular VSC topologies and verify the system static security based on the proposed generalized expression of DC power flow.

1.4 Simulation Models and Tools

For facilitating specific studies on HVDC systems, various computational models (from RMS load-flow model to full physics based model) are defined in current literature with

different computational complexity [101]. Also, the computational models cover EMT and electromechanical transient (phasor) models. The types of computational models and the detailed modeling approaches are summarized in Chapter 2.3 and Chapter 4.2, respectively.

In this thesis, PLECS-Blockset [102] and Simulink [103] are the used simulation tools for different HVDC simulation models. Detailed equivalent models are used for simulations in Chapter 4, 5, 6 and 7 to study MMC-based DC grid, LCC and AAC-based PTP HVDC system, LCC and AAC-based MTDC system, and LCC, MMC and AAC-based multiconverter DC grid, respectively.

1.5 Thesis Contributions

The main contributions of the thesis are:

- The available benchmark models for HVDC converters, systems and DC grid stuides are reviewed (Chapter 2). The research gaps, data availability and expansion capability of benchmark models are discussed for the development of future complex HVDC benchmark models.
- Based on the DC power flow calculation methods of single P/V and I/V droop control, a generalized expression of DC power flow is proposed to ensure the static security of MTDC systems and DC grids under mixed P/V and I/V droop control (Chapter 4). The detailed theoretical analysis is conducted for deriving the initial DC power flow for normal operation and the DC power flow after converter outage. The theory results are verified by the simulation results in an MMC-based DC grid.
- The AAC in a hybrid HVDC system is demonstrated in Chapter 5 constituting a hybrid LCC-AAC PTP HVDC system with detailed description of control hierarchy and parameter design scheme. The steady-state operation and reference tracking of the hybrid HVDC system are demonstrated. The corresponding control schemes for handling AC and DC faults are proposed in developed hybrid HVDC system, which are verified by the related simulation results showing the better DC fault handling capability of AAC in the hybrid HVDC system than HB-MMC.
- An LCC and AAC-based hybrid MTDC system is further developed in Chapter 6. The simulation results of the LCC-AAC hybrid MTDC system are presented under multiple operating scenarios, AC and DC faults with P/V droop control demonstrating the feasibility of AAC in hybrid MTDC systems and laying the fundation for understanding the AAC in more complex hybrid MTDC systems and future DC super grids combining multiple converters.

• A hybrid DC grid combining multiple converters is presented, including the LCC, MMC and AACs, with different combinations of DC OHLs and cables (Chapter 7). The combined AC and DC fault handling schemes in LCC, MMC and AAC for a hybrid DC grid are investigated validating the transient performance and fault recovery capability under AC and DC faults for the developed hybrid DC grid. The DC power flow for initial operation and contingencies under mixed P/V and I/V droop control in the hybrid DC grid is also demonstrated by the proposed generalized expression of DC power flow.

1.6 List of Publications

The work presented in this thesis has resulted in a number of peer-reviewed journal publications and refereed papers presented in international conferences.

1.6.1 Journal Papers

The following list of papers have been published in international journals.

- [1] **P. Sun**, H. R. Wickramasinghe, and G. Konstantinou, "Hybrid LCC-AAC HVDC transmission system," *Electric Power Systems Research*, vol. 192, 106910, Mar. 2021.
- [2] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "Hybrid Multiterminal HVDC System based on Line-Commutated and Alternate Arm Converters," *IEEE Transactions on Power Delivery*, pp. 1-1, 2021.

1.6.2 International Conference Papers

The following papers have been presented in international conferences:

- [3] P. Sun, F. Arrano-Vargas, H. R. Wickramasinghe, and G. Konstantinou, "Benchmark models for HVDC systems and DC-grid studies," in *Proc. ICPES 2019*, Dec. 2019, pp. 1-6.
- [4] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "An LCC-AAC hybrid highvoltage DC transmission system," in *Proc. 2020 IEEE ECCE-Asia*, Nov. 2020, pp. 1516-1521.

1.6.3 Papers Under Revision or Review

[5] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "Fault and Power Flow Analysis in Hybrid Multi-Converter DC Grids," *International Journal of Electrical Power & Energy Systems*, 2021. (Under Review)

1.6.4 Other Publications

Additionally, sections of the work completed in this thesis has contributed to the following publications:

- [6] H. R. Wickramasinghe, P. Sun, and G. Konstantinou, "Interoperability of Modular Multilevel and Alternate Arm Converters in Hybrid HVDC Systems," *Energies*, vol. 14 (5), 1363, 2021.
- [7] H. R. Wickramasinghe, P. Sun, and G. Konstantinou, "A hybrid VSC-HVDC system based on modular multilevel converter and alternate arm converter," in *Proc. IEEE IECON 2020*, Oct. 2020, pp. 4141–4146.

The association between the chapters of the thesis and the journal and conference publications is listed in the following table.

Chapters	Publications
Chapter 2. Benchmark Models for HVDC Converters, HVDC Systems and DC Grid Studies	[3]
Chapter 3. Topologies for Hybrid HVDC Systems	
Chapter 4. Modeling Approaches and Power Flow Analysis of HVDC Systems	[5]
Chapter 5. LCC and AAC-based Hybrid PTP HVDC Transmission System	[1], [4]
Chapter 6. LCC and AAC-based Hybrid MTDC Transmission System	[2]
Chapter 7. LCC and AAC-based Hybrid Multi-Converter DC Grid	[5]

1.7 Thesis Outline

The thesis is organized as follows:

Chapter 2 reviews all the available benchmark models for HVDC converters, systems and DC grid studies. Eleven available benchmark models are described in detail. Also, the research gaps, data availability and expansion capability of current HVDC benchmark models are further discussed, which would facilitate the establishment of future more complex HVDC models.

Chapter 3 analyzes the conventional LCC, state-of-the-art MMC and emerging DC-fault tolerant AAC for hybrid HVDC systems in the following chapters. The basic topologies, operation principles and control schemes of the LCC, MMC and AAC are discussed in detail.

Chapter 4 summarizes the modeling approaches of HVDC systems, including the EMT, electromechanical transient and two dynamic frequency capturing modeling methods. A generalized expression of DC power flow under mixed P/V and I/V droop control is proposed in this chapter to assess the system static security based on the detailed description of droop control in HVDC systems. The accuracy of the proposed generalized expression of DC power flow is verified in an MMC-based DC grid by the analysis of initial power flow for normal operation and the power flow after converter outage.

Chapter 5 develops an LCC and AAC-based hybrid PTP HVDC system with detailed description of control hierarchy and system parameter selection. The simulation results of steady-state operation, reference tracking are demonstrated first, then the fault handling capability are validated with proposed corresponding control schemes for handling AC and DC faults. The DC fault simulation results also verify the AAC in hybrid HVDC system shows better DC fault handling capability than HB-MMC.

Chapter 6 extends the LCC and AAC-based hybrid PTP HVDC system to a hybrid MTDC system combining LCC and two AACs. The performance of the hybrid MTDC model is verified for multiple operating scenarios including AC and DC faults via simulation. The results demonstrate the feasibility of the developed LCC and AAC-based hybrid MTDC system under P/V droop control and proposed DC FRT scheme.

Chapter 7 combines an LCC, an MMC and two AACs to constitute a hybrid multiconverter DC grid with different combinations of DC OHLs and cables. The simulation results validate the AC and DC FRT capability of the hybrid DC grid by the combined AC and DC fault handling schemes. The DC power flow analysis is also conducted in the hybrid DC grid by the generalized expression of DC power flow proposed in Chapter 4.

Finally, Chapter 8 concludes the main goals of the thesis and introduces future work.

Chapter 2

Benchmark Models for HVDC Converters, HVDC Systems and DC Grid Studies

2.1 Introduction

New research on topologies, control design, protection strategy and other aspects for HVDC systems can be greatly augmented by the availability of HVDC benchmark models. Benchmark models serve two key purposes. Firstly, they allow comparison of performance of control methods, control hardware and high-level coordination and protection strategies for different HVDC systems. Their secondary purpose is to provide a common reference basis for replication and cross-validation of results. Benchmark models developed with input from the industry can also address issues with parameter selection, typical design decisions and common configurations without intellectual property restrictions or project specific information. The objective of this chapter is to provide an overview of existing and most commonly used HVDC benchmark models covering all possible topology, station and system configurations. The goal of such review is to identify current gaps in availability of benchmark models and facilitate the development of more accurate and relevant HVDC benchmark models enabling future research pathways, especially towards the establishment of hybrid and complex converter and network models.

2.2 HVDC Transmission - From Converters to Systems

The nature of studies related to HVDC transmission ranges from detailed investigation of converter operation to integration of HVDC links to AC systems and DC-grids. This following sections briefly summarize different aspects of HVDC from power electronics topologies to systems.

2.2.1 HVDC Converter Topologies

LCC (Fig. 2.1(a)) represents the earliest and most mature converter technology for HVDC transmission. The power and voltage ratings of thyristors make them suitable for high-power transmission and UHVDC. Multiple projects with capacity over 6000 MW at 800 kV or 1100 kV have been commissioned in the last five years [6, 11, 104]. Nevertheless, grid strength requirements, AC filters and commutation failures are key challenges in the implementation of LCC-HVDC systems. The CCC of Fig. 2.1(b) addresses some of the issues, offering improved immunity to commutation failures, system stability and connection to weaker grids.

VSC-HVDC provides the grid with the controllability and flexibility of voltage source converters but is mainly limited by IGBT ratings. The series connection of IGBTs and switching frequency requirements in the original 2L-VSC of Fig. 2.1(c) were major limitations in expansion to higher power. Modular solutions such as the MMC (Fig. 2.1(d)) or the AAC (Fig. 2.1(e)) are the current state-of-the-art for VSC-HVDC and the topologies of choice for projects in the next decade. A power level of approximately 2 GW defines the maximum power rating with a variety of developments across multiple areas required to compete with LCC-HVDC in terms of maximum transfer capacity. The detailed operation and control of LCC and VSC topologies are outside the scope of this chapter, as they are analyzed in Chapter 3.

2.2.2 HVDC Converter Stations

In terms of converter station configurations, HVDC stations can be divided into monopolar station or bipolar station depending on the number of DC connectors that are at high voltage (Fig. 2.2(a) and (b), respectively). Monopole HVDC systems are simpler structures with a single (either the positive or the negative polarity) conductor at high voltage. This design option leads to overall lower cost systems. VSC-HVDC is usually configured as a symmetrical monopole (Fig. 2.2(a)). However, the additional flexibility and reliability that is provided by the bipolar configuration, despite the added cost, makes bipolar systems



FIGURE 2.1: HVDC converter topologies: (a) 12-pulse LCC, (b) 6-pulse CCC, (c) 2L-VSC, (d) MMC (one phase) and (e) AAC (one phase).



FIGURE 2.2: HVDC station configurations: (a) monopole station, (b) bipole station and (c) series-connected station.

the preferable option for LCC-HVDC. Connections between the two stations can be done either through the earth (or sea in case of undersea cables), through a metallic conductor (metallic return) in case of monopoles or with two terminals in the case of bipoles.

2.2.3 HVDC Systems

A classification of HVDC systems looks at their length and geographical extent. Backto-back (BTB) systems are two-terminal systems with no transmission line inbetween, located in the same converter station (Fig. 2.3(a)). They are mainly applied for power transmission between two asynchronous systems or systems at different frequency (i.e. Japan) [105].



FIGURE 2.3: HVDC system configurations: (a) back-to-back (BTB) system, (b) point-topoint (PTP) system and (c) multiterminal (MTDC) system.

Systems with two separate terminals and a transmission line (either an overhead line, underground or undersea cable) are the most common applications of HVDC systems (Fig. 2.3(b)). Extended connections of HVDC converters create multiterminal systems or DC-grids (Fig. 2.3(c)).

2.3 Types of HVDC Computational Models

Modeling of HVDC systems and all related simulations are typically aimed at a specific application which defines the level of detail included in each model. Another benefit of this approach is that simulation tools, solving algorithms and computation times can be identified and selected accordingly. In order to facilitate such analysis, seven types of computational models have been defined to describe the level of detail included in a model [101]. These are:

- Type 1 Full Physics Based Model
- Type 2 Full Detailed Model
- Type 3 Simplified Switchable Resistances Model
- Type 4 Detailed Equivalent Circuit Model
- Type 5 Average Value Model
- Type 6 Phasor Model
- Type 7 Root mean square (RMS) Load-Flow Model

Based on the above classification, the computational complexity of a model decreases as more simplifications are made. The detailed physics parameters required in Type 1 make the model extremely complex and difficult to simulate. Simplifications can be achieved by providing a generic but non-linear representation of switches (Type 2) and equivalent two-resistance switch models (Type 3).

Further simplifications through Thevenin/Norton equivalent circuits (Type 4) can lead to the development of average models where the switching functions are substituted by their averaged values (Type 5). Longer simulation times and system level evaluation such as simpler harmonic level analysis and load flows is then achieved with Type 6 and Type 7 models.

The level of detail in a model does not only define the computation time required for a specific system study but also the hardware requirements for developing real-time simulations and interfacing with external equipment in hardware-in-the-loop (HiL) configurations, a critical element of HVDC modeling and validation that will be discussed in following contents.

2.4 Review of HVDC Benchmark Models

Unsurprisingly, the first HVDC benchmark models were developed for LCC-HVDC systems. The CIGRE *"First Benchmark Model"* [106, 107] with the detailed parameters of a \pm 500 kV, 1000 MW PTP transmission system has been the basis of a number of studies on LCC-HVDC systems and implemented as an HVDC transmission example in many simulation software (Table 2.1). The models has since seen multiple revisions to its converter, filter and control parameters [108, 109] as well as a redesign for 60 Hz networks [110]. An additional LCC-HVDC model was developed for comparison between the NETOMAC and EMTP EMT simulation software in [111] while a benchmark model for a CCC BTB HVDC system was proposed in [112]. The CCC benchmark model, developed by the CIGRE B4-34 working group, was based on the Garabi (Brazil - Argentina) HVDC link.

The first 2L-VSC topologies were installed and commissioned in 1999 - 2000. However, despite their relative popularity in HVDC projects and the immense research interest they have attracted over the last 20 years, a benchmark model for the 2L-VSC HVDC is not commonly available. The same also applies to the 3L-VSC topology that has seen limited development in the mid-2000s. Since 2L/3L converters are no longer considered for the use in HVDC systems, it is expected that such a model will be made unavailable. Just in 2010, a benchmark model based on VSC topologies were proposed for phasor level simulations and DC-grid studies [113].

Interest in VSC-HVDC benchmark models was sparked by the development and subsequent commercial success of the MMC topology. A detailed MMC model was established for the planning of the Spain/France MMC-HVDC interconnector in 2012 [114]. In 2013, an extended benchmark model of MMC-based VSC-HVDC was developed by CIGRE B4-57 and B4-58 working groups [101, 115]. The model includes a total of 11 MMC terminals in multiple grid configurations and two different voltage levels (\pm 200 kV and \pm 400 kV). The benchmark model also included two DC-DC converters for HVDC applications, an application that has not seen any practical demonstration and is considered as a critical element for the future development of DC-grids with multiple voltage levels. The original CIGRE model has then been developed for both commercial real-time power system simulation platforms, RSCAD by RTDS Inc. and HYPERSIM by Manitoba Hydro and OPAL-RT [116].

In order to address the gap of the CIGRE model, which focused predominanty in the integration of offshore wind farms with VSC-HVDC, a model that combined LCC-HVDC with VSC-HVDC systems in an extended AC/DC grid was proposed by the State Grid Research Institution of China (SGRI). In addition to this model, a simplified benchmark model for interconnection of two AC systems was also built [117]. Following the development of hybrid and modular multilevel topologies, a benchmark model of an AAC-based PTP VSC-HVDC was also developed for real-time studies [25, 118].

In total, there are eleven HVDC benchmark models currently available in the literature, predominantly developed for phasor, EMT and power flow studies. In addition, CIGRE WG B4-72 working group in 2020 develops a comprehensive DC grid benchmark model, which combines the summarized seventh, ninth and tenth benchmark models (Fig. 2.4). A comprehensive summary of these benchmark models is provided in Table 2.1.



FIGURE 2.4: CIGRE B4-72 comprehensive DC grid benchmark model.

J F	Kel.		[107] [106]		[108]		[111]		[112]				[112]	[CTT]				[114]
Model	type		DE (4)		DE (4)		DE (4)		DE (4)					[IN (U)				AVM (5)
	study		EMT		EMT		EMT		EMT				Dhacor	100011				EMT
Software/	Hardware	FGH Parity Simulator	EMTDC EMTP	NETOMAC	No data	FMTD	T T TATT	NETOMAC	PSCAD/EMTDC	No data						EMTP		
Line	(km)		UGC (100)		UGC (100)	USC (66)	OHL1 (43)	OHL2 (42)	UGC (1000)	DGC						UGC (70)		
E	leriminals		7		2		7		2	1	c	٩	c	٩		3 (DC Hub)		2
Power	(MM)		1000		1000		600		2000	1400	1000	700	1000	700	1400	600	400	1000
Voltage	(lkV)		500		500		400		± 500	±500 ±320						± 320		
,	recurrorogy		TCC		TCC		LCC		CCC	VSC					MMC			
17.	rear		1991		1994		1995		2008				0100	0107				2012

TABLE 2.1: Summary of HVDC benchmark models.

Voor	Tophaologi	Voltage	Power	Tominulo	Line	Software/	Ctudur	Model	₽°€
Ісаі	reciliology	(kV)	(MM)	ICIIIIII	(km)	Hardware	ornuy	type	.Tel
		DCS1: ± 200	8000	2	UGC (200)				
			200		נוטט עצטט)				
		DCS2: ±200	800	4	000 000	MATLAB	Power Flow		
2013	MMC		1200			RSCAD	EMT		[101]
			2×400			HYPERSIM	RT	RMS (7)	
		DCS3: ±400	$2{\times}800$	IJ					
			2×1200						
2014	TCC	500	1000	2	UGC (100)	PSCAD/EMTDC	EMT	DE (4)	[110]
		DCS-A: ±500		4	OHL (2500)				
		DCS-B: ± 800		7	OHL (6250)				
2015	LCC/VSC	DCS-C: ±320	No data	2	OHL (500)	PSS/E	Power Flow	RMS (7)	[117]
		100 - ים צעם		y	UGC (1050)				
		レ し ふ-レ. 日400		D	OHL (1300)				
			1×1000						
		± 800	1×1500						
2015	VSC	± 500	2×3200	13	OHL (9100)	PSS/E	Power Flow	RMS (7)	[117]
		±400	5×7000						
			3×14000						
2019	AAC	± 200	800	2	UGC (200)	RSCAD	RT	DE (4)	[118]
Detailed L: Overhea	equivalent circuit	models PM: Ph Underground cable	asor models USC: Une	AVM: Averag dersea cable	ge value models	RMS: RMS load flo	w models RT	r: Real-time	

DE: I OHL

2.5 Discussion and Conclusion

2.5.1 Research Gaps in Benchmark Models

With the rapid development of HVDC technology, continuous increase in voltage level and capacity expansion of multiterminal networks, HVDC systems are again a very important research topic. However, HVDC transmission now is not limited to a small range of PTP transmission systems. For long-distance, high-power transmission, UHVDC is the first choice, but currently there is no established benchmark models for UHVDC transmission. Such a model, that removes some of the assumptions in the development of studies and control methods will set the basis for further research in the field potentially contributing to improve economic efficiency and pave the way for future large-scale UHVDC project construction.

There is also plenty of room for further development of the existing multiterminal benchmark models. Although the CIGRE DCS benchmark model and the SGRI DCS-M large-scale model can be used to conduct related multiterminal model research at this stage, the two models still have some limitations, as they were both created to meet specific project requirements. The CIGRE DCS benchmark model was developed for the integration of offshore wind farms and only considers a single topology, the MMC, as the basis of its converters. SGRI's DCS-M is a versatile benchmark model that covers onshore renewable energy grid, energy storage and long-distance power transmission studies but also offers room for improvement. For instance, it uses simple AC network models at both ends of the interconnections which removes some of the complexities in integrating large DC systems into AC networks. Some other gaps in existing benchmark models are:

- i) models suited for large-scale AC and DC-grid studies,
- ii) models for validation of grid support functions (for example virtual inertia) by HVDC systems,
- iii) coordination of multiple converters to address issues such as controller interoperability and multi-vendor system compatibility - as analyzed in [119].

2.5.2 Data Availability

A critical element of all benchmark models is their openness, typically through availability of appropriate data so that the models can be used for different studies and across various software. The nature of the first HVDC benchmark model [108] is relatively simple, especially compared to modern implementations so a full of set of data (e.g. voltages, converter and line impedances, transformer ratios, firing and extinction angles, shortcircuit-ratios (SCR)). Therefore, a large number of model simulation studies have been performed based on this HVDC benchmark model, such as control improvement, small signal analysis, modeling method, etc. [120–127]. A similar approach in data availability can be seen in the development of the CCC model, with the proposed benchmark model verified through PSCAD/EMTDC and ATP simulations [112].

With regards to VSC-HVDC benchmark models, the 401-level MMC benchmark model [114] has detailed converter parameters for EMT type simulations together with associated results for validation while the AAC benchmark model has been fully developed for the RTDS real-time digital simulator with detailed parameters of the HVDC stations, transmission lines, and control parameters openly provided.

For larger benchmark models, data availability becomes a more complex issue. For example, the North Sea VSC-HVDC benchmark test system [113] only provides basic data, i.e. DC voltage levels and converter power ratings. Additional details are provided in the SGRI DCS-M benchmark model [117] although the full details of power rating cannot be easily found on the Internet. The CIGRE DCS models [115] are currently the most open DC-grid models in terms of data availability which has facilitated further developments for example through model conversion to real-time simulators.

The above discussion signifies the importance of developing fully open benchmark models for HVDC converters, HVDC systems and DC grid studies. Broader industry participation and direct involvement in what benchmark models and studies and what high-level assumptions can be made for HVDC systems can help in focusing research efforts around the world without the need for multiple validations of proposed approaches, for example how a particular design choice might exaggerate or obfuscate certain HVDC converter or system behaviors.

2.5.3 Expansion of Types and Complex Models

As the nature of studies involving HVDC systems is continuously expanding, it is necessary that benchmark models provide the same versatility and be available - to the extent that it is possible - for multiple simulation types. The CIGRE DCS model [101] and its current availability for multiple simulation types provides a practical example of the benefits of such an approach.

An additional benefit from co-ordinated benchmark development with open data that can be easily interchanged would be the expansion into extended and more complicated models. These extended HVDC models can be derived by combination or integration of two or more smaller benchmark models, e.g. for multiterminal system studies, embedded HVDC systems, extended DC-grids, etc.

It is envisioned that open and fully developed benchmark models across multiple softwares/harwares from academia and industry will provide reliable reference basis, available parameter selection, and allow performance comparison of different HVDC systems.

Chapter 3

Topologies for Hybrid HVDC Systems

3.1 Introduction

The LCC and VSC are the two main technologies in current HVDC projects. The modular VSC technology has the most potential for future MTDC systems due to its modularity, scalability and improved harmonic properties of output voltage waveforms. Compared to the LCC, the VSC can regulate power flow flexibly and absorb/generate reactive power independently. Although the LCC lacks flexible power reversal capability and has to be equipped with additional AC filters, the thyristor-based LCC is still the best choice for bulk power or long-distance UHVDC transmission because of the high current and voltage ratings of thyristors. LCC and VSC topologies can be used in different power transmission scenarios, and the combination of various LCC and VSC topologies is a promising solution in future more complex HVDC systems. The state-of-the-art MMC and DC-fault tolerant AAC are the two representative modular VSC topologies, which are also used for the study of hybrid HVDC systems combining with the LCC in the following chapters of this thesis.

This chapter aims to provide an overview of the LCC, MMC and AAC for the development of hybrid HVDC systems. The system topologies, operation characteristics and control schemes of the LCC, MMC and AAC are analyzed comprehensively.

3.2 Line-Commutated Converter (LCC)

The technology of LCC was developed rapidly from 1970s to 1980s [128]. The LCC is a conventional and mature HVDC technology, which is suitable for UHVDC transmission due to the large voltage and current rating of thyristors [129]. The LCC uses thyristor-type

valves to commutate and works at foundamental frequency, hence the switching losses are less than the VSC [1, 3]. Considering better harmonic performance, double 6-pulse (12-pulse) bridge is usually used in LCC-based HVDC projects. The past few decades have seen the maturity of the LCC. The analysis and operation of the LCC have been reported extensively in the literature [1, 3, 49, 50, 53, 129–137]. This following section describes the basic topology, operation and control of the LCC.

3.2.1 Basic Topology and Operating Principles of the LCC



FIGURE 3.1: Converter topology of the LCC: (a) 6-pulse converter and (b) 12-pulse converter.

Fig. 3.1(a) and (b) depict the LCC in the rectifier side with 6-pulse and 12-pulse converter, respectively. Each phase-leg of the LCC has a thyristor valve with several seriesconnected thyristors. The thyristors of the LCC perform switch-on operation, while have to be forced to switch off. Each thyristor valve switches on with pulse signals generated by the firing angle, and the switching orders follow with the number sequence of thyristor valves in the figure. If the commutation process is ignored, there are only 2 valves (4 valves for 12-pulse converter) to be switched on at the same time, and the ideal no-load average DC voltage can be expressed as:

$$V_{dcr1} = kNV_{2tr}\cos\alpha,\tag{3.1}$$

where k is $3\sqrt{2}/\pi$ for a 6-pulse thyristor bridge, N represents the number of 6-pulse converters (N = 2 in 12-pulse converter), V_{2tr} refers to the rms value of line-to-line voltage in the secondary side of transformer, and α is the firing angle. However, the commutation process has to be considered due to the non-negligible system inductance. During the commutation process, three valves are switched on together leading to the overlap of

valve currents, hence the average DC voltage is decreased following with:

$$V_{dcr2} = kNV_{2tr}\cos\alpha - \frac{3N\omega L_{cr}}{\pi}I_{dc},$$
(3.2)

where L_{cr} is the commutation inductance (leakage inductance of transformer) in rectifier side. Similarly, the average DC voltage for the LCC in inverter side can be calculated as:

$$V_{dci} = kNV_{2ti}\cos\beta + \frac{3N\omega L_{ci}}{\pi}I_{dc}, \quad \text{and}$$
(3.3)

$$V_{dci} = kNV_{2ti}\cos\gamma - \frac{3N\omega L_{ci}}{\pi}I_{dc},$$
(3.4)

where β is the advance angle, γ is the extinction angle and $\pi = \alpha + \beta = \alpha + \gamma + \mu$ (μ is the commutation or overlap angle).

The DC voltage of 6-pulse and 12-pulse LCC contains 6k and 12k order characteristic harmonics, respectively. Moreover, the AC current for Yy or Yd connected transformer contains $6k \pm 1$ characteristic harmonics without considering commutation process as:

$$i_{Yy} = \frac{2\sqrt{3}}{\pi} I_d(\cos\omega t - \frac{1}{5}\cos 5\omega t + \frac{1}{7}\cos 7\omega t - \frac{1}{11}\cos 11\omega t + \frac{1}{13}\cos 13\omega t - \frac{1}{17}\cos 17\omega t + \frac{1}{19}\cos 19\omega t...), \quad \text{and}$$
(3.5)

$$i_{Yd} = \frac{2\sqrt{3}}{\pi} I_d(\cos\omega t + \frac{1}{5}\cos 5\omega t - \frac{1}{7}\cos 7\omega t - \frac{1}{11}\cos 11\omega t + \frac{1}{13}\cos 13\omega t + \frac{1}{17}\cos 17\omega t - \frac{1}{19}\cos 19\omega t...).$$
(3.6)

The 12-pulse LCC uses a set of Yy and Yd connected transformer cancelling out the $(12k - 6 \pm 1)$ order harmonics, thus the AC current for the 12-pulse LCC is:

$$i_{Yy,Yd} = \frac{4\sqrt{3}}{\pi} I_d(\cos\omega t - \frac{1}{11}\cos 11\omega t + \frac{1}{13}\cos 13\omega t - \frac{1}{23}\cos 23\omega t + \frac{1}{25}\cos 25\omega t...).$$
 (3.7)

Eq. (3.7) shows there are only $12k \pm 1$ order characteristic harmonics in the AC current for Yy and Yd connected transformer, hence the tuning points of AC filters should be set at $12k \pm 1$ order, especially the 11^{st} and 13^{rd} characteristic harmonics.

Since the existence of harmonics in DC link and AC side, the LCC is necessary to be equipped with AC and DC filters. Additionally, AC filters play the role of reactive power compensation. The configuration of AC and DC filters are discussed in Chapter 3.2.3. The next section analyzes the conventional and improved control schemes of LCC.

3.2.2 Control Schemes of the LCC

3.2.2.1 Conventional Control Schemes

The LCC in the rectifier and inverter side generates firing signal to thyristor valves via different control modes [128, 130]. For the LCC in the rectifier side, there are four conventional control modes:

- i) direct firing angle (α) control that a constant firing angle is set for the LCC in the rectifier side.
- ii) DC current (I_{dc}) control that the DC current is controlled to be constant via the dynamic adjustment of the firing angle.
- iii) DC power control (P_{dc}), but it is generally achieved by controlling the DC current derived from the expected DC power.
- iv) DC voltage (V_{dc}) control that the firing angle is adjusted dynamically via the DC voltage controller to maintain constant DC voltage.

Moreover, the LCC in the inverter side can adopt:

- i) direct advance angle (β) control and the advance angle is below 90° due to the firing angle is above 90° for the LCC in the inverter side.
- ii) direct extinction angle (γ) control that the extinction angle is maintained to be constant via the adjustment of advance angle.
- iii) DC current (I_{dc}) control that the output DC current tracks the specified reference by adjusting the advance angle in the inverter side.
- iv) DC voltage (V_{dc}) control that the DC voltage is maintained to be constant by the DC voltage controller in the inverter side.

Table 3.1 summarizes the 13 possible combination schemes. Scheme 6 and 7 are the two commonly used control strategies in commissioned LCC-based HVDC projects considering control flexibility and automatic control capability [128]. In the control scheme 6, the LCC in the rectifier side adopts constant DC current control and the extinction angle is controlled by the inverter side, while the inverter side in the control scheme 7 controls the system DC voltage. The steady-state operating points for two schemes are shown in Fig. 3.2(a) and (b), respectively.

Scheme Terminal	1	2	3	4	5	6	7	8	9	10	11	12	13
Rectifier	α	α	α	α	I_{dc}	I_{dc}	I_{dc}	P_{dc}	P_{dc}	P_{dc}	V_{dc}	V_{dc}	V_{dc}
Inverter	β	γ	I_{dc}	V_{dc}	β	γ	V_{dc}	β	γ	V_{dc}	β	γ	I_{dc}

TABLE 3.1: Basic control schemes of the LCC.



FIGURE 3.2: Conventional control schemes of the LCC: (a) control scheme 6 and (b) control scheme 7.

However, the above described conventional control schemes for the LCC do not include backup control. The next section discusses improved control schemes of the LCC including the limitation of minimum firing angle, maximum advance angle and minimum extinction angle, and VDCOL control for avoiding commutation failures and assisting fault recovery.

3.2.2.2 Improved Control Schemes

Fig. 3.3 depicts the corresponding improved control schemes of scheme 6 and 7. The V-I characteristic curves for the LCC in the rectifier side and inverter side consist of 4 sections and 6 sections, respectively. The 4 sections of the LCC in the rectifier side describe the following control characteristics:

- AB (A'B'): minimum firing angle control characteristic,
- BC (B'C'): constant DC current control characteristic,
- CD (C'D'): VDCOL control characteristic in the rectifier side, and
- DE (D'E'): constant DC current control characteristic after VDCOL.

The 6 sections of the LCC in the inverter side describe:



FIGURE 3.3: Improved control schemes of the LCC: (a) improved control scheme 6 and (b) improved control scheme 7.

- FO (F'O'): constant extinction angle control characteristic (constant DC voltage control characteristic with minimum extinction angle limitation),
- OG (O'G'): DC current deviation (ΔI) control characteristic,
- GH (G'H'): constant DC current control characteristic,
- HI (H'I'): VDCOL control characteristic in the inverter side,
- IJ (I'J'): constant DC current control characteristic after VDCOL, and
- JK (J'K'): maximum advance angle control characteristic.

It is noted that the VDCOL control sets in rectifier side guarantees smooth recovery of DC current from AC faults, while the risk of commutation failures can be reduced if it is set in inverter side [50, 138]. In addition to the described improved control schemes, the LCC can also adjust AC system frequency and reactive power to satisfy some other specific requirements such as offshore wind farm connection [135].

3.2.3 AC and DC Filters of the LCC

AC and DC filters in the LCC include both passive and active filters, and passive filters are commonly used in commissioned HVDC projects. The passive filters can be divided into:

- i) tuned filters (e.g. single-tuned filter, double-tuned filter and triple-tuned filter) and
- ii) damped filters (e.g. single-tuned damped filter, double-tuned damped filter, tripletuned damped filter and C-type damped filter).

In damped filters, additional resistors are connected with inductors in parallel to provide damping characteristics for filters. The damped filters show better frequency deviation tolerance than the tuned filters and the resonance can be prevented, while the high impedance of damped filters at tuning points leads to relatively poor tuning performance [128, 130].

Fig. 3.4 shows the AC filter of CIGRE LCC benchmark model in rectifier side [106, 107] and a designed DC filter of the LCC, respectively. The AC filter in Fig. 3.4(a) is composed of a shunt capacitor, C-type damped filter (tuning point is set at 3^{rd}) and single-tuned damped filter (filtering 11^{st} harmonics). The designed DC filter in Fig. 3.4(b) is a double-tuned damped filter for filtering 12^{nd} and 24^{th} harmonics. Table 3.2 lists the parameters of AC and DC filters.



FIGURE 3.4: AC and DC filters of the LCC: (a) AC filter of CIGRE LCC benchmark model in rectifier side, and (b) a designed DC filter of the LCC.

In addition, the design of AC filters in the LCC has to consider the reactive capacity for compensating the reactive power consumption of the converter. Typically, the reactive power absorbed regardless converters in rectifier side or inverter side is about 40% to 60% of rated active power [128, 130]. The rated active power of CIGRE LCC benchmark model is 1000 MW, and the reactive power consumption is 542.4 MVar (54.24% of rated active power) [106, 107]. The total reactive capacity for the AC filter of CIGRE LCC

Daramotors	AC f	DC filter		
Farameters	Shunt capacitor	3^{rd}	11^{st}	$12^{nd}/24^{th}$
Capacity (MVar)	125	250	250	-
C_{a1} (μ F)	3.342	-	-	-
C_{a2} (μ F)	-	6.685	-	-
C_{a3} (μ F)	-	74.28	-	-
C_{a4} (μ F)	-	-	6.685	-
L_{a1} (mH)	-	0.1364	-	-
L_{a2} (mH)	-	-	0.0136	-
R_{a1} (Ω)	-	29.76	-	-
R_{a2} (Ω)	-	261.87	-	-
R_{a3} (Ω)	-	-	83.32	-
C_{d1} (μ F)	-	-	-	0.35
C_{d2} (μ F)	-	-	-	0.81
<i>L</i> _{<i>d</i>1} (mH)	-	-	-	89.35
L_{d2} (mH)	-	-	-	48.86
R_{d1} (Ω)	-	-	-	10000

TABLE 3.2: Detailed parametters of the AC and DC filters.

benchmark model in rectifier side is 625 MVar, hence it can compensate the reactive power consumption of the converter.

3.3 Modular Multilevel Converter (MMC)

Since the first MMC-HVDC project (Trans Bay Cable) was commissioned in 2010 [13], many MMC-based HVDC projects including the MTDC projects have been commisioned in succession or are now under construction. The MMC uses series-connected SMs rather than direct series-connected switches in each arm, which offers several salient features as modularity, scalability and excellent harmonic performance [2, 4, 4, 12, 14–19, 101, 114, 139–147]. Hence, the MMC is a potential candidate in HVDC transmission systems. The following sections first describe the topology of the MMC, then the operation and control of the MMC are discussed in detail.

3.3.1 Basic Topology and Operating Principles of the MMC

The schematic of a three-phase MMC is shown in Fig. 3.5. Each phase-leg of the MMC includes N SMs per arm in series and one arm inductor (L), and the arm inductor are used to limit the circulating current and the fault currents. There are various of SMs can

be used for the MMC including unipolar and bipolar SMs, as HBSM, FBSM, flying-capacitor SM, clamped-double SM, etc. [148]. The HBSM is the widely used simplest SM for the MMC, and the following description studies the HBSM-based MMC. The HBSMs in each arm of the MMC are controlled to generate the AC phase voltages, and the actual switching states decide the output voltage of each SM which is either equal to the capacitor voltage (v_c) or zero.



FIGURE 3.5: Three-phase topology of the MMC.

Employing the superposition theory, each phase-leg of the MMC can be separated into common mode circuit and differential mode circuit. The common and differential mode voltages (v_{commk} and v_{diffk}) can be expressed as:

$$v_{commk} = \frac{v_{uk} + v_{lk}}{2}, \qquad \text{and} \tag{3.8}$$

$$v_{diffk} = \frac{v_{uk} - v_{lk}}{2},\tag{3.9}$$

respectively, where v_{uk} and v_{lk} are the voltages across to the upper and lower arm inductors, and k is the corresponding phase ($k \in (a, b, c)$). Similarly, the common and differential mode currents (i_{commk} and i_{diffk}) are:

$$i_{commk} = \frac{i_{uk} + i_{lk}}{2}, \qquad \text{and} \tag{3.10}$$

$$i_{diffk} = \frac{i_{uk} - i_{lk}}{2},$$
 (3.11)

respectively. The differential mode current in each arm also refers to the circulating current (i_{circk}) in this thesis, and the direction of circulating current is depicted in Fig. 3.5. Correspondingly, the arm currents (i_{uk} and i_{lk}) in three phases under balanced operating conditions can be obtained as:

$$i_{uk} = \frac{i_k}{2} + i_{diffk}, \qquad \text{and} \tag{3.12}$$

$$i_{lk} = \frac{i_k}{2} - i_{diffk},\tag{3.13}$$

The differential voltage is employed to regulate circulating current in the MMC, which can also be expressed as:

$$v_{diffk} = L\frac{d}{dt}i_{diffk} = L\frac{d}{dt}i_{circk}.$$
(3.14)

Fig. 3.6 depicts the modulation waveform (v_{am}) and duty cycles (d_{ua} and d_{la}) in phase a, which are expressed as:

$$v_{am} = m_a \cos \omega t, \tag{3.15}$$

$$d_{ua} = \frac{1 - v_{am}}{2},$$
 and (3.16)

$$d_{la} = \frac{1 + v_{am}}{2},\tag{3.17}$$

respectively, where m_a is the modulation index of the MMC and $m_a \in (0, 1)$. The number of SMs to be inserted/bypassed in each arm are determined by the modulation stage based on the above obtained duty cycles. The switching states (0 and 1) for upper and lower arms are defined as s_{uk} and s_{lk} , hence the voltages applied to the upper and lower arm



FIGURE 3.6: Operation of the MMC arms.

inductors (v_{uk} and v_{lk}) are:

$$v_{uk} = \frac{v_{dc}}{2} - \sum_{i=1}^{N} (s_{uki} \cdot v_{cuki} + i_{uk}R_{on} + \operatorname{sgn}(i_{uk}) \cdot V_f), \quad \text{and} \quad (3.18)$$

$$v_{lk} = -(\frac{v_{dc}}{2} - \sum_{i=1}^{N} (s_{lki} \cdot v_{clki} - i_{lk}R_{on} - \operatorname{sgn}(i_{lk}) \cdot V_f)),$$
(3.19)

where v_{cuki} and v_{clki} are the capacitor voltages in upper and lower arms, R_{on} and V_f are the on-resistance and forward voltage drop across the MMC semiconductors, respectively.

3.3.2 Control Schemes of the MMC

3.3.2.1 Control Hierarchy

The control hierarchy of the MMC in HVDC consists of four main layers (Fig. 3.7). The function of the top layer (AC/DC grid control) is for necessary scheduling and dispatching. The second layer (coordinated system control) handles unscheduled events and defines the reference set-points for converters. The third layer (converter station control) performs the high level control of the MMC. The active/reactive power, node voltages/currents and frequency in the third layer are regulated based on the references. The bottom layer is the internal converter control, which is also the low-level control of the MMC that regulates the i) SM sorting, ii) SM energy, iii) circulating current and iv) arm energy [50, 51].



FIGURE 3.7: Control hierarchy of the MMC.

The high and low level control for the MMC can be divided into three main stages: i) outer controller, ii) inner current controller and iii) modulation & sorting. The outer controller for the MMC corresponds to the high level control generating the decoupled output current references (i_d^*, i_q^*) for inner current controller to calculate the final output voltage reference. The required number of SMs for each arm is determined in modulation stage and different modulation techniques can be applied while the nearest level modulation (NLM) is preferable to other pulse width modulation (PWM)-based methods with the increase of SMs [118]. The following section introduces the modulation techniques of the MMC.

3.3.2.2 Modulation Techniques

Different modulation techniques are used in the modulation stage to determine the switching states of SMs in the upper and lower arms based on expected modulation waveform. The typical modulation techniques are PWM- and staircase modulation-based methods. PWM-based methods deliver good tracking performance of modulation waveform with relatively simple implementation, and carrier-based PWM (CB-PWM), selective harmonic elimination PWM (SHE-PWM), space vector PWM (SV-PWM), etc. can be applied to the MMC. CB-PWM is the commonly used modulation scheme in multi-level converters, which can be classified into phase-shift PWM (PS-PWM) and level-shift PWM (LS-PWM) [139, 140, 142, 149].

The implementation of staircase modulation methods is simpler than PWM-based modulation methods with the increase of voltage levels [141]. The commonly applied staircase modulation method is NLM and two solutions can be used which are angle- and amplitudebased approaches. Based on (3.20) and (3.21), the pre-calculation of angles and reference voltage amplitude can be obtained. By comparison, the angle-based approach is more efficient than the amplitude-based approach due to less high frequency oscillations [101].

$$\theta_{(h,h+1)} = \sin^{-1}\left(\frac{2h}{N-1}\right),$$
(3.20)

$$N_{arm} = \frac{V_{dc}}{\bar{V}_c},\tag{3.21}$$

where \bar{V}_c is the average value of capacitor voltage.

3.3.2.3 SM Capacitor Voltage Balancing

The energy storage in DC-link of the MMC is maintained by series connected capacitors in arms, hence it is not only crucial to maintain constant DC voltage, but also to balance all SM capacitor voltages. The widely used algorithm of SM capacitor voltage balancing is called "sort and select" [144]. Fig. 3.8 illustates the basic process of conventional SM capacitor voltage balancing algorithm, which are separated into three steps:

- 1) sorting SM capacitor voltages,
- 2) measuring arm current directions, and
- 3) selecting SMs to be inserted/bypassed based on arm current directions and instantaneous values of SM capacitors.

The sorting stage in the SM capacitor voltage balancing algorithm is intended to sort the SMs based on the detailed arm current directions, determined number of SMs to be inserted and current capacitor voltages [144]. However, PS-PWM can use simple proportional-integral (PI) controllers to regulate the SM capacitor voltages avoiding sorting stage, while the harmonic performance is reduced [41].



FIGURE 3.8: Conventional SM capacitor voltage balancing algorithm.

The conventional voltage balancing algorithm, while simple and efficient, will lead to high switching frequency. Various improved methods are reported in current literature for reducing switching frequency [146, 150, 151]. A modified algorithm, shown in Fig. 3.9, can reduce the switching frequency of SMs and ensure low complexity. This algorithm consists of descending sorting stage, ascending sorting stage and comparator to determine the selection of SMs. In addition, the restricted algorithm (Fig. 3.9(b)) can avoid additional switching transitions by adding a constant voltage offset K compared to the algorithm in Fig. 3.9(a), hence it further decreases switching frequency and losses [143].

3.3.2.4 Circulating Current Control

The circulating current control in the MMC also refers to arm current control, which can i) suppress second harmonic current, ii) maintain SM capacitor voltage, iii) regulate average SM energy and iv) balance upper/lower arm energy. The most commmly used methods for



FIGURE 3.9: Modified SM capacitor voltage balancing algorithm: (a) conventional algorithm and (b) restricted algorithm.

arm current control are proportional-resonant (PR) controllers and double-frequency synchronous reference frame (DFSRF)-based techniques [145]. Circulating current suppression control (CCSC) and forced circulating current control (FCCC) are the two prevalent techniques representing the two commmly used methods [146, 147].

Eq. (3.22) shows that the differential currents in three phases consist of AC part (second harmonic current component) and DC part (one-third of the total DC current), and the steady-state second harmonic currents are of negative sequence. The purpose of CCSC is to eliminate the second harmonic AC component in differential currents. Two DC components ($i_{diff,d}$ and $i_{diff,q}$) can be obtained by transforming three-phase differential currents in the DFSRF as (3.23).

$$\begin{cases} i_{diffa} = \frac{I_{dc}}{3} + \hat{I}_{2a}\cos(2\omega t + \phi), \\ i_{diffb} = \frac{I_{dc}}{3} + \hat{I}_{2b}\cos\left[2(\omega t - \frac{2\pi}{3}) + \phi\right] = \frac{I_{dc}}{3} + \hat{I}_{2b}\cos(2\omega t + \frac{2\pi}{3} + \phi), \quad (3.22) \\ i_{diffc} = \frac{I_{dc}}{3} + \hat{I}_{2c}\cos\left[2(\omega t + \frac{2\pi}{3}) + \phi\right] = \frac{I_{dc}}{3} + \hat{I}_{2c}\cos(2\omega t - \frac{2\pi}{3} + \phi). \\ \left[\begin{array}{c} v_{diff,d} \\ v_{diff,q} \end{array} \right] = L\frac{d}{dt} \left[\begin{array}{c} i_{diff,d} \\ i_{diff,q} \end{array} \right] + \left[\begin{array}{c} 0 & -2\omega L \\ 2\omega L & 0 \end{array} \right] \left[\begin{array}{c} i_{diff,d} \\ i_{diff,q} \end{array} \right]. \quad (3.23) \end{cases}$$

Fig. 3.10(a) shows the CCSC configuration based on (3.23). The differential current (i_{diff}) is first derived from arm currents following with (3.11), and PI controllers are used to minimize the second harmonic current via driving the two transformed DC components

to zero. The differential voltage references ($v_{diff,d}$ and $v_{diff,q}$) are finally obtained from PI controllers with feedforward coupling compensation. In general, CCSC suppresses the second harmonic current in differential current (circulating current), but it does not perform the average SM energy regulation, and upper/lower arm energy balancing [146].



FIGURE 3.10: Circulating current controller: (a) CCSC and (b) FCCC.

Different from CCSC, FCCC is based on direct differential current control considering the average SM energy and energy exchange between the upper and lower arms in each phase. The first input in Fig. 3.10(b) defines the DC component ensuring the phase-leg energized, the second input maintains the average SM energy via a PI controller, and the third input balances the upper and lower arm energy in each phase by adding a fundamental frequency component. The differential current reference (i_{diff}^*) obtaining from the sum of three inputs is fed into the current controller composed of PI controller and resonant (R) controllers tuned at fundamental frequency and other even order harmonic frequencies. In addition to the DC component, the first input can also contain AC components with different even order harmonics for reducing the SM capacitor voltage ripples, while zero-sequence voltage injection can perform the same function by attenuating low frequency ripples in the capacitor voltages [147].

3.4 Alternate Arm Converter (AAC)

AAC belongs to the family of modular multilevel converters which combines the characteristics of the two-level converter and MMC in a modular topology [22–36]. Unlike the typical HBSM-based MMC, the AAC requires bipolar SMs, preferably FBSMs and provides DC-fault tolerant capability [37–41]. The alternate arm operation of the AAC leads to the decreased number of SMs in each arm and reduced total volume and losses [42–54]. Therefore, the AAC is a promising topology in the HVDC systems and future DC grids. The following sections describe the basic topology, operating principle, control schemes, and the extended overlap for the AAC.

3.4.1 Basic Topology and Operating Principles of the AAC

Fig. 3.11 shows the schematic of a three-phase AAC. Each arm of the AAC contains N series connected SMs (waveshaping circuit), one arm inductor (L), and one direct switch (DS) composed of IGBTs in series. Moreover, the AAC has reduced energy requirements compared to an MMC with similar voltage and power ratings but requires partial DC filtering due to the six-pulse harmonic in the DC-side. The extended overlap AAC (EO-AAC) offers reduced filtering requirements in the DC side, compared with short overlap AAC (SO-AAC), at the cost of increased number of devices, hence higher losses [49].



FIGURE 3.11: Three-phase topology of the AAC.

AAC can achieve over-modulation operation ($m_a>1$), and the two arms in each phase alternatively generate the AC output voltage and conduct the total AC current:

$$v_a = \frac{m_a V_{dc}}{2} \cos(\omega t), \tag{3.24}$$

$$i_a = \hat{I}_a \cos(\omega t + \phi). \tag{3.25}$$

The waveforms of duty cycles in the two arms of phase a are shown in Fig. 3.12 and the duty cycles can be expressed as:

$$d_{ua} = \frac{1 - v_{am}}{\hat{m}_a}, \qquad \text{and} \tag{3.26}$$

$$d_{la} = \frac{1 + v_{am}}{\hat{m}_a},$$
 (3.27)

where \hat{m}_a is the maximum modulation index for the AAC. The total voltage in each arm of the AAC is lower than the full DC voltage due to the alternate arm operation, hence the number of SMs each arm for the AAC is:

$$N_{SM} = \frac{V_a}{V_c} = \frac{\hat{m}_a V_{dc}}{2V_c}.$$
(3.28)

In addition, the peak voltage stress of DSs is the half of DC voltage as:

$$\hat{V}_{DS} = \frac{V_{dc}}{2},\tag{3.29}$$



FIGURE 3.12: Operation of the AAC arms.

thus the number of devices in each DS per arm is:

$$N_{DS} = \frac{\hat{V}_{DS}}{V_c} = \frac{N_{SM}}{\hat{m}_a}.$$
 (3.30)

3.4.2 Control Scheme of the AAC

3.4.2.1 Control Hierarchy

The control hierarchy of the AAC also consists of four main layers as shown in Fig. 3.13. The function of first three layers (AC/DC grid control, coordinated system control and converter station control) of the AAC is similar as the MMC which has been described in Section 3.3.2.1. The internal converter control (low-level control) of the AAC regulates the i) SM sorting, ii) SM energy, iii) zero-current switching, iv) circulating current and v) overlap period.

The switching signals of DSs and the required number of redunduant SMs are determined by the overlap period and circulating current controller, respectively. In addition to the aforementioned control, it is significant to achieve the zero current switching (ZCS) in the AAC for reducing the voltage stress of DSs, and on-load tap changer (OLTC) coordination can realize the region extension of ZCS [33].



FIGURE 3.13: Control hierarchy of the AAC.

3.4.2.2 Energy Balancing Strategy

The inherent energy balancing point (standard operating point) for the AAC is called the "sweet-spot" ($m_a = M_a = 4/\pi$) that is determined by the zero net energy (3.33) from the energy exchange between AC side (3.31) and DC side (3.32).

$$E_{ac} = \int_{-T/4}^{T/4} \frac{m_a V_{dc} \hat{I}_a}{2} \cos(\omega t) \cos(\omega t + \phi) dt,$$
 (3.31)

$$E_{dc} = \int_{-T/4}^{T/4} \frac{V_{dc}\hat{I}_a}{2} \cos(\omega t + \phi)dt,$$
(3.32)

$$\Delta E_{arm} = E_{ac} - E_{dc} = \frac{V_{dc} \hat{I}_a T \cos \phi}{8} (m_a - \frac{4}{\pi}).$$
(3.33)

In practical applications, the AAC could operate away from the sweet-spot causing energy surplus or deficit ($\Delta E_{arm} \neq 0$) within arms. The reported energy balancing methods in the literature are based on zero sequence current injection [37], overlap onset control (OOC) [40], overlap period [24] and circulating current gradient [30].

As described in Section 3.3.2.4, the circulating current is always present in the MMC which can be used to balance the arm energy. The six DSs in the AAC operate at foundamental frequency and two DSs in each phase-leg switch on/off alternatively, hence no circulating current presents in the AAC. A typical solution is to introduce an overlap period that the upper and lower DSs switch on together for mimicking the MMC operation enabling circulating current to balance arm energy. The overlap period and circulating current control is discussed in later sections.

3.4.2.3 Modulation and Direct Switch (DS) Control

Fig. 3.14(a) shows the schematic diagram of the modulation stage in the AAC. The AAC and MMC have similar topology structure, hence the commonly used modulation techniques in the MMC (Chapter 3.3.2.2) can also be applied in the AAC without modification. The modulation stage only defines the total number of SMs to be inserted in each arm if with the modulation techniques of CB-PWM and staircase. Hence, the detailed switching signals of SMs in the AAC are generated by the subsequent sorting and selecting algorithm.

Due to the introduction of DSs in the AAC, it is also necessary to determine the switching signals of all DSs. Upper and lower DSs in two arms of a phase-leg conduct half of cycle, respectively. Since the DSs only decide the alternate operation of six arms, comparators are used to generate the switching signals of DSs as Fig. 3.14(b) [41].


FIGURE 3.14: Modulation stage and direct switch control of the AAC: (a) modulation stage of the AAC and (b) direct switch control of the AAC.

3.4.2.4 Overlap Period Control

For achieving the arm energy balance in the AAC, an overlap period (t_{ov}) is introduced around the zero-crossing points of the modulation waveform for mimicking the MMC operation as shown in Fig. 3.15. The circulating current presents in the AAC during the overlap period, and the duration of overlap period defines the amount of energy exchanged by the circulating current. Moreover, the magnitude and direction of circulating current also influences such energy exchange [30].



FIGURE 3.15: Overlap period configuration of the AAC: (a) conventional configuration and (b) configuration considering arm redundant voltage.

Fig. 3.15(a) and (b) show the conventional overlap period configuration of the AAC and configuration considering arm redundant voltage (V_r), respectively. For avoiding the output voltage distortion, the maximum achievable overlap period is defined as (3.34) considering the redundant arm voltage (3.35).

$$\hat{t}_{ov} = \frac{\pi - 2\cos^{-1}(\frac{M_a - 1}{m_a})}{\omega},$$
(3.34)

$$V_r = NV_c - \frac{V_{dc}}{2}.$$
 (3.35)

The overlap period for the AAC can be divided into "short-overlap" period and "extended overlap" period. The short overlap angle is typically in the range of 15° to 18° , which can be achieved by symmetrical- and asymmetrical-based fixed and variable overlap period [23, 43]. The extended overlap angle of the AAC is 60° , and the basic operation principle is discussed in Section 3.4.3.

3.4.2.5 Circulating Current Control

Section 3.3.2.4 has introduced the basic circulating current control of the MMC. The AAC mimics the MMC operation during the overlap period, hence the upper and lower arm currents in three phase legs are expressed as:

$$i_{uk} = \frac{i_k}{2} + i_{circk}, \qquad \text{and} \tag{3.36}$$

$$i_{lk} = \frac{i_k}{2} - i_{circk},\tag{3.37}$$

In addition, the arm inductance determines the circulating current control dynamics of the AAC:

$$\frac{d}{dt}i_{circk} = \frac{1}{L}v_{diffk}.$$
(3.38)

The AAC has smaller arm inductance compared with the MMC that leads to fast rates of changes in circulating currents and arm currents.

The typical circulating current control methods for the AAC in current literature are classical PI control [38, 39], hysteresis control [23, 35] and gradient-based control [30].

3.4.2.6 Zero-Current Switch (ZCS) of the DSs

The alternate arm operation in the AAC leads to current interruption of arm inductors which will increase the voltage stress on the DSs. Moreover, the hard-switching of DSs also increases the switching losses of series-connected IGBTs in DSs. Therefore, ZCS should be ensured that forces the arm currents to zero before the DS is opened.

The LC resonant circuit composed of arm inductor and inserted SM capacitors during a fixed overlap period can achieve ZCS by generating a negative voltage across arm inductor [36]. Moreover, the double-band hysteresis current control based on asymmetric overlap period provides soft-switching for DSs without distorting current in the DC-link [23]. Also, the use of snubber circuits can reduce the voltage stress on DSs [34] and OLTC coordination can extend the ZCS region via changing tap positions of transformer [33].

3.4.3 Extended Overlap AAC (EO-AAC)

The principle of the EO-AAC aims to extend the overlap angle to 60° providing a full-time conduction path for DC current in one phase [44–46, 48]. Fig. 3.16 shows the conductions states of six arms in the AAC. There are six conduction states in one period describes as below:

- Conduction state 1: DS_{ua} , DS_{la} , DS_{uc} and DS_{lb} switch on $\left(-\frac{2\pi}{3} \le \omega t \le -\frac{\pi}{3}\right)$
- Conduction state 2: DS_{uc} , DS_{lc} , DS_{ua} and DS_{lb} switch on $(-\frac{\pi}{3} \le \omega t \le 0)$
- Conduction state 3: DS_{ub} , DS_{lb} , DS_{ua} and DS_{lc} switch on $(0 \le \omega t \le \frac{\pi}{2})$
- Conduction state 4: DS_{ua} , DS_{la} , DS_{ub} and DS_{lc} switch on $(\frac{\pi}{3} \le \omega t \le \frac{2\pi}{3})$
- Conduction state 5: DS_{uc} , DS_{lc} , DS_{ub} and DS_{la} switch on $(\frac{2\pi}{3} \le \omega t \le \pi)$
- Conduction state 6: DS_{ub} , DS_{lb} , DS_{uc} and DS_{la} switch on $(\pi \le \omega t \le \frac{4\pi}{3})$



FIGURE 3.16: Extended overlap period configuration of the AAC.

Therefore, the AC currents can always circulate through upper and lower arms in one phase resulting in decoupling of DC and AC currents, and they are cancelled within converter. This also implies that no six-pulse ripples appear in the DC-link, hence EO-AAC does not need to install DC filters. Another significant feature for EO-AAC is the elimination of energy balancing point (removal of sweet spot) because there is always a energy exchange path for DC and AC energy from DC-link to two arms in one phase, hence a wider AC voltage magnitude [45].

However, each arm for EO-AAC has to match the AC voltage during a longer time (more than half-cycle), hence more SMs are required. Although the zero sequence voltage injection can reduce the number of SMs by lowering the maximal voltage in each arm that SMs have to generate, the voltage rating of DSs (required number of IGBTs) will be increased. Another issue needs to be overcome for EO-AAC is the ZCS and active filtering cannot be achieved simultaneously as the circulating current can only control either of them. Detailed description of EO-AAC can be found in [44, 45].

3.5 Conclusion

The LCC is a conventional and mature HVDC technology which is suitable for bulk power transmission, while it has restricted power reversal capability and needs reactive power compensation. The VSC especially the modular VSC overcomes the drawbacks of LCC and has been commercially available. Although the state-of-the-art MMC and the emerging AAC all belong to the modular VSC topology, the typical MMC with HBSMs lacks the vital DC-fault tolerant capability. The emerging AAC presents DC-fault tolerant capability due to the use of bipolar SMs as FBSMs, hence the AAC is a potential candidate for MTDC systems and future DC grids.

The conventional LCC, state-of-the-art MMC and emerging AAC are the three basic topologies used in Chapter 4 to 7 for the verification of proposed generalized expression of DC power flow analysis and the development of different hybrid HVDC systems. The topologies, operating principles and control schemes of the three converters are described in detail with comprehensive theory analysis in this chapter. Besides, the design principle of AC/DC filters for the LCC and the extended overlap operation of the AAC are discussed as well.

Chapter 4

Modeling Approaches and Power Flow Analysis of HVDC systems

4.1 Introduction

This chapter first focuses on the available modeling approaches of HVDC systems including the EMT modeling, electromechanical transient modeling, dynamic frequency capturing modeling and the combination of different modeling methods. EMT modeling is the main focus in this chapter, which is analyzed from the aspects of different analysis methods in simulation tools, numerical integration theory, and possible parallel computation method. Detailed droop control theory for HVDC systems is also described which includes single P/V droop control, single I/V droop control and mixed P/V, I/V droop control. Based on the description of droop control, a generalized expression of DC power flow is proposed considering the initial power flow for normal operation and power flow after converter outage. A detailed equivalent model of a four terminal MMC-based DC grid in PLECS-Blockset and Simulink further verifies the accuracy of proposed generalized expression of DC power flow.

4.2 Modeling Approaches for HVDC Systems

The various digital simulation tools play significant role in the study of time-domain transient characteristics of HVDC systems [152]. The widely-adopted time-simulation methods can be classified into two major categories:

i) electromagnetic transient (EMT) simulation [153, 154] and

ii) electromechanical transient simulation [155, 156].

In addition to the two modeling methods, some other modeling approaches can also be used:

- i) dynamic phasor (DP) modeling method based on time-varying Fourier transform [136, 157–160],
- ii) shifted frequency phasor (SFP) modeling method based on frequency shift of complex signal [161–166], and
- iii) co-simulation method combining different modeling approaches [167–171].

This section provides a detailed introduction of these modeling approaches for HVDC systems.

4.2.1 Electromagnetic Transient (EMT) Modeling

EMT simulation is a convenient method to study the dynamic behaviours of HVDC systems accurately, which can precisely capture the switching actions of different semiconductors due to the much smaller time step (usually 20-50 μ s or even less). The modeling method of EMT simulation is based on the differential equations to describe the transient process of the whole AC and DC systems, while with complex computing procedure [153]. This section discusses the analysis methods in EMT programs, based on the numerical integration theory establishing discrete models. In addition, the general parallel computation method for HVDC systems is also summarized in this section.

4.2.1.1 Analysis Methods in EMT Programs

Three basic analysis methods can be employed for current in industry-grade EMT programs tailoring for both offline and real-time analyzes:

- i) nodal analysis method represented by offline-simulation-based PSCAD/EMTDC [172], real-time simulation-based RSCAD [173] and HYPERSIM [174],
- ii) state-space method represented by offline programs as MATLAB/Simulink [175] and Plecs [102], and
- iii) state-space nodal (SSN) method typified by RT-LAB platform [176].

The process of nodal analysis can be divided into two steps [152]:

1) establishment of companion branch models by discretizing differential equations of dynamic system components via numerical integration.

$$\begin{cases} \frac{dx}{dt} = f(x,t), \\ x(0) = x_0, \end{cases}$$
(4.1)

where x is the system variables that can refer to inductor current and capacitor voltage, and x_0 is the known initial value.

2) combination of the discretized differential equations to create a system of nodal admittance matrix.

$$YV = I \tag{4.2}$$

The commonly used numerical integration method solving differential equations in nodal analysis is implicit trapezoidal integration based on Dommel simulation algorithm [177], which will be discussed in the following section in detail.

The state-space method uses differial equations or differential-algebraic equations (DAEs) in state-space form describing the composite system:

$$\begin{cases} \frac{d\boldsymbol{x}}{dt} = \boldsymbol{A}\boldsymbol{x} + \boldsymbol{B}\boldsymbol{u}, \\ \boldsymbol{y} = \boldsymbol{C}\boldsymbol{x} + \boldsymbol{D}\boldsymbol{u}, \end{cases}$$
(4.3)

where x is the system state vectors refering to inductor current and capacitor voltage, u is the input vectors, y is the output vectors, and the state-space matrices A, B, C and D correspond to the permutation of switches and piecewise linear device segments [178]. In addition, the automated state model generation method separates the composite system to different state-space groups, which can obviously reduce the system formulation time [179]. Variety of numerical integration methods can be used in solving the state-space equations depending on actual requirements. It is noteworthy that the system matrices in state-space method are independent of the simulation step size, hence both the fixed and variable step-size solvers can be adopted, and the variable step-size solvers can reduce simulation time, as different ODE solvers in Simulink [175].

However, massive switching events cannot be handled by the discrete state-space solvers efficiently. The SSN method combines the nodal and state-space methods together and adopts parallel computation with diakoptic analysis to accelerate simulation [178], which has been validated in RT-LAB by ARTEMIS (art 5) algorithm [176]. Additionally, the parallel computation is significant for HVDC systems as the switching states either for

LCC or VSC change many times per duty cycle. The basic solving process is described in Chapter 4.2.1.3.

4.2.1.2 Numerical Integration Theory

Based on the specific EMT process, the system components of HVDC systems can be described by lumped and distributed parameters, which are represented by ordinary differential equations (ODEs) and partial differential equations (PDEs), respectively. In addition, the nodal analysis method-based EMT simulation programs adopt discrete companion models to solve differential equations by numerical integration algorithms.

This section discusses the numerical integration theory in HVDC systems based on the nodal analysis method, while the state-space method is also based on the selection of numerical integration techniques after formulation. The key of discrete companion models aims to simplify the lumped energy storage components and distributed components into a combination of resistance and voltage source (Thevenin equivalent) or inductance and current source (Norton equivalent). In other words, the network equations described by differential equations are converted into the networks of algebraic equations via model discretization [152].

Numerical integration algorithms are typically divided into the explicit integration method and the implicit integration method. From the basic Euler method to improved Euler method (or trapezoidal integration method), Runge-Kutta (RK) method, Simpson method, Adams method, Gears method, etc [180], the selection of a proper integration algorithm for HVDC systems should be considered from the three aspects:

- i) the numerical stability,
- ii) the local truncation error and
- iii) the self-startup characteristic.

In general, the implicit integration method shows better performance in the aspect of numerical stability and local truncation error than explicit integration method, while it requires estimation and iteration due to the lack of self-startup capability [152, 180].

According to the above analysis, the EMT simulation of HVDC systems usually adopts the implicit Euler method and trapezoidal method. Moreover, the implicit Euler method is also the Gear 1 method, and the trapezoidal method belongs to the RK 2 method. The following contents analyze the resistance-inductance (R-L) series branch, inductance branch (Fig 4.1) and capacitance branch based on the two integration methods (Fig 4.2).



FIGURE 4.1: Discrete companion model of R-L series branch and inductance branch.



FIGURE 4.2: Discrete companion model of capacitance branch.

The differential equation of an R-L series branch is:

$$v(t) = Ri(t) + L\frac{di(t)}{dt},$$
(4.4)

and

$$i'(t) = \frac{di(t)}{dt} = \frac{v(t) - Ri(t)}{L}.$$
(4.5)

Employing the implicit Euler method and assuming the step size is h, (4.5) is discretized into:

$$i_{n+1} = i_n + hi'_{n+1} = i_n + \frac{h}{L}(v_{n+1} - Ri_{n+1}),$$
(4.6)

then it can be simplified into:

$$i_{n+1} = \frac{h}{L+hR}v_{n+1} + \frac{L}{L+hR}i_n = G_{ERL}v_{n+1} + J_{ERLn},$$
(4.7)

also,

$$w_{n+1} = \frac{L+hR}{h}i_{n+1} - \frac{L}{h}i_n = R_{ERL}i_{n+1} - E_{ERLn},$$
(4.8)

where G_{ERL} and R_{ERL} is the equivalent conductance and resistance of R-L series branch under explicit Euler method, respectively, and J_{ERLn} and E_{ERLn} is the corresponding equivalent current source and voltage source, respectively. If only consider the inductance branch, eq. (4.7) and (4.8) can be rewitten by eliminating R:

$$i_{n+1} = \frac{h}{L}v_{n+1} + i_n = G_{EL}v_{n+1} + J_{ELn}$$
, and (4.9)

$$v_{n+1} = \frac{L}{h}i_{n+1} - \frac{L}{h}i_n = R_{EL}i_{n+1} - E_{ELn},$$
(4.10)

where G_{EL} and R_{EL} is the equivalent conductance and resistance of inductance branch under explicit Euler method, respectively, and J_{ELn} and E_{ELn} is the corresponding equivalent current source and voltage source, respectively. Employing the trapezoidal method with the same step size, (4.5) now can be discretized into:

$$i_{n+1} = i_n + \frac{h}{2}(i'_n + i'_{n+1}) = i_n + \frac{h}{2L}(v_n - Ri_n + v_{n+1} - Ri_{n+1}),$$
(4.11)

hence,

$$i_{n+1} = \frac{h}{2L + hR} v_{n+1} + \left(\frac{h}{2L + hR} v_n + \frac{2L - hR}{2L + hR} i_n\right) = G_{TRL} v_{n+1} + J_{TRLn},$$
(4.12)

also,

$$v_{n+1} = \frac{2L + hR}{h}i_{n+1} - (v_n + \frac{2L - hR}{h}i_n) = R_{TRL}u_{n+1} - E_{TRLn},$$
(4.13)

where G_{TRL} and R_{TRL} is the equivalent conductance and resistance of R-L series branch under trapezoidal method, respectively, and J_{TRLn} and E_{TRLn} is the corresponding equivalent current source and voltage source, respectively. Similarly, (4.12) and (4.13) can be simplified as (4.14) and (4.15) when only consider the inductance branch.

$$i_{n+1} = \frac{h}{2L}v_{n+1} + (\frac{h}{2L}v_n + i_n) = G_{TL}v_{n+1} + J_{TLn},$$
 and (4.14)

$$v_{n+1} = \frac{2L}{h}i_{n+1} - (v_n + \frac{2L}{h}i_n) = R_{TL}v_{n+1} - E_{TLn},$$
(4.15)

where G_{TL} and R_{TL} is the equivalent conductance and resistance of inductance branch under trapzoidal method, respectively, and J_{TLn} and E_{TLn} is the corresponding equivalent current source and voltage source, respectively.

The EMT simulation of HVDC systems is usually based on the piecewise linearization method handling nonlinear problem of components. Hence, it is paramount to study the boundary point (breaking point) and the initial value at the boundary point. Eq. (4.9) and (4.10) of implicit Euler method show that the equivalent current source (J_{ELn}) and voltage source (E_{ELn}) are only determined by the current value (i_n) , while the equivalent current source (J_{TLn}) (4.14) and voltage source (E_{TLn}) (4.15) in trapezoidal method include both the current value (i_n) and the voltage value (v_n) . At the boundary point, only one value $(i_n \text{ or } v_n)$ can be guaranteed to be continuous, hence the trapezoidal method is not suitable for the first step calculation. Additionally, the equivalent conductance (G_{EL}) in implicit Euler method is twice the value of such conductance (G_{TL}) in trapezoidal method, since the explicit Euler method is based on first-order equation with the local truncation error of $O(h^2)$, while trapezoidal method is based on second-order equation with the local truncation error of $O(h^3)$. Therefore, a feasible scheme for EMT simulation at boundary point is to use implicit Euler method with reduced step size for the first step, then use trapezoidal method for the second step after boundary point.

This conclusion can also be confirmed in the discretized capacitance branch. The differential equation for the capacitance branch is:

$$v_c'(t) = \frac{dv_c(t)}{dt} = \frac{i_c(t)}{C}.$$
(4.16)

Similar as the derivation process for R-L series branch, (4.16) can be discretized into (4.17) and (4.18) by the implicit Euler method.

$$v_{n+1} = v_n + hv'_{n+1} = v_n + \frac{h}{C}i_{n+1} = R_{EC}i_{n+1} + E_{ECn},$$
(4.17)

$$i_{n+1} = \frac{C}{h}v_{n+1} - \frac{C}{h}v_n = G_{EC}v_{n+1} - J_{ECn},$$
(4.18)

where R_{EC} and G_{EC} is the equivalent resistance and conductance of capacitance branch under explicit Euler method, respectively, and E_{EC} and J_{EC} is the corresponding equivalent voltage source and current source, respectively. With trapezoidal method, (4.16) can be expressed as:

$$v_{n+1} = v_n + \frac{h}{2}(v'_n + v'_{n+1}) = v_n + \frac{h}{2C}(i_n + i_{n+1}) = \frac{h}{2C}i_{n+1} + (v_n + \frac{h}{2C}i_n)$$

$$= R_{TC}i_{n+1} + E_{TCn},$$
(4.19)

also,

$$i_{n+1} = \frac{2C}{h}v_{n+1} - (i_n + \frac{2C}{h}v_n) = G_{TC}v_{n+1} - J_{TCn},$$
(4.20)

where R_{TC} and G_{TC} is the equivalent resistance and conductance of capacitance branch under trapezoidal method, respectively, and E_{TC} and J_{TC} is the corresponding equivalent voltage source and current source, respectively.

4.2.1.3 Parallel Computation Method

This section focuses on the parallel computation method of EMT fast simulation for avoiding cumbersome computations in HVDC systems. Since the network equations have to be re-established with every change of switching states, hence the parallel computation with switching analysis is of importance to improve computational efficiency. The parallel computation method aims to divide the whole system to different subdivisions which will be calculated independently and encapsulated in interfaces [154, 181].

Based on the analysis methods in EMT programs (Chapter 4.2.1.1), the parallel computation can use either nodal analysis method or state-space method for each subdivision. The basic process can be divided into three steps:

- 1) discretize the equations of subdivisions by nodal analysis method or state-space method obtaining equivalent branches,
- 2) replace the original subdivisions by the equivalent branches acquiring the whole equivalent network, and
- 3) compute the acquired equivalent network and obtain all parameters in the original network.

The aforementioned SSN method of Chapter 4.2.1.1 is based on the basic process, which uses the state-space method solving the equations of subdivisions and nodal analysis is adopted for interface computation.

Fig. 4.3 shows the parallel computation method in LCC and MMC-based HVDC systems. The whole system can be subdivided into different units mainly including AC networks, converters and DC lines. In addition, the MMC requires to establish discrete models for SM units. The HBSMs are the commonly used SMs for MMC, the next contents in this section introduces the discrete model of single HBSM and further derive the arm equivalent model based on the nodal analysis mentioned in Chapter 4.2.1.2 and the modeling method of full detailed equivalent model (Type 4) (Chapter 2.3).

Eq. (4.19) describes the discrete model of capacitance branch, which can be used in HBSM model discretization as:

$$v_{c}(t) = v_{c}(t-h) + \frac{h}{2C}(i_{c}(t-h) + i_{c}(t))$$

= $\frac{h}{2C}i_{c}(t) + (v_{c}(t-h) + \frac{h}{2C}i_{c}(t-h))$
= $R_{c}i_{c}(t) + E_{eq,c}(t-h),$ (4.21)



FIGURE 4.3: Parallel computation method of LCC and MMC.

where R_c is the equivalent resistance of HBSM capacitor, and $v_{ceq}(t - h)$ is the history equivalent voltage source of HBSM capacitor with the step size h. If use variable resistors (R_1, R_2) to replace the IGBTs with different switching states, the equivalent model of each HBSM can be established as Fig. 4.4, and the equation of such model follows:

$$v_{sm}(t) = R_{eq,sm}i_{sm}(t) + E_{eq,sm}(t-h)$$

$$= \frac{R_2(R_1 + R_c)}{R_1 + R_2 + R_c}i_{sm}(t) + \frac{R_2}{R_1 + R_2 + R_c}E_{eq,c}(t-h).$$
(4.22)

Hence, the arm instantaneous voltage can be expressed as:

$$v_{arm}(t) = \sum_{i=1}^{N} v_{sm}^{i}(t) = (\sum_{i=1}^{N} R_{eq,sm}^{i})i_{arm}(t) + \sum_{i=1}^{N} E_{eq,sm}^{i}(t-h)$$

$$= R_{eq,arm}i_{arm}(t) + E_{eq,arm}(t-h).$$
(4.23)



FIGURE 4.4: Discrete model of HBSMs and equivalent arm structure.

4.2.2 Electromechanical Transient Modeling

Since the extensive use of semiconductor devices with high switching frequency in HVDC systems, EMT simulation cannot be appplied to the study of large-scale HVDC systems, especially the interaction of AC and DC systems, although many equivalent models can be used to reduce the EMT simulation time [114, 154, 182–186]. However, electromechanical transient simulation ignores the harmonic influence and only considers the system operation and control characteristics under foundamental frequency. Hence, electromechanical transient simulation is more suitable for the study of large-scale HVDC systems. In addition, a balanced system is assumed at all times in the electromechanical transient simulation and the whole system can be described by the positive-sequence component under foundamental frequency, thus such modeling approach cannot reflect the system inner characteristics [155, 156, 187].

The electromechinal transient model is also called root mean square (RMS) model or phasor model as it ignores the system high frequency components. From the aspect of large-scale AC/DC grid, the purpose of electromechanical transient modeling for HVDC systems is intended to tackle two main issues:

- i) the power flow calculation of DC and AC systems, and
- ii) the mathematical description about steady-state and transient response characteristics of HVDC systems.

During the process of electromechanical transient modeling, the AC and DC systems are solved independently, where the AC systems are represented by phasor under foundamental frequency, the DC systems can be regarded as a variable source or load for AC systems and emulated by a branch with variable power (Fig. 4.5). Moreover, the DC systems can be solved if the AC bus voltages and converter control schemes are confirmed [156].



FIGURE 4.5: Equivalent function of DC systems in electromechanical modeling.

The general modeling approach for electromechanical transient characteristics of HVDC systems aims at formulating corresponding algebraic-differential equations. In addition, the unified modeling framework can be separated into three main steps:

- 1) converter modeling including the equivalent models in AC and DC sides,
- 2) controller modeling including the high- and low-level control, and
- 3) DC network modeling with different equivalent line models.

The initialization of electromechnical transient model is from the detailed system power flow calculation, which can be divided into sequential AC/DC power flow [188] and unified AC/DC power flow [189]; the unified method is easy to realize in simulators [156]. After detailed power flow calculations, the dynamic simulation can be proceeded with the assignment of initial values.

4.2.3 Other Modeling Approaches and Co-Simulation Studies

The EMT modeling approach can study the detailed switching characteristics due to its ultra-wide frequency-band capturing capability, while the foundamental frequency-based electromechanical transient modeling approach can be used in overall system stability analysis. In addition, some other modeling approaches that can capture dynamic frequency are developed for avoiding time-consuming calculation while retaining simulation accuracy. This section introduces two main techniques in the time-domain, DP model and SFP model, for capturing wide frequency-band interactions between different components in large-scale AC/DC grids. Besides, this section also discusses the potential of different modeling appraches for co-simulation study.

4.2.3.1 Dynamic Frequency Capturing Modeling

The DP modeling approach uses state-space equations to describe electrical variables (e.g. voltage or current) in one period T_s (sliding window), which are decomposed into harmonics of different frequencies based on Fourier transform, hence any high frequency components can be contained depending on the analysis requirements (Fig. 4.6). In addition, the corresponding Fourier series (also called DPs) are changed with the move of sliding window [157].



FIGURE 4.6: Time-varing Fourier transform in DP modeling approach.

A periodic variable $x(\tau), (\tau \in (t - T_s, T_s])$ with the period T_s can be expressed in the time-varying Fourier series by:

$$x(\tau) = \sum_{k=-\infty}^{\infty} \langle x \rangle_k(t) e^{j\omega_s \tau},$$
(4.24)

where $\langle x \rangle_k(t)$ is the k^{th} Fourier series of $x(\tau)$, and $\omega_s = 2\pi/T_s$. The k^{th} Fourier series $\langle x \rangle_k(t)$ is determined by:

$$\langle x \rangle_k(t) = \frac{1}{T_s} \int_{t-\tau}^t x(\tau) e^{-j\omega_s \tau} d\tau.$$
(4.25)

The features of product and differential for $\langle x \rangle_k(t)$ are provided as:

$$\langle xy \rangle_k = \sum_i \langle x \rangle_{k-i} \langle y \rangle_i, \quad \text{and}$$
 (4.26)

$$\frac{d\langle x\rangle_k(t)}{dt} = \langle \frac{dx}{dt} \rangle_k(t) - j\omega_s \langle x \rangle_k(t).$$
(4.27)

Applying (4.27) to inductance and capacitance branch, the corresponding DP equations are:

$$L\frac{d\langle i_L\rangle_k(t)}{dt} = \langle v_L\rangle_k(t) - j\omega_s L\langle i_L\rangle_k(t), \quad \text{and} \quad (4.28)$$

$$C\frac{d\langle v_c\rangle_k(t)}{dt} = \langle i_c\rangle_k(t) - j\omega_s C\langle u_c\rangle_k(t).$$
(4.29)

In general, the DP modeling approach can be used in the situation of analyzing specific frequency range. It is noted that the simulation accuracy will not be sacrificed within the selected frequencies.



FIGURE 4.7: Spectra of original signal and complex signal: (a) original signal and (b) complex signal.

The nodal analysis-based SFP modeling approach uses the bandpass characteristics of electrical variables around the center frequency to construct a complex signal. A bandpass signal x(t) with a bilateral spectrum is assumed to be with a center frequency ω_c (Fig. 4.7(a)). For obtaining the signal only with positive frequency (single-side spectrum), introduce a new complex signal $x_S(t)$ (Fig. 4.7(b)), and it can be expressed as:

$$x_S(t) = x(t) + jx_T(t),$$
 (4.30)

where $x_T(t)$ is the transform of x(t) by specific methods, as Hilbert transform $x_T(t) = x_H(t) = \int_{-\infty}^{\infty} \frac{x(\tau)}{\pi(t-\tau)} d\tau$ [161], differential transform $x_T(t) = x_D(t) = -\frac{x'(t)}{\omega_c}$ [163] and integral transform $x_T(t) = x_I(t) = \int_{-\infty}^t x(\tau) d\tau$ [190]. In addition, the spectra of complex signals $(x_{SH}(t) = x(t) + jx_H(t), x_{SD}(t) = x(t) + jx_D(t), x_{SI}(t) = x(t) + jx_I(t))$ under different transforms can be obtained via Fourier transform, which are $X_{SH}(\omega) = (1 + \text{sgn}(\omega))X(\omega), X_{SD}(\omega) = (1 + \frac{\omega}{\omega_c})X(\omega)$ and $X_{SI}(\omega) = (1 + \frac{\omega_c}{\omega})X(\omega)$, respectively. It is noted that Hilter transform has no negative frequency components compared with other two methods (Fig. 4.8).

To reconstruct (4.30) by frequency shift of ω_s , (4.31) can be deduced. If the shift frequency is set to zero ($\omega_s = 0$), the SFP model will be equal to conventional EMT model, while the complex signal will convert to complex envelope signal if the shift frequency is set to the center frequency ($\omega_s = -\omega_c$ where the minus refers to left shift) as (4.32).

$$x_{SS}(t) = x_S(t)e^{j\omega_S t},\tag{4.31}$$

$$x_E(t) = x_S(t)e^{-j\omega_c t}.$$
 (4.32)



FIGURE 4.8: Spectra of complex signal under different transforms.

Fig. 4.9 shows the maximum frequency of complex envolope signal is far less than the original signal. Therefore, based on Shannon sampling theorem, SPF modeling approach can provide larger time step without sacrificing simulation accuracy [161].



FIGURE 4.9: Spectra of complex signal and complex envolope signal: (a) complex signal and (b) complex envolope signal.

Based on the description of numerical integration in Chapter 4.2.1.2, the discrete companion model of system components can be established. Assuming $\frac{dx_S(t)}{dt} = f_S(t)$, hence the differential equation of complex envolope signal is:

$$\frac{dx_E(t)}{dt} = f_E(t) - j\omega_s x_E(t), \qquad (4.33)$$

where $f_E(t) = f_S(t)e^{-j\omega_s t}$. Employing traprzoidal integration method to discretize (4.33):

$$x_E(t) = x_E(t-h) + \frac{h}{2}((f_E(t) - j\omega_s x_E(t)) + (f_E(t-h) - j\omega_s x_E(t-h))), \quad (4.34)$$

Moreover, combining (4.31) and (4.34), the discrete equation of complex signal can be further acquired as:

$$x_{S}(t) = x_{S}(t-h)e^{j\omega_{s}h} + \frac{h}{2}((f_{S}(t) - j\omega_{s}x_{S}(t)) + (f_{S}(t-h)e^{j\omega_{s}h} - j\omega_{s}x_{S}(t-h)e^{j\omega_{s}h})).$$
(4.35)

4.2.3.2 Co-simulation Studies

The purpose of co-simulation studies in HVDC systems is to achieve a balance between the modeling scale and accuracy. Based on the aforementioned different modeling approaches, a typical co-simulation study is that AC systems are modeled by EMT modeling approach and electromechanical transient method is used for converter modeling [167, 168]. By interfacing the two basic modeling approaches, the computation speed can be increased with acceptable simulation accuracy [168]. Additionally, the EMT modeling method has been used to combine DP or (and) SFP approaches for specific study requirements in current literature [169–171].

4.3 Detailed Description of Droop Control

In LCC-based MTDC systems, the DC voltage is usually controlled by one converter for improving the system operation stability, thus the balance of DC current and power has to be coordinated by the upper layer control [191]. Unlike LCC-based MTDC systems, there are three basic control schemes in system level for VSC-based MTDC systems and DC grids [192–194]:

- i) master/slave control,
- ii) DC voltage margin control, and
- iii) droop control.

In the first two schemes, one converter has to act as a slack bus controlling the DC voltage of the whole system, and other converters adopt constant AC active power (DC power) or DC current control [195]. The DC voltage margin control requires no communication compared to the master-slave control in a fixed operation range, while it requires relatively complex controller design, especially the two stage DC voltage control method [193]. By comparison, the droop control is easy to implement since all converters in an MTDC system coordinate to balance the DC voltage by given droop constants, hence operate independent of communication [195]. Power/voltage (P/V) droop control is a conventional droop control scheme, which can be extended to DC current/voltage (I/V) droop control and mixed droop control combining them for different converters. This section discusses the single P/V, I/V droop and mixed droop control for HVDC systems.

4.3.1 Basic Theory

The conventional P/V droop control for MTDC systems or DC grids is originated from the power/frequency (P/f) droop control in AC systems [194]. For P/V droop control, the control of DC voltage is shared by different converters with droop control, and the droop control can be considered as an extension of constant power or DC voltage control, which can improve the system reliability and reduce the telecommunication dependence among all converters as well. The function of I/V droop control is similar as P/V droop control that the DC current is controlled rather than power. In addition, it is also possible to arrange P/V and I/V droop control in different converters constituting mixed droop control.

4.3.2 Static Characteristic

The static characteristic and controller structure of droop control are shown in Fig. 4.10 where a three terminal HVDC system is adopted as an example. Converter 1 in rectifier side delivers power to converter 2 and 3 in inverter side, and DC voltage is balanced by the three converters via corresponding droop characteristic. Matrix K_{droop} represents the droop constants of all converters in an HVDC system, and the ratio of characteristic curve is defined as $R = -(K_{droop})^{-1}$. It is noted that the power limitation is a concern for each converter, since the converter will lose DC voltage balance capability if it reaches the maximum power rating after system power disturbance. When an HVDC system with droop control is in steady-state operation, eq. (4.36) and (4.37) can be established for P/V droop control and I/V droop control, respectively.

$$(P_{dc}^{*} - P_{dc}) + \text{diag}(K_{droop_{P}})(V_{dc}^{*} - V_{dc}) = 0, (P_{dc0}^{*} - P_{dc0}) + \text{diag}(K_{droop_{P}})(V_{dc0}^{*} - V_{dc0}) = 0,$$
(4.36)

$$\begin{cases} (I_{dc}^{*} - I_{dc}) + \operatorname{diag}(K_{droop_{I}})(V_{dc}^{*} - V_{dc}) = \mathbf{0}, \\ (I_{dc0}^{*} - I_{dc0}) + \operatorname{diag}(K_{droop_{I}})(V_{dc0}^{*} - V_{dc0}) = \mathbf{0}, \end{cases}$$
(4.37)

where the superscript * and subscript $_0$ refer to corresponding reference values and initial values, respectively. In addition, the droop constants for P/V and I/V droop control in this thesis are defined as (4.38) and (4.39), respectively .

$$K_{droop_P} = \frac{P_{dc_{rated}}}{V_{dc_{rated}}\delta_{droop}}$$
(MW/kV), (4.38)



FIGURE 4.10: Static characteristic of droop control in HVDC systems: (a) single P/V droop control, (b) single I/V droop control, (c) mixed P/V and I/V droop control, and (d) structure of droop controller.

$$K_{droop_{I}} = \frac{P_{dc_{rated}}}{V_{dc_{rated}}^{2} \delta_{droop}}$$
(A/kV), (4.39)

where $P_{dc_{rated}}$ and $V_{dc_{rated}}$ is the rated DC power and rated DC voltage of each converter, respectively, and δ_{droop} in this thesis refers to the DC voltage deviation ratio. It is noted that the converter will run in constant DC power (DC current) control or DC voltage control if $K_{droop} = 0$ or $K_{droop} = \infty$, respectively.

4.4 Generalized Expression of DC Power Flow

Power flow analysis is important for HVDC systems as it is used for static security assessment. Different from HVAC systems, the DC power flow analysis has to be conducted in HVDC systems [188, 189], which should consider the effect of single P/V or I/V droop control. The initial DC power flow can be obtained by different power flow calculation methods as Newton Raphson [196]. Moreover, the DC power flow under reference change can be determined by the droop characteristic of different converters [192, 195]. However, the detailed DC power flow analysis under mixed P/V and I/V droop control is not reported in current literature.

Based on the description of droop control in Chapter 4.3, this section first introduces the basic single droop control-based DC power flow calculation. Furthermore, it is expanded to a generalized calculation method obtaining the initial DC power flow for normal operation and the DC power flow after converter outage under mixed P/V and I/V droop control. The power limitation of all remaining converters after a converter outage is also considered for avoiding long-term overload operation. The accuracy of proposed generalized expression of DC power flow is verified in an MMC-based four terminal DC grid.

4.4.1 Initial DC Power Flow for Normal Operation

4.4.1.1 Single P/V or I/V Droop Control

An MTDC system or a DC grid of n buses under P/V droop control is assumed to be with one known DC voltage reference, and n - 1 known DC power references. The initial power flow can be solved by Newton-Raphson method as (4.40), obtaining one unknown DC power and n - 1 DC voltages (n unknowns) [192]. In addition, the DC currents of all terminals and branches can also be derived.

$$P_{dc0_P} = V_{dc0_P} \odot (Y_{dc} V_{dc0_P}), \tag{4.40}$$

where P_{dc0_P} and V_{dc0_P} refer to the column vectors of terminal DC powers and DC voltages under P/V droop control, respectively, Y_{dc} is the nodal admittance matrix and \odot refers to the Hadamard product.

$$\begin{cases} I_{dc0_I} = Y_{dc} V_{dc0_I}, \\ P_{dc0_I} = V_{dc,ini_I} \odot I_{dc0_I}, \end{cases}$$

$$(4.41)$$

where I_{dc0_I} , P_{dc0_I} and V_{dc0_I} refer to the column vectors of terminal DC currents, DC powers and DC voltages under I/V droop control, respectively.

The basic initial DC power flow calculation for single droop control aims to establish the relationship between powers or currents and voltages via nodal admittance matrix, creating a one-to-one correspondence.

4.4.1.2 Mixed P/V and I/V Droop Control

In the scenario of mixed P/V and I/V droop control, there are one known DC voltage reference, *a* known DC power references, and *b* known DC current references. Therefore, it is necessary to build the relationship amongst currents, voltages and powers, creating a one-to-many (two) correspondence and line currents can be as a bridge connecting the three.

First, *b* DC current references (*a* DC power references) are assumed to be *b* DC power references (*a* DC current references), then the current directions in each transmission line can be determined by the single P/V or I/V droop calculation method described in Chapter 4.4.1.1. Based on the known line current directions, the final initial power flow can be derived from (4.42), obtaining n - a DC powers, n - 1 DC voltages, n - b DC terminal currents and k DC line currents (2n + k unknowns).

$$\begin{cases}
P_{dc0_M} = V_{dc0_M} \odot I_{dc0_M} \\
\operatorname{diag}(G_l) V_{l0_M} = I_{l0_M} \\
I_{dc0_M} = \Lambda I_{l0_M}
\end{cases}, \tag{4.42}$$

where P_{dc0_M} and V_{dc0_M} refer to the column vectors of terminal DC powers and DC voltages under mixed droop control, respectively, G_l is a single column matrix composed by k line conductances, V_{l0_M} and I_{l0_M} refer to the corresponding line voltage drop and line current expressed in single column matrix, respectively, and Λ is the coefficient ($n \times k$ matrix) connecting DC currents between terminals and branches and the value depends on the actual system. Fig. 4.11 shows the flowchart of initial power flow calculation for single and mixed droop control. The results of initial DC power flow in Chapter 4.4.1.1 and 4.4.1.2 are used as references of each terminal for single and mixed droop control.



FIGURE 4.11: Flowchart of initial DC power flow calculation.

4.4.2 DC Power Flow after Converter Outage

The impact of converter outage to a MTDC system or DC grid can be considered as a system power disturbance, and it is necessary to re-construct the network equations for obtaining redistributed DC power flow. For single P/V droop control, the deviation of DC power and DC voltage after system disturbance in each terminal can be expressed as:

$$\begin{cases} \Delta P_{dc}^{*} = P_{dc}^{*} - P_{dc0}^{*}, \\ \Delta P_{dc} = P_{dc} - P_{dc0}, \\ \Delta V_{dc}^{*} = V_{dc}^{*} - V_{dc0}^{*}, \\ \Delta V_{dc} = V_{dc} - V_{dc0}. \end{cases}$$
(4.43)

Similarly, the deviation of DC current and DC voltage after system disturbance in each terminal for I/V droop control are:

$$\begin{cases}
\Delta I_{dc}^{*} = I_{dc}^{*} - I_{dc0}^{*}, \\
\Delta I_{dc} = I_{dc} - I_{dc0}, \\
\Delta V_{dc}^{*} = V_{dc}^{*} - V_{dc0}^{*}, \\
\Delta V_{dc} = V_{dc} - V_{dc0}.
\end{cases}$$
(4.44)

Combining (4.36) and (4.43), (4.45) can be obtained for single P/V droop control. For single I/V droop control, eq. (4.46) can also be yielded combining (4.37) and (4.44).

$$\Delta P_{dc} - \Delta P_{dc}^* = \operatorname{diag}(K_{droop_P})(\Delta V_{dc}^* - \Delta V_{dc}), \qquad (4.45)$$

$$\Delta I_{dc} - \Delta I_{dc}^* = \operatorname{diag}(K_{droop_I})(\Delta V_{dc}^* - \Delta V_{dc}).$$
(4.46)

Additionally, the relationship between ΔP_{dc} and ΔV_{dc} for single P/V droop control can be approximately established as:

$$\Delta P_{dc} \approx J_{dc} \Delta V_{dc}, \tag{4.47}$$

where J_{dc} is DC jacobian matrix of certain MTDC system or DC grid. The relationship between ΔI_{dc} and ΔV_{dc} for single I/V droop control is:

$$\Delta I_{dc} = Y_{dc} \Delta V_{dc}. \tag{4.48}$$

The following contents analyze the detailed DC power flow variation after converter outage with single P/V or I/V droop control and mixed P/V and I/V droop control.

4.4.2.1 Single P/V or I/V Droop Control

For determining the re-distributed DC power flow after converter outage with single P/V or I/V droop control, eq. (4.49) and (4.50) are established to obtain the DC power and DC voltage deviation by combining (4.45), (4.47) and (4.46), (4.48), respectively.

$$\begin{cases} \Delta P_{dc} = \frac{\Delta P_{dc}^{*} + \operatorname{diag}(K_{droop_{P}})\Delta V_{dc}^{*}}{I + \operatorname{diag}(K_{droop_{P}})J_{dc}^{-1}}, \\ \Delta V_{dc} = \frac{\Delta P_{dc}^{*} + \operatorname{diag}(K_{droop_{P}})\Delta V_{dc}^{*}}{\operatorname{diag}(K_{droop_{P}}) + J_{dc}}, \\ \begin{cases} \Delta I_{dc} = \frac{\Delta I_{dc}^{*} + \operatorname{diag}(K_{droop_{I}})\Delta V_{dc}^{*}}{I + \operatorname{diag}(K_{droop_{I}})Y_{dc}^{-1}}, \\ \Delta V_{dc} = \frac{\Delta I_{dc}^{*} + \operatorname{diag}(K_{droop_{I}})\Delta V_{dc}^{*}}{\operatorname{diag}(K_{droop_{I}})\Delta V_{dc}^{*}}, \end{cases}$$
(4.49)

where I is the $n \times n$ identity matrix. Supposing no deviation of DC voltage references $(\Delta V_{dc}^* = 0)$, eq. (4.49) and (4.50) can be simplified as:

$$\begin{cases} \Delta P_{dc} = \frac{\Delta P_{dc}^{*}}{I + \operatorname{diag}(K_{droop_{P}})J_{dc}^{-1}} = A_{P}\Delta P_{dc}^{*}, \\ \Delta V_{dc} = \frac{\Delta P_{dc}^{*}}{\operatorname{diag}(K_{droop_{P}}) + J_{dc}} = B_{P}\Delta P_{dc}^{*}, \end{cases}$$
(4.51)

and

$$\begin{cases} \Delta I_{dc} = \frac{\Delta I_{dc}^{*}}{I + \operatorname{diag}(K_{droop_{I}})Y_{dc}^{-1}} = A_{I}\Delta I_{dc}^{*}, \\ \Delta V_{dc} = \frac{\Delta I_{dc}^{*}}{\operatorname{diag}(K_{droop_{I}}) + Y_{dc}} = B_{I}\Delta I_{dc}^{*}, \end{cases}$$
(4.52)

respectively, where A_P , B_P , A_I and B_I refer to the corresponding coefficient matrix. The outage of certain converter in an MTDC system or DC grid, reflecting in (4.51) and (4.52), can be summarized as:

$$\Delta P_{dc,out}^* = -P_{dc0,out},$$

$$K_{droop,out_P} = 0,$$
(4.53)

and

$$\begin{cases} \Delta I_{dc,out}^* = -I_{dc0,out}, \\ K_{droop,out_I} = 0, \end{cases}$$
(4.54)

respectively, then the final DC power flow for single droop control can be obtained.

However, the above description does not consider the power limitation of converters after converter outage. If one converter with P/V droop control is overloaded after preliminary calculation (4.55), the converter should be in constant DC power control in actual operation, and the actual DC power variation of such converter can be calculated as (4.56).

$$P_{dc0,over} + \Delta P_{dc,pre} > P_{dc,max}, \tag{4.55}$$

$$\Delta P_{dc,act_P} = P_{dc,max} - P_{dc0,over}.$$
(4.56)

Setting the droop constant of the oveloaded converter as zero (4.57), then the final DC power flow for single P/V droop control can be accurately derived by substituting (4.53), (4.56) and (4.57) into (4.51).

$$K_{droop,over} = 0 \tag{4.57}$$

For single I/V droop control, if a certain converter is overloaded after converter outage (4.58), such converter in the text is designed to switch into constant DC power control $(K_{droop,over} = 0)$ for avoiding long-term overload operation.

$$(V_{dc0,over} + \Delta V_{dc,pre})(I_{dc0,over} + \Delta I_{dc,pre}) > P_{dc,max},$$
(4.58)

Correspondingly, the single I/V droop control transforms into mixed P/V and I/V droop control, and the remaining maximum power capacity of an overloaded converter is:

$$\Delta P_{dc,act_I} = P_{dc,max} - V_{dc0,over} I_{dc0,over}.$$
(4.59)

It is worth mentioning that the further derived line current directions under single droop control can be used in mixed droop control for building the relationship amongst the DC powers, currents and voltages in the remaining converters. The following section introduces the power flow calculation method for mixed P/V and I/V droop control.

4.4.2.2 Mixed P/V and I/V Droop Control

The current directions of branches in an MTDC system or a DC grid may change after converter outage. Thus, the line current directions should be first obtained based on single droop control assumption as described in the previous section. Then (4.60) can be established yielding ΔP_{dc} , ΔV_{dc} , ΔI_{dc} in each station and ΔI_l in transmission lines. There are in total 3(n - m) + k' unknowns if m converters are isolated, and k' is the remaining number of branches.

$$\begin{cases} P_{dc0_M} + \Delta P_{dc} = (V_{dc0_M} + \Delta V_{dc}) \odot (I_{dc0_M} + \Delta I_{dc}) \\ \operatorname{diag}(G_l)(V_{l0_M} + \Delta V_l) = I_{l0_M} + \Delta I_l, \\ (I_{dc0_M} + \Delta I_{dc}) = \Lambda'(I_{l0_M} + \Delta I_l), \end{cases}$$

$$(4.60)$$

where Λ' is a $(n - m) \times k'$ coefficient matrix here, and G_l is a single column matrix of k' elements. In addition, the references are not changed for the remaining n - m converters, hence eq. (4.45) and (4.46) can be reduced to (4.61) and (4.62), respectively.

$$\Delta P_{dc} = -\text{diag}(K_{droop_P})\Delta V_{dc}$$
(4.61)

$$\Delta I_{dc} = -\text{diag}(K_{droop_I})\Delta V_{dc}$$
(4.62)

If no converter is overloaded, the final DC power flow can be derived by combining (4.61), (4.62) and (4.60).

When one converter with P/V droop control exceeds the maximum power limitation (4.55), (4.56) has to be used to substitute corresponding ΔP_{dc} in (4.61), and droop constant of such converter should be set as zero ($K_{droop,over} = 0$) as well. In addition, if one converter with I/V droop control is overloaded, such converter should switch into constant DC power control, hence (4.61) and (4.62) will be updated. The final accurate DC power flow can be obtained by substituting updated (4.61) and (4.62) into (4.60).

The flowchart of power flow calculation after converter outage for single and mixed droop control is shown in Fig. 4.12. In general, the power flow calculation method for mixed droop control aims to construct the detailed network equations based on actual steady-state operation after converter outage. Since (4.60) in mixed droop control can also be expressed as (4.47) by ignoring the nonlinear terms in single P/V droop control, or (4.48) in single I/V droop control, the calculation methods are equivalent for single and mixed droop control.





4.4.3 Case Study

A detailed equivalent model of MMC-based four-terminal DC grid (Fig. 4.13) is estalished in PLECS-Blockset and Simulink for verifying the accuracy of described generalized expression of DC power flow. The parameters of the four MMCs are listed in Table 4.1, and Table 4.2 provides the parameters of DC transmission lines. In addition, MMC1 is located at rectifier side delivering power to other MMCs (MMC2, MMC3 and MMC4).



FIGURE 4.13: MMC-based four terminal DC grid.

TABLE 4.1: Parameters of four MMCs in the DC grid.

Parameter	MMC1	MMC2	MMC3	MMC4
Rated Power (MVA)	800	400	150	250
DC Voltage (kV)	± 200	± 200	± 200	± 200
AC Voltage (L-L rms) (kV)	380	145	145	145
Transformer Ratio	380/220	145/220	145/220	145/220
Number of SMs per Arm	400	400	400	400
Nominal Operating Point (m_a)	0.9	0.9	0.9	0.9

TABLE 4.2: Parameters of DC transmission lines in the MMC-based DC grid.

Parameters	Line 1	Line 2	Line 3	Line 4
Туре	Cable	Cable	Cable	Cable
Distance (km)	100	150	80	50
Resistance (Ω /km)	0.011	0.011	0.011	0.011
Capacitance (µF/km)	0.2185	0.2185	0.2185	0.2185
Inductance (mH/km)	0.2615	0.2615	0.2615	0.2615

4.4.3.1 Verification in MMC-based DC Grid under Single P/V Droop Control

In this case, a single P/V droop control is adopted in the MMC-based DC grid (Fig. 4.14(a)). Table 4.3 lists the initial references and droop constants of four MMCs for deriving the initial power flow. Based on eq. (4.40), DC power of MMC1 and DC voltages of MMC2, MMC3 and MMC4 can be obtained. Besides, DC currents of all terminals and branches can be further derived (Table 4.4).



FIGURE 4.14: Detailed current directions of MMC-based DC grid under single P/V droop control in the MMC-based DC grid: (a) initial power flow, and (b) power flow after MMC4 outage.

TABLE 4.3: References of outer	control loop	for single	P/V d	lroop	control	in	the	MMC-
based DC grid.								

Parameter	MMC1	MMC2	MMC3	MMC4
P_{dc}^* (MW)	-	-380	-100	-120
I_{dc}^{*} (kA)	-	-	-	-
V_{dc}^{*} (kV)	400	-	-	-
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40 MW/kV	20 MW/kV	7.5 MW/kV	12.5 MW/kV

TABLE 4.4: Calculation of initial DC power flow for single P/V droop control in the MMCbased DC grid.

Parameter	MMC1	MMC2	MMC3	MMC4
P_{dc} (MW)	603.1145	-380.0000	-100.0000	-120.0000
I_{dc} (kA)	1.5078	-0.9551	-0.2511	-0.3016
V_{dc} (kV)	400.0000	397.8845	398.1976	397.8730
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40	20	7.5	12.5
Line currents	I_{l1}	I_{l2}	I_{l3}	I_{l4}
(kA)	0.9616	0.5462	0.0065	0.2951

The obtained voltages and powers are used as references for four MMCs with P/V droop control in simulation. The steady-state results in simulation are shown in Table 4.5, which verifies the accuracy of initial power flow calculation. Additionally, the corresponding DC jacobian matrix (J_{dc}) of the MMC-based DC grid can also be obtained as:

$$\boldsymbol{J_{dc}} = \begin{bmatrix} 304.5381 & -181.8182 & -121.2121 & 0\\ -180.8566 & 405.9723 & 0 & -226.0707\\ -120.6659 & 0 & 482.4126 & -361.9978\\ 0 & -226.0642 & -361.7027 & 587.4653 \end{bmatrix}$$
(4.63)

TABLE 4.5: Simulation results of initial DC power flow for single P/V droop control in the MMC-based DC grid.

Parameter	MMC1	MMC2	MMC3	MMC4
P_{dc} (MW)	603.1145	-379.9999	-100.0000	-120.0005
I_{dc} (kA)	1.5078	-0.9551	-0.2511	-0.3016
V_{dc} (kV)	400.0000	397.8835	398.1976	397.8730
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40	20	7.5	12.5
Line currents	I_{l1}	I_{l2}	I_{l3}	I_{l4}
(kA)	0.9616	0.5462	0.0065	0.2951

When MMC4 is isolated due to outage ($\Delta P_{dc,MMC4}^* = 120 \text{ MW}$, $K_{droop,MMC4} = 0$), the power flow is re-distributed in the remaining three MMCs. As described in Chapter 4.4.2.1, the first step is to conduct preliminary power flow calculation deriving ΔP_{dc} and ΔV_{dc} based on (4.51). The matrix of droop constants in four MMCs (K_{droop_P}) is [40, 20, 7.5, 0]^T under MMC4 outage, hence the resulting coefficient matrices ($A_P^{(1)}$, $B_P^{(1)}$) are:

$$\boldsymbol{A}_{\boldsymbol{P}}^{(1)} = \begin{bmatrix} 0.3895 & -0.5617 & -0.5699 & -0.5673 \\ -0.2794 & 0.6679 & -0.2999 & -0.3126 \\ -0.1064 & -0.1126 & 0.8650 & -0.1265 \\ 0.0000 & 0.0000 & 0.0000 & 1.0000 \end{bmatrix}, \text{ and } (4.64)$$
$$\boldsymbol{B}_{\boldsymbol{P}}^{(1)} = \begin{bmatrix} 0.0153 & 0.0140 & 0.0142 & 0.0142 \\ 0.0140 & 0.0166 & 0.0150 & 0.0156 \\ 0.0142 & 0.0150 & 0.0180 & 0.0169 \\ 0.0141 & 0.0156 & 0.0169 & 0.0181 \end{bmatrix}.$$
(4.65)

Then the deviations of DC powers (ΔP_{dc}) and DC voltages (ΔV_{dc}) are [-68.0812, -37.5125, -15.1836, 120.0000]^T and [1.7020, 1.8756, 2.0245, 2.1725]^T, respectively. However, the results show MMC2 is overloaded (4.55), hence the droop constant of MMC2 should be set as zero ($K_{droop,MMC2} = 0$) and the maximum power deviation of MMC2 is -20 MW (4.56). Hence, the updated matrix of droop constants (K_{droop_P}) and power reference deviations (ΔP_{dc}^*) are $[40, 0, 7.5, 0]^{\mathrm{T}}$ and $[0, -20, 0, 120]^{\mathrm{T}}$, respectively.

The next step is to update (4.51) obtaining a new set of coefficient matrices $(A_P^{(2)}, B_P^{(2)})$:

$$\boldsymbol{A_{P}^{(2)}} = \begin{bmatrix} 0.1545 & -0.8410 & -0.8221 & -0.8302\\ 0.0000 & 1.0000 & 0.0000 & 0.0000\\ -0.1535 & -0.1685 & 0.8144 & -0.1792\\ 0.0000 & 0.0000 & 0.0000 & 1.0000 \end{bmatrix}, \text{ and } (4.66)$$
$$\boldsymbol{B_{P}^{(2)}} = \begin{bmatrix} 0.0211 & 0.0210 & 0.0206 & 0.0208\\ 0.0209 & 0.0249 & 0.0225 & 0.0234\\ 0.0205 & 0.0225 & 0.0247 & 0.0239\\ 0.0206 & 0.0234 & 0.0239 & 0.0254 \end{bmatrix}.$$
(4.67)

Therefore, the current ΔP_{dc} and ΔV_{dc} are $[-82.8086, -20.0000, -18.1348, 120.0000]^{T}$ and $[2.0702, 2.3110, 2.4180, 2.5823]^{T}$, respectively. Table 4.6 lists the calculation results of DC power flow in the remaining three MMCs, and Fig. 4.14(b) shows the new topology after MMC4 outage with detailed current directions.

Parameter	Parameter MMC1		MMC3
P_{dc} (MW)	520.3059	-400.0000	-118.1348
I_{dc} (kA)	1.2941	-0.9995	-0.2949
V_{dc} (kV)	402.0702	400.1955	400.6155
δ_{droop}	0.05	-	0.05
K_{droop}	40	0	7.5
Line currents	I_{l1}	I_{l2}	I_{l3}
(kA)	0.8521	0.4408	0.1469

TABLE 4.6: Calculation of DC power flow for single P/V droop control after MMC4 outage in the MMC-based DC grid.

TABLE 4.7: Simulation results of DC power flow for single P/V droop control after MMC4 outage in the MMC-based DC grid.

•	•		
Parameter	MMC1	MMC2	MMC3
P_{dc} (MW)	520.4074	-399.9999	-118.0999
I_{dc} (kA)	1.2943	-0.9995	-0.2948
V_{dc} (kV)	402.0677	400.1915	400.6109
δ_{droop}	0.05	-	0.05
K_{droop}	40	0	7.5
Line currents	I_{l1}	I_{l2}	I_{l3}
(kA)	0.8529	0.4415	0.1467

Table 4.7 presents the simulation results of DC power flow after MMC4 outage. Because jacobian matrix refers to linear mapping, the obtained calculation results inevitably have errors compared with the results in simulation, while the errors are in acceptable range.

4.4.3.2 Verification in MMC-based DC Grid under Mixed Droop Control

The second case is that MMC2 adopts I/V droop control and the remaining MMCs still use P/V droop control constituting mixed P/V and I/V droop control (Table 4.8). Based on the description in Chapter 4.4.1.2, the line current directions have to be determined first for building the relationship amongst DC powers, currents and voltages of four MMCs. Supposing the initial reference of MMC2 is still DC power (-380 MW), the branch current directions can be obtained as Fig. 4.15(a). Then the coefficient Λ in (4.42) can be further acquired as:

$$\boldsymbol{\Lambda} = \begin{vmatrix} 1 & 1 & 0 & 0 \\ -1 & 0 & 1 & 0 \\ 0 & -1 & 0 & 1 \\ 0 & 0 & -1 & -1 \end{vmatrix},$$
(4.68)

where 1 refers to current flows out from a converter while -1 indicates current flows into a converter. The line conductance matrix (G_l) of the DC grid is $[0.4545, 0.3030, 0.5682, 0.9091]^{T}$. Substituting the known values (2 DC powers, 1 DC current and 1 DC voltage) into (4.42), the initial DC power flow can be derived shown in Table 4.9. Similar as single droop control, the references of outer loop control for four MMCs with mixed droop control in simulation use the derived parameters. Also, the simulation results in Table 4.10 evidences the precise initial DC power flow calculation.



FIGURE 4.15: Detailed current directions of MMC-based DC grid under mixed droop control: (a) initial power flow, and (b) power flow after MMC4 outage.

Parameter	MMC1	MMC2	MMC3	MMC4
P_{dc}^{*} (MW)	-	-	-100	-120
I_{dc}^{*} (kA)	-	-0.95	-	-
V_{dc}^{*} (kV)	400	-	-	-
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40 MW/kV	50 A/kV	7.5 MW/kV	12.5 MW/kV

TABLE 4.8: References of outer control loop for mixed droop control in the MMC-based DC grid.

TABLE 4.9: Calculation of initial DC power flow of mixed droop control in the MMC-based DC grid.

Parameter	MMC1	MMC2	MMC3	MMC4
Turumeter	minor		minco	miner
P_{dc} (MW)	601.0913	-377.9981	-100.0000	-120.0000
I_{dc} (kA)	1.5027	-0.9500	-0.2511	-0.3016
V_{dc} (kV)	400.0000	397.8927	398.2020	397.8788
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40	50	7.5	12.5
Line currents	I_{l1}	I_{l2}	I_{l3}	I_{l4}
(kA)	0.9579	0.5449	0.0079	0.2937

TABLE 4.10: Simulation results of initial DC power flow of mixed droop control in the MMC-based DC grid.

Parameter	MMC1	MMC2	MMC3	MMC4
P_{dc} (MW)	601.0911	-377.9981	-99.9999	-120.0000
I_{dc} (kA)	1.5027	-0.9500	-0.2511	-0.3016
V_{dc} (kV)	400.0002	397.8927	398.2020	397.8788
δ_{droop}	0.05	0.05	0.05	0.05
K_{droop}	40	50	7.5	12.5
Line currents	I_{l1}	I_{l2}	I_{l3}	I_{l4}
(kA)	0.9578	0.5450	0.0079	0.2937

For obtaining the final DC power flow under mixed droop control after an outage of MMC4, the line current directions in current topology should also be determined first by assuming MMC2 operates with P/V droop control, which indicates the current directions in single P/V droop control (Chapter 4.4.3.1) can be used for such mixed droop control (Fig. 4.15 (b)). The next step is the establishment of power flow calculation equation (4.60) for the current topology after MMC4 isolation. The current coefficient Λ' is:

$$\boldsymbol{\Lambda}' = \begin{bmatrix} 1 & 1 & 0 \\ -1 & 0 & -1 \\ 0 & -1 & 1 \end{bmatrix},$$
(4.69)

and the current line conductance matrix G_l is $[0.4545, 0.3030, 0.3497]^T$. In addition, the references of remaining three MMCs are not changed, thus substituting known values into (4.60), (4.61) and (4.62), the DC power flow after MMC4 outage can be yielded.

Nevertheless, the power of MMC2 exceeds the maximum power limitation after the preliminary calculation (4.58), and the actual remaining power capacity of MMC2 is 22.0019 MW (4.58). Hence, the control mode of MMC2 switches into constant DC power control for protecting converter, and the updated K_{droopP} and ΔP_{dc}^* are $[40, 0, 7.5]^{T}$ and $[0, -22.0019, 0]^{T}$, respectively. Then the final DC power flow can be accurately derived by re-constructing (4.60), (4.61) and (4.62) considering the updated vectors of droop constants and actual DC power deviations (Table 4.11). The simulation results are shown in Table 4.12, which demonstrates the accuracy of proposed DC power flow calculation method under mixed droop control.

Parameter	Parameter MMC1		MMC3
P_{dc} (MW)	520.0655	-400.0000	-117.7626
I_{dc} (kA)	1.2936	-0.9996	-0.2940
V_{dc} (kV)	402.0256	400.1499	400.5703
δ_{droop}	0.05	-	0.05
K_{droop}	40	0	7.5
Line currents	I_{l1}	I_{l2}	I_{l3}
(kA)	0.8526	0.4410	0.1470

TABLE 4.11: Calculation of DC power flow for mixed droop control after MMC4 outage in the MMC-based DC grid.

TABLE 4.12: Simulation results of DC power flow for mixed droop control after MMC4 outage in the MMC-based DC grid.

Parameter	MMC1	MMC2	MMC3
P_{dc} (MW)	520.0654	-400.0000	-117.7627
I_{dc} (kA)	1.2935	-0.9996	-0.2939
V_{dc} (kV)	402.0256	400.1499	400.5704
δ_{droop}	0.05	-	0.05
K _{droop}	40	0	7.5
Line currents	I_{l1}	I_{l2}	I_{l3}
(kA)	0.8525	0.4410	0.1471

4.5 Conclusion

Various modeling approaches are described in the simulation of HVDC systems based on actual study requirements. Also, numerical integration plays significant role in solving network differential equations. EMT modeling captures the detailed switching behaviours and parallel computation avails the increase of EMT simulation speed. Electromechanical transient modeling approach is suitable for studying large-scale HVDC systems by simplifying controller design and ignoring high frequency dynamic behaviours. DP and SFP modeling methods increases simulation speed while retains accuracy by capturing dynamic frequency of HVDC systems. Co-simulation study combines different modeling approaches that serves specific requirements of accuracy in subsystems.

Droop control is a conventional control strategy for MTDC systems and DC grids. Detailed P/V, I/V and mixed droop control schemes are described considering maximum power limitation. The proposed generalized DC power flow expression studies the initial DC power flow for normal operation and the DC power flow after converter outage. A four-terminal MMC-based DC grid simulation model is built to obtain the initial power flow and the power flow after MMC outage. The maximum power limitation of MMC is also considered. The simulation results are consistent with the obtained results in simulation verifying the accuracy of proposed generalized expression of DC power flow, which can also be used to assess the static security of MTDC systems and DC grids.
Chapter 5

LCC and AAC-Based Hybrid PTP HVDC Transmission System

5.1 Introduction

This chapter introduces the development of a hybrid PTP HVDC transmission system based on an LCC and an AAC. The basic operation principles and control methods of the developed LCC-AAC hybrid PTP HVDC system are demonstrated following with appropriate system design and parameter selection. In addition, a set of control schemes is proposed for riding through AC faults and handling DC faults. The operation and performance of both converters in such a hybrid HVDC configuration are verified through a detailed set of results based on an LCC-AAC hybrid PTP HVDC system. These include steady state operation, AC-grid harmonics & DC-side ripple analysis, reference tracking (active power change at LCC station, reactive power change at AAC station) and system performance under AC-grid faults without blocking converter at both terminals of the HVDC system. Also, the response of the hybrid system under DC fault presents potential DC fault handling capability compared with the hybrid system based on LCC and MMC with HBSMs. The hybrid LCC-AAC PTP HVDC transmission system lays preliminary steps for research on complex mixed converter multiterminal DC configurations in future DC super grids.

5.2 Converter Description and Control Hierarchy

5.2.1 Converter Description

The topologies and operations of the LCC and AAC were extensively discussed in Chapter 3 and only the necessary information relevant to this chapter are described in this chapter. The operation of the LCC relies on firing angle enabling the LCC to work in rectifier or inverter mode. The typical firing angle for the LCC in rectifier mode is around 5° or 15° [138]. A smaller firing angle is more attractive due to the reduced reactive power consumption and lower overall losses, at the cost of limited system adjusting capability [138]. The normal operation of the AAC is different from the MMC due to the alternate operation of the two arms in a phase during each half-cycle of the output voltage, which also means the arms conduct the total output current (i_a). The definition of output voltage in the AAC, $v_{ac} = m_a \frac{V_{dc}}{2} \cos(\omega t + \phi)$ is similar to the MMC. The distinct operating point that has the inherent energy balancing of the AAC is called the "sweet-spot", which was explained in Chapter 3.4.2.2. The sweet-spot operation ($m_a = M_a = 4/\pi$) can be determined by the zero net energy from the energy exchange between AC side and DC side [118].

5.2.2 Control Hierarchy

The control hierarchy of the hybrid PTP HVDC systems is shown in Fig. 5.1. Following with the control hierarchy description of the MMC and AAC in Chapter 3.3.2.1 and 3.4.2.1, respectively. The first two layers (AC/DC grid control & coordinated system control) apply to both the LCC and AAC. The third layer (converter station control) performs the high level control of the LCC and AAC stations. The active/reactive power, node voltages/currents, frequency and firing/extinction angle in the third layer are regulated based on the references. The bottom layer is the internal converter control, which is also the low-level control of the LCC and AAC that regulates the i) pulse signal triggering, ii) SM sorting, iii) SM energy, iv) zero-current switching, v) circulating current, vi) overlap period and vii) OLTC coordination for extending ZCS region [118].

The LCC can adopt either constant DC current or voltage control generating proper firing angle to pulse controller, while constant firing angle/extinction angle control, if necessary, can be employed as well [128, 137]. In addition, LCC at the inverter side can equip with VDCOL avoiding coummutation failures, while the VDCOL at the rectifier side can ensure the fast and stable recovery of the LCC from faults to some extent [138]. The high and low level control for the AAC were introduced in Chapter 3.4.2.1.



FIGURE 5.1: Control Hierarchy of the LCC and AAC-based hybrid PTP HVDC transmission system.

5.3 System Topology and Control Scheme

5.3.1 System Topology

The topology of the hybrid HVDC system, based on a \pm 500 kV, 3000 MVA bipole system with two 12-pulse LCCs in rectifier side and an AAC in the inverter side is shown in Fig. 5.2. Similarly to other LCC-VSC hybrid HVDC systems (Chapter 1.2), the LCC and AAC can be either in the rectifier or in the inverter terminal. Accordingly, the LCC side can adopt constant DC current or voltage control, while the AAC can adopt constant DC voltage control or power control, respectively. In this chapter, the hybrid LCC-AAC HVDC system embraces the natural immunity to commutation failures as the AAC is in the inverter side. Thus, it is not necessary for such configuration to set additional control units to handle commutation failures.

The system circuit under DC fault is shown in Fig. 5.3, which depicts the potential DC fault current path of pole-to-pole DC fault occuring at the midpoint of DC transmission line.



FIGURE 5.2: Topology of the LCC and AAC-based hybrid PTP HVDC transmission system.



FIGURE 5.3: System circuit of the LCC and AAC-based hybrid PTP HVDC transmission system under DC fault.

5.3.2 Control Scheme

5.3.2.1 Control Scheme under Normal Operation

Fig. 5.4 provides a complete description of all levels of control considered in the hybrid PTP LCC-AAC HVDC system. As mentioned earlier, the LCC side controller does not consider additional functionalities for handling commutation failures and the AAC side control has been implemented without energy regulation of the AAC during direct or indirect control [30]. At the LCC station, the DC current is controlled by generating firing angles from 5° to 150° . As the system is based on a 12-pulse rectifier, the controller generates two sets of pulses with 30 degree shift. In order to coordinate with the controller of the LCC station, the DC voltage to maintain constant power with additional

reactive power control as the outer control loops. In addition, the AC output voltage reference generated from the inner current controller is used the modulation stage, overlap period and circulating current.



FIGURE 5.4: Control Structure of the LCC and AAC-based hybrid PTP HVDC transmission system.

The main purpose of the modulation stage is to determine the number of SMs utilized in each arm, and the nearest level modulation (NLM) is adopted for the AAC, which is also the most advantageous method for the large number of SMs [197]. The overlap period controller determines the alternate operation of director switches (DSs) and the energy balancing when the AAC operates with a short overlap period, while the circulating current controller decides the number of additional SMs for over-modulation operation [118].

5.3.2.2 Control Scheme under AC and DC Faults

Based on the aforementioned control structure of the LCC and AAC-based hybrid HVDC system, the LCC cannot maintain DC current constant when AC fault occuring at the LCC side, which means the DC current will decrease with the fault level. When the AC fault is cleared, the DC voltage of the LCC will experience instantaneous increase which would lead to a transient overcurrent in the DC-link of the LCC side. Fig. 5.5 (a) shows the recovery control strategy for riding through AC fault of the LCC side and avoiding overcurrent. First, the VDCOL can be employed for stable recovery of DC current [138]. In addition, the integrator of DC current controller will be reset to 0 for increasing the firing angle, then reducing the DC voltage of LCC side to a maximal degree.

Similarly, in the case of an AC fault on the AAC side, the outer control loop for the AAC is temporarily disabled in order to assist the response of the converter due to the fast response of the inner current controller compared to the outer loop. In addition, the controller of the AAC will be re-enabled when receiving the clearing signal of the AC fault, as shown in Fig. 5.5 (b).



FIGURE 5.5: Control scheme for riding through AC fault in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) AC fault occurring at LCC side and (b) AC fault occurring at AAC side.

Fig. 5.6 (a) and (b) show the control schemes for a handling permanent DC fault in the hybrid HVDC system based on the LCC with AAC and HB-MMC, respectively. First, the LCC has to force the firing angle into the maximum value, then the LCC will operate as inverter [88]. Moreover, the AAC and HB-MMC are blocked when detecting the DC fault for protecting IGBTs of SMs, and the DC voltage and reactive power references of the AAC are set to 0 at the same time. For the combination of LCC and HB-MMC, the AC breaker for the HB-MMC has to open for utterly cutting off the fault current while the DC breaker is



FIGURE 5.6: Control scheme for handling permanent DC fault in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) LCC-AAC hybrid HVDC system and (b) LCC-HBMMC hybrid HVDC system.

more efficient at the cost of high investment. It is worth mentioning that it is necessary to switch the DSs off during the DC fault for thoroughly cutting off the fault current, since the antiparallel diodes in DSs further ensures the fault current can only flow from fault point to upper arms via lower arms. The comparison of DC fault handling capability between the LCC with the HB-MMC and AAC will be shown in Chapter 5.5.4.

5.4 System Parameter Selection

5.4.1 Main Parameters of the LCC

The detailed model parameters of the LCC are given in Table 5.1, with the rated AC voltage level for the LCC side selected as 525 kV. The DC output voltage (V_{dcl}) of LCC terminal is:

$$V_{dcl} = R_{dc}I_{dcl} + V_{dca},\tag{5.1}$$

where I_{dcl} and R_{dc} is the DC current of the LCC side and line resistance respectively, V_{dca} is the DC output voltage at the AAC DC terminals (±500 kV). The active power and reactive power can be calculated as (5.2) and (5.3), where ϕ is power factor angle.

$$P_{dcl} = V_{dcl} I_{dcl}, \tag{5.2}$$

$$Q_{dcl} = P_{dcl} \tan \phi. \tag{5.3}$$

Parameters	LCC Rectifier Side	
Rated Power	3000 MVA	
Nominal Frequency	50 Hz	
AC Voltage (L-L,rms)	525 kV	
Transformer Ratio	525/218	
Transformer Leakage Inductance (p.u.)	0.18	
AC System Impedance	23.5∠84.3°	
Reactive Compensation Capacity	1400 MVar (140 MVar/group)	

TABLE 5.1: Rectifier parameters (LCC) in the LCC and AAC-based hybrid PTP HVDC transmission system.

The secondary side voltage of transformer follows (5.4), where N is the number of 6pulse converters (N = 2 in 12-pulse converter), k_s is $3\sqrt{2}/\pi$ for a 6-pulse thyristor bridge, α is the firing angle and X_L is the leakage reactance of transformer.

$$V_{2l} = \frac{V_{dcl}}{k_s N \cos \alpha} + \frac{X_L I_{dcl}}{\sqrt{2} \cos \alpha} = \frac{V_{dcl}}{k_s N (\cos \alpha - \frac{X_{p.u.}}{2})}.$$
(5.4)

5.4.2 Main Parameters of the AAC

Each arm in the AAC consists of H-bridge SMs and an array of DSs. Adopting H-bridge SMs enables the AAC to operate in the over-modulation range and embrace DC line fault ride-through capabilities. In addition, the arm voltage in the AAC is only half of the DC voltage, since two DSs in each phase alternately conduct the AC current. The parameters of the AAC station are provided in Table 5.2, which can be divided into four main aspects: i) system standard operating point; ii) the number of SMs and IGBTs in each DS; iii) overlap period and iv) SM capacitance and arm inductance [118].

The energy exchange between the arms in each phase is limited by the alternate operation of DSs, and the final AC and DC energy balancing point is defined as the system operating point, which is also called sweet spot ($M_a = 4/\pi = 1.27$), but considering $\pm 10\%$ voltage variation under normal operation, the standard operating point (m_a) adopted in the hybrid system is expressed as (5.5).

$$m_a = \frac{M_a}{k_a},\tag{5.5}$$

Parameters	AAC Inverter Side
Rated Power	3000 MVA
Nominal Frequency	50 Hz
AC Voltage (L-L,rms)	525 kV
Transformer Ratio	525/702
Transformer Leakage Inductance (p.u.)	0.18
Transformer Resistance (p.u.)	0.006
Number of SMs per arm	640
Nominal SM Voltage	1 kV
Stored Energy	11 kJ/MVA
SM Capacitance	17.2 mF
Arm Inductance (p.u.)	0.02
Nominal Operating Point (m_a)	1.15
AC System Impedance	7.9 ∠ 84.3°

TABLE 5.2: Inverter parameters (AAC) in the LCC and AAC-based hybrid PTP HVDC transmission system.

where $k_a \in (1, 1.1)$. Accordingly, the peak voltage stress of DSs (\hat{V}_{DS}) and peak output voltage of the AAC (\hat{V}_a) at such standard operating point are:

$$\hat{V}_{DS} = \frac{V_{dc}}{2} \tag{5.6}$$

$$\hat{V}_a = m_a \hat{V}_{DS} = \frac{M_a V_{dc}}{2k_a} \tag{5.7}$$

FBSMs adopted in the AAC can block the AC voltage at the standard operating point and enable the AAC to operate at over-modulation range. In addition, the maximum withstand voltage for nonconducting DSs per arm is half of the DC voltage. Hence, the number of SMs (N_{SM}) and IGBTs in each DS (N_{DS}) per arm at standard operating point is determined as:

$$N_{SM} = \frac{\hat{V}_a}{V_c} = \frac{M_a V_{dc}}{2k_a V_c} \tag{5.8}$$

$$N_{DS} = \frac{\hat{V}_{DS}}{V_c} = \frac{k_a N_{SM}}{M_a},$$
(5.9)

where V_c is the nominal capacitor voltage of each SM in the AAC and is set as 1 kV in this chapter.

The overlap period in the AAC allows exchange of energy between upper arm and lower arm, and the output current flowing two arms is equally distributed. The period of overlap should be within a reasonable range avoiding output voltage distortions, and the maximum value of it is limited by the redundant voltage (V_r). Additionally, the redundant voltage varies within a $\pm 10\%$ range considering the SM capacitor voltage ripple. Thus, assuming the output reference voltage $v_o = m_a \cos(\omega t)$, the range of redundant voltage and overlap period (\hat{t}_{ov}) in the AAC can be respectively expressed as:

$$\frac{V_{dc}(0.9M_a - k_a)}{2k_a} \le V_r \le \frac{V_{dc}(1.1M_a - k_a)}{2k_a}, \quad \text{and} \quad (5.10)$$

$$\hat{t}_{ov} = \frac{\pi - 2\cos^{-1}(1 - \frac{k_a}{M_a})}{\omega}$$
(5.11)

The energy stored in AAC is set to 11 kJ/MVA. The value of capacitance in each SM of the converter is calculated based on (5.12). The value of the arm inductance follows with the calculation in [118] and can be determined as (5.13), which considers the redundant voltage utilization efficiency to achieve circulating current control, and a minimum circulating current gradient.

$$C_{SM} = \frac{2SE_{AAC}}{6N_{SM}V_c^2} \tag{5.12}$$

$$L \le \frac{3V_{dc}V_r(\pi - 2\cos^{-1}(1 - \frac{k_a}{M_a}))}{8\omega\pi S}$$
(5.13)

where, S is the converter rating and E_{AAC} is the energy requirement in each SM of the AAC.

5.4.3 DC Transmission Line and Control Parameters

The system includes a 500 km DC transmission line (cable) at voltage level of \pm 500 kV. The parameters of the transmission line (lumped parameter pi-section) are listed in Table 5.3. In addition, Table 5.3 also provides the control parameters used in the low-level and high-level controllers for both stations.

	Transmission Line	Parameters
	DC Line Voltage (kV)	± 500
	Length (km)	500
	Resistance (Ω/km)	0.011
	Inductance (mH/km)	0.2615
	Capacitance (μ F/km)	0.2185
	Conductance (μ S/km)	0.055
Control Para		meters
	DC Current Control - LCC	$K_P = 0.25, K_I = 92.85$
	DC Voltage Control - AAC	$K_P = 8, K_I = 272.5$
	Reactive Power Control - AAC	$K_P = 1.3, K_I = 30$
	Energy Balancing - AAC [30]	$K_P = 2.9, K_I = 75$
	PLL - LCC	$K_P = 5, K_I = 10$
	PLL - AAC	$K_P = 0.084, K_I = 4.69$

TABLE 5.3: DC transmission line and control parameters in the LCC and AAC-based hybrid PTP HVDC transmission system.

5.4.4 AC and DC Link Filters

Harmonics in AC and DC sides need to be considered in HVDC transmission systems, as the harmonic currents lead to voltage and current distortion at the point of common coupling (PCC) and DC-link. Therefore, AC and DC filtering requirements are major design considerations for LCC-VSC hybrid HVDC systems.

The LCC-based HVDC station has both characteristic and uncharacteristic harmonics in the AC and DC sides [198]. Thus, AC and DC filters are required to reduce harmonic levels. The tuning points for AC filters of LCC are set at 3rd, 11st, 13rd, 24th and 36th,

respectively. There are in total 10 groups of AC filters (140 MVar/group) installed for flexible switching, which are further classified into i) 3 groups of shunt capacitor (SC), ii) 4 groups of double-tuning (DT) filters tuned at $11^{st}/13^{rd}$ and iii) 4 groups of triple-tuning (TT) filters tuned at $3^{rd}/24^{th}/36^{th}$. The configurations and detailed parameters are shown in Fig. 5.7 and Table. 5.4. In addition, the $12^{nd} \& 24^{th}$ harmonics are the main characteristic harmonics in the DC side of LCC, hence a DT DC filter is designed as Fig. 5.8 with detailed parameters in Table 5.6. Moreover, the smoothing reactor in DC-link of LCC station is necessary to be installed for decreasing DC current ripple and avoiding current discontinuity in minimum power [128, 199].



FIGURE 5.7: AC filters in the LCC station.



FIGURE 5.8: DC filters in the LCC station.

The DC-link current at the AAC side contains a six-pulse ripple, hence the AAC also requires the use of a DC filter [22, 118]. However, AC filters are not necessary for

Parameters	SC	DT (11/13)	TT (3/24/36)
Number of Groups	3	4	3
Capacity (MVar)	140	140	140
C _{a11} (µF)	1.616	-	-
C_{a21} ($\mu { m F}$)	-	1.605	-
C_{a22} ($\mu { m F}$)	-	56.824	-
C_{a31} (μ F)	-	-	1.578
C_{a32} (μ F)	-	-	7.218
C_{a33} ($\mu { m F}$)	-	-	7.704
L _{a11} (mH)	2.721	-	-
L _{a21} (mH)	-	44.731	-
L_{a22} (mH)	-	1.239	-
L _{a31} (mH)	-	-	8.116
L _{a32} (mH)	-	-	129.39
L _{a33} (mH)	-	-	1.634
R_{a21} (Ω)	-	2500	-
R_{a31} (Ω)	-	-	400
R_{a32} (Ω)	-	-	1500

 TABLE 5.4: Detailed parameters of AC filters in the LCC station.

the AAC, owing to the near sinusoidal multilevel output voltages. The AAC with short overlap period requires a DC capacitor to realise sufficient energy exchange between DC side and SMs, and the energy stored in DC-link capacitor for the AAC is equal to about one third of the total capacitive energy in SMs. Considering the energy deviation $(\Delta E_{AAC} \in (0.9, 1.1))$ of SMs, the energy storage of DC capacitor can be calculated as $C_{dc}V_{dc}^2/2 = S\Delta E_{AAC}E_{AAC}/3$ [200], then the required capacitance value at DC-link can follow with (5.14).

$$C_{dc} > \frac{2S\Delta E_{AAC} E_{AAC}}{3V_{dc}^2}$$
(5.14)

The DC-link capacitor and smoothing reactor are essential in AAC terminal due to the six-pulse current ripple caused by the alternate operation of the AAC arms [118]. For filtering the 6^{th} harmonic and other higher harmonics, the reactance value is calculated as (5.15), following the characteristics of an LC filter with a cut-off frequency below 6f.

$$L_{dc} > \frac{1}{144\pi^2 f^2 C_{dc}} \tag{5.15}$$

The smoothing reactance value in DC-link of the LCC is far larger than the AAC [118, 128] which is usually from 200 mH to 700 mH [201]. Additionally, the DC voltage in developed system is controlled by the AAC side, thus the DC voltage deviation at the AAC station will make an impact on DC voltage of the LCC, although the main characteristic

harmonics are filterd by two sets of double-tuned damping DC filters. In general, the DC filter and smoothing reactor design of the LCC can follow with conventional scheme [128], and the values of DC capacitance and smoothing reactance of the AAC can be decided by (5.14) and (5.15) (Table 5.5).

Parameters LCC (rectifier)		AAC (inverter)
L_{dc1}/L_{dc2}	200 mH~700 mH	>11.6 mH
DC filter/ C_{dc}	double (triple)-tuned damping filter	$>$ 24.2 μ F

TABLE 5.5: The limitation of DC-link filters in the LCC and AAC stations.

For determining the appropriate DC-link filters in the LCC-AAC hybrid system, it is necessary to analyze DC-side ripples for both the LCC and AAC stations, since the low ripples mean high quality of output waveforms. Moreover, the low ripples also reflect less AC components (harmonics) in DC voltage and current.

Based on the above analysis, the related values of DC-link filters for developed system are shown in Table 5.6 restricting DC current ripples below 3% and voltage ripples below 2% of both stations. The values of smoothing reactance of the LCC and AAC should be in the range of limitation in Table 5.5 and can be changed by the expected DC-link ripples. In addition, the DC capacitor of AAC with proper value can not only coordinate with smoothing reactor to obtain smoother DC current and decrease voltage ripples, but also ensure system stable operation by balancing power between DC and AC. The DC-link ripples can be further reduced, if necessary, with larger smoothing reactor, capacitor as well as multi-tuned DC filter at the cost of higher investment.

DC-link filters at LCC and AAC stations		
	$C_{d1} = 0.35 \ \mu\text{F}, L_{d1} = 89.35 \ \text{mH}$	
DC Filter of LCC	$C_{d2} = 0.81 \ \mu$ F, $L_{d2} = 48.86 \ m$ H	
	$R_{d1} = 10000 \ \Omega$	
Smoothing Reactance of LCC	250 mH	
DC Capacitance of AAC	65 <i>µ</i> F	
Smoothing Reactance of AAC	25 mH	

TABLE 5.6: Detailed parametters of DC-link filters in the LCC and AAC stations.

5.4.5 Comparison between the MMC and the AAC

In order to demonstrate benefits of the AAC in hybrid HVDC systems, the main differences between the MMC and the AAC are described with the same power rating and voltage level. The related parameters of the MMC are in Table 5.7. Compared with HB-MMC,

AAC has over-modulation capability as the employment of bipolar SMs, hence potential DC fault handling capability. In addition, the AAC requires a lower number of SMs per arm compared with FB-MMC due to the alternate operation of two arms in each phase. Although the introduction of DSs for the AAC lead to additional IGBTs per arm, the total number of semiconductor devices for the AAC are still less than the FB-MMC. Considering the $\pm 10\%$ capacitor voltage deviation, the energy requirement of SMs in the AAC is only 11 kJ/MVA, which is about one third of such in the MMC. The capacitors of SMs occupy the main volume in modular VSCs, thus the total converter size of the AAC is less than the MMC while the AAC has to install additional DC-link filter [118].

Based on the above discussion, the AAC is suitable for future HVDC market. Hence, the AAC can replace the MMC combining with the mature LCC to constitute hybrid LCC and AAC-based HVDC system by employing individual benefits.

Parameters	MMC Inverter Side
Rated Power	3000 MVA
Nominal Frequency	50 Hz
AC Voltage (L-L,rms)	525 kV
Transformer Ratio	525/551
Transformer Leakage Inductance (p.u.)	0.18
Transformer Resistance (p.u.)	0.006
Number of SMs per Arm	1000
Nominal SM Voltage	1 kV
Stored Energy	30 kJ/MVA
SM Capacitance	30 mF
Arm Inductance (p.u.)	0.15
Nominal Operating Point (m_a)	0.9
R/X Ratio of AC System	0.1

TABLE 5.7: Inverter Parameters (MMC) in the LCC and MMC-based hybrid PTP HVDC transmission system.

5.5 Simulation Results

In order to verify the performance of the LCC-AAC hybrid HVDC system, a detailed simulation model (Fig. 5.2) is implemented using PLECS-Blockset and Simulink. The results based on the implemented 3000 MVA hybrid HVDC system demonstrate four different operating scenarios: i) steady state operation with harmonic and ripple analysis for AC side and DC link, respectively, ii) reference tracking including the active power change of LCC side and reactive power change of AAC side, iii) AC faults at both stations, and iv) DC fault at the midpoint of the DC tranmission line, validating the operation of hybrid HVDC system within widely accepted HVDC operating standards [101, 202].

5.5.1 Steady-State Operation

In steady-state operation, the LCC station (rectifier) operates in DC current $(I_{dc1}^* = 3 \text{ kA})$ control mode obtaining firing angle about 15°. The AAC station (inverter) operates in the DC voltage ($V_{dc}^* = 1000 \text{ kV}$) and reactive power ($Q^* = 0 \text{ MVAr}$). Fig. 5.9 and Fig. 5.10 show the AC and DC side voltages and currents of the hybrid HVDC system, respectively.



FIGURE 5.9: Steady state waveforms of the LCC side in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) DC voltage, (b) AC grid voltage, (c) DC current and (d) AC grid current.

The DC current at the rectifier side is regulated to the reference, I_{dc}^* with a ripple of 2.5%. The rectifier DC voltages is at $1.033V_{dc}^*$ and the direct converter output voltage before DC filters consists of the twelve-pulse ripple of 18.3%, while such ripple is reduced to 1.2% after DC filters. The inverter side DC voltage is regulated to the reference, V_{dc}^* with ripple of 0.8% and the DC current comprises of a sixth harmonic ripple of 2% which is imposed by the alternate operation of the AAC arm [22]. The DC voltages and currents demonstrate that all ripples are contained within the network standard limits at full rated power and also the appropriate design and selection of the DC filter parameters from Chapter 5.4.4.

Fig. 5.11(a) and Fig. 5.12(a) respectively show the harmonics of DC side after filters in the LCC and AAC station. The DC current in the LCC station mainly contains 6k order harmonics (Fig. 5.11(a)), and the low order harmonics of DC current in the AAC station



FIGURE 5.10: Steady state waveforms of the AAC side in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) DC voltage, (b) AC grid voltage, (c) DC current and (d) AC grid current.

(Fig. 5.12(a)) is larger than the LCC station, with each individual harmonic for the DC side of the hybrid HVDC system below 1%.

The AC voltages and currents are also maintained to comply with network operating standards, limiting the %THDs below 2% as shown in Fig. 5.11(b) and Fig. 5.12(b),



FIGURE 5.11: Harmonics of the LCC station in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) harmonics of DC voltage and DC current and (b) harmonics of AC voltage and AC current.

which also satisfy the IEEE standard for THD at PCC (2% for AC voltage, 1.5% for AC current) [202]. In addition, the power from both AC systems agree with the desired operation of the HVDC system according to the references.

The AAC requires active energy balancing for the normal operation which is performed using the low-level control functions [22, 30, 118]. Fig. 5.13(a) and (b) show the SM capacitor voltages and arm currents of the AAC HVDC station, and the arm current direction follows with the Fig. 5.2. The active energy balancing and overlap circulating current control method of [30] is employed in the AAC HVDC station. The SM capacitor voltages are regulated to their nominal references and the arm currents remain balanced demonstrating the performance of low-level energy balancing and circulating current control functions. Hence, these results demonstrate the performance of high-level control functions of the LCC and AAC HVDC stations and the feasible operation of the LCC-AAC hybrid system.

5.5.2 Reference Tracking / Steady-State

5.5.2.1 DC Current Reference Change

The LCC side adopts a constant DC current control mode to control active power, and the initial active power is set at 3000 MW (1 p.u.). The system performance when the active



FIGURE 5.12: Harmonics of the AAC station in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) harmonics of DC voltage and DC current and (b) harmonics of AC voltage and AC current.



FIGURE 5.13: Capacitor voltages and arm currents of the AAC in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) SM capacitor voltages and (b) arm currents.

power reference decreases to 0.6 p.u at 0.2 s. is shown in Fig. 5.14(1). Fig. 5.14(1) (a)-(b) show the power variation of both terminals, where the decrease of active power at the LCC side coincides with the increase of reactive power due to the decreased reactive power consumption of LCC, while the reactive power of the AAC side is maintained constant as the AAC can independently control the reactive power at its AC terminal. Moreover, the AC voltage and current change of the LCC side is shown in Fig. 5.14(1) (e)-(f). However, the AC voltage of the AAC keeps constant and the AC current drops with change of power (Fig. 5.14(1) (g)). In addition, the ripple of capacitor voltage in Fig. 5.14(1) (h) reduces significantly during the process.

5.5.2.2 Reactive Power Reference Change

Fig 5.14(2) shows the hybrid system response when reactive power of the AAC terminal increases to 0.5 p.u. from 0 at 0.2 s. Fig 5.14(2) (a) and (b) show the power regulation of both stations. Fig 5.14(2) (c) and (d) demonstrate the DC voltage and current can maintain constant during reactive power change. Moreover, there are only small deviations of both AC voltage and current (Fig 5.14(2) (e) and (g)) in the AAC station. The result demonstrates that the developed hybrid system can work under reactive power reference change while with larger ripple of capacitor voltage and arm current. In addition, the change of reactive power reference has no impact to the LCC side (Fig 5.14(2) (f)), which guarantees the system reliability when providing reactive power support at the AAC side.



(1) DC current reference change.

(2) Reactive power reference change.

FIGURE 5.14: Performance under DC current and reactive power reference changes in the LCC and AAC-based hybrid PTP HVDC transmission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current, (e) AC grid voltage of LCC, (f) AC grid current of LCC, (g) AC grid current of AAC and (h) SM capacitor voltage of AAC.

5.5.3 Performance under AC Faults

The AC FRT capability of hybrid HVDC system ensures robustness of the hybrid HVDC system. Thus, this part will discuss both the single-line to ground fault (SLG) and three-phase short circuit fault at both the LCC and AAC sides. Both AC-grid faults are applied at the terminals of each station and with AC voltage drop about 60%, defining the envelope of performance for the hybrid system.

5.5.3.1 AC Faults at the AC-Grid of the LCC

Fig. 5.15(1) shows the behavior of the overall system during a 200 ms SLG fault at the AC-grid of the LCC side. The power oscillation during the fault is shown in Fig. 5.15(1) (a)-(b) with a DC side voltage and current deviation in Fig. 5.15(1) (c)-(d) due to the oscillations of AC side at the LCC terminal (Fig. 5.15(1) (e)-(f)). The AC current reduction at the AAC side (Fig. 5.15(1) (g)) is due to the instantaneous power drop in the DC side. In addition, the system performance under 200 ms three-phase short circuit fault of the LCC side (Fig. 5.15(2)) is similar as SLG fault, but power (Fig. 5.15(2) (a)-(b)) and AC current (Fig. 5.15(2) (e) and (g)) of both sides temporarily reduce to zero. The capacitor voltage (Fig. 5.15(1) (h) and Fig. 5.15(2) (h)) increases approximately by 15% and 20%, respectively at the moment of fault recovery with the increase of DC voltage.

It is noteworthy that the firing angle increases, reducing the higher instantaneous DC voltage of the LCC terminal via integrator reset of DC current controller, hence there is no obvious DC current increase for both the LCC and AAC stations. In addition, VDCOL is adopted for fast recovery of DC current. In general, the developed system under AC faults at the LCC station presents the same performance of other LCC-VSC topologies demonstrating the accuracy of developed LCC-AAC hybrid system [65, 77, 80].

5.5.3.2 AC Faults at the AC-Grid of the AAC

Fig. 5.16(1) and Fig. 5.16(2) show the system performance during a SLG and three-phase short circuit fault at the AAC side, respectively. Similarly to the previous analysis, the power oscillation, DC side voltage and current are shown in Figs. 5.16(1) (a)-(d) and Figs. 5.16(2) (a)-(d). The AC voltage and current oscillations are respectively shown in Fig. 5.16(1) (e)-(h) and Fig. 5.16(2) (e)-(h). In addition, the capacitor voltage deviation is within 0.5 p.u during a SLG fault (Fig. 5.16(1) (i)) and 0.7 p.u. during a three-phase short circuit fault (Fig. 5.16(2) (i)).

The results show that the developed hybrid system can ride through AC faults of the AAC station and maintain power transmission under low voltage level without blocking converter. When the AC faults are cleared, the whole system can recover to normal operation in less than 150 ms, demonstrating the recovery capability of developed hybrid HVDC system.



(1) Single-line to ground fault at the AC-grid of the (2) Three-phase short circuit fault at the AC-grid of the LCC.

FIGURE 5.15: Single-line to ground and short-circuit faults at the AC-grid of the LCC in the LCC and AAC-based hybrid PTP HVDC transimission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current, (e) AC grid voltage of LCC, (f) AC grid current of LCC, (g) AC grid current of AAC and (h) SM capacitor voltage of AAC.

5.5.4 Permanent Pole-to-Pole DC Fault

The LCC can block DC fault current by increasing its firing angle to above 90°, switching to inverter operation mode. The AAC is capable of clearing DC faults without AC and DC-breakers via converter blocking. Thus, the developed LCC-AAC hybrid system can achieve



(1) Single-line to ground fault at the AC-grid of the (2) Three-phase short circuit fault at the AC-grid of AAC. the AAC.



DC fault self-clearance capability. For comparison, this part also shows the result of DC fault handling capability in the hybrid HVDC system based on the LCC and the HB-MMC.

A pole-to-pole permanent DC fault at the midpoint of the line occurs at 0.2s with

resistor of 10 Ω as shown in Fig. 5.17. When the DC fault is detected, the firing angle in the LCC side increases to maximum value (150°) forcing the LCC to inverter operation (Fig. 5.17(f)). The AAC converter is blocked after 4 ms. The active and reactive power changes are shown in Fig. 5.17(a)-(b). The active power of both the LCC and AAC stations drops to 0 after blocking converter. The reactive power is constant during the fault in the AAC station, while the increase of reactive power in the LCC station increases is due to the presence of AC filters. Oscillations in the DC voltages during DC fault caused by the discharge of the DC capacitances and inductances, as shown in Fig. 5.17(c). In addition, the DC current in the LCC station increases to 7 kA at the moment of DC fault, while the DC fault current at point A of Fig. 5.3 in the AAC station increases to 5 times the nominal value and presents obvious oscillation due to the discharging of DC capacitor (Fig. 5.17(d)). Although the fault current to the converter of the AAC (point B in Fig. 5.3) increases to about 8 kA when DC fault occuring, it rapidly decreases to zero after converter blocking (Fig. 5.17(e)). The capacitor voltages of the AAC first discharge to the DC-link when DC fault occuring, then increase after converter blocking to partially absorb energy from the DC-link and AC system (Fig. 5.17(h)).

Fig. 5.18 shows the results of the HB-MMC replacing the AAC in hybrid HVDC system. The MMC with HB-SMs does not posssess the DC fault self-clearing capability, hence it is necessary to open AC-grid breakers after converter blocking. In the hybrid HVDC system based on the LCC and the HB-MMC, the tripping signals of AC-breaker are triggered together with the converter blocking (after a delay of 4 ms) and within 10 ms all the AC breakers open, then the fault current of the MMC side will gradually decrease from 14 kA to 0, and the decay time is about 50 ms (Fig. 5.18(b)). The oscillation of DC voltage in the DC-link of the MMC is less than the AAC due to the absence of DC filter (Fig. 5.18(a)). The SM capacitors of the HB-MMC discharge first when DC fault occurring, but the SM capacitor voltages of the HB-MMC keep constant after converter blocking (Fig. 5.18(e)). In general, the results demonstrate the DC fault handling capability of the developed hybrid LCC-AAC HVDC transmission system, and the AAC in hybrid HVDC system offers improved performance than the HB-MMC.



FIGURE 5.17: DC fault at the midpoint of the line in the hybrid PTP HVDC transmission system based on the LCC and the AAC: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current of LCC and AAC at point A, (e) DC current of LCC and AAC at point B, (f) AC grid voltage of AAC, (g) AC grid current of AAC, (h) SM capacitor voltage of AAC and (i) Firing angle of LCC.



FIGURE 5.18: DC fault at the midpoint of the line in the hybrid PTP HVDC transmission system based on the LCC and the HB-MMC: (a) DC voltage, (b) DC current, (c) AC grid voltage of HB-MMC, (d) AC grid current of HB-MMC and (e) SM capacitor voltage of HB-MMC.

5.6 Conclusion

This chapter develops a hybrid PTP HVDC transmission system based on the combination of an LCC and an AAC converter as the system rectifier and inverter stations, respectively. Firstly, the employed converters and control hierarchy are described for the hybrid PTP HVDC transmission system. Also, the topology of the developed system is introduced with proposed control schemes under normal/fault operations. Finally, through analytical selection of the system parameters and considering various system operating conditions, verification and validation of the performance of the hybrid LCC-AAC system considering widely accepted HVDC operating standards is performed. The detailed set of results demonstrate that the hybrid HVDC system operate under acceptable harmonic levels and within DC voltage & current ripple limits. Moreover, the system performance under various transient conditions demonstrate the robust operation including power reference tracking, AC and DC fault ride through capabilites, while ensuring prompt energy regulation and SM capacitor voltage balancing of the AAC. The results of DC fault also verify the AAC in hybrid HVDC system shows preferable DC fault handling capability compared with the HB-MMC. Development of the LCC and AAC-based hybrid PTP HVDC transmission system complements to the research and development of more complex hybrid HVDC systems combining the LCC and different modular VSCs.

Chapter 6

LCC and AAC-Based Hybrid MTDC Transmission System

6.1 Introduction

Challenges in long distance power transmission and large-scale integration of renewable energy systems drive the development of HVDC systems. Various hybrid HVDC systems can be configured as integrations of LCCs, VSCs by combining their benefits. Although each hybrid configuration poses its own challenges and requires specific control, the integration leads to unique benefits. Thus, it is necessary to explore the performance of different configurations, especially the more complex hybrid MTDC systems and future DC grids. This chapter develops the steps for integration of AACs in hybrid MTDCs and demonstrates the operation of AACs in a hybrid MTDC system. The detailed system structure and control hierarchy are developed. The performance of the hybrid MTDC system under multiple operating scenarios including AC and DC faults are demonstrated, verifying the feasibility of an LCC and AAC-based hybrid MTDC system.

6.2 Detailed System Description

6.2.1 System Topology

For further exploring the performance of the AAC in hybrid MTDC systems, the chapter develops a three-terminal hybrid HVDC system with one LCC station and two AAC stations. The detailed system topology of the hybrid MTDC system is shown in Fig. 6.1, and Fig. 6.2 describes the system control structure. The voltage level is selected as \pm 500 kV, the power

ratings for the three stations are 2000 MVA, 800 MVA and 1200 MVA, respectively, and the detailed system parameters are shown in Table 6.1. Both AACs in the inverter side adopt P/V droop control and reactive power control while the LCC controls the DC current. The basic comparison of the required number of devices in hybrid MTDC system under the same voltage level with the LCC and different modular VSCs (HB-MMCs, FB-MMCs, clamped-double (CD)-MMCs and AACs) is provided in Table 6.2. In addition, the required number of devices for the configuration of LCC with MMC (hybrid SMs) is between the configuration of the LCC with HB-MMCs and with FB-MMCs. The following sections will



FIGURE 6.1: Configuration of the LCC and AAC-based hybrid MTDC transmission system.



FIGURE 6.2: Control structure of the LCC and AAC-based hybrid MTDC transmission system.

discuss the detailed parameter selection for the LCC and AACs-based hybrid MTDC system.

Parameter	LCC	AAC1 (AAC2)
Rated Power (MVA)	2000	800 (1200)
Nominal Frequency (Hz)	50	50
DC Voltage (kV)	± 500	± 500
AC Voltage (L-L rms) (kV)	525	525
R/X Ratio	0.1	0.1
Transf. Ratio	525/217	525/702
Transf. Leakage Inductance (p.u.)	0.18	0.18
Transf. Resistance (p.u.)	0	0.006
Number of SMs per arm	-	640
Nominal SM Voltage (kV)	-	1
Stored Energy (kJ/MVA)	-	11
SM Capacitance (mF)	-	4.6 (6.9)
Arm Inductance (p.u.)	-	0.02
Nominal Operating Point (m_a)	-	1.15
DC Inductance (mH)	250	30 (25)
DC Capacitance (µF)	-	90 (60)

TABLE 6.1: Parameters of the LCC and AAC-based hybrid MTDC transmission system.

TABLE 6.2: Comparison of main modular VSCs in hybrid MTDC systems.

Parameters	LCC-MMCs (HBSM)	LCC-MMCs (FBSM)	LCC-MMCs (CDSM)	LCC-AACs (FBSM)
DC Voltage Level (kV)	±500	±500	±500	±500
Operaing Point (m_a)	0.9	0.9	0.9	1.27 (4/π)
Number of SMs (per arm)	900	900	450	640
Number of IGBTs (per arm)	1800	3600	2250	2560 (SMs)
				+ 500 (DSs)
Number of Diodes (per arm)	1800	3600	3150	2560 (SMs)
Number of Diodes (per unit)				+ 500 (DSs)
Number of Capacitors (per arm)	900	900	900	640
Energy Requirement of SMs	30~40	30~40	30~40	11
(kJ/MVA)	30.040			
Requirement of DC-Link Filters	LCC	LCC	LCC	$LCC + AACs^*$

* EO-AAC can provide active filtering for DC-link.

6.2.2 System Parameter Selection

6.2.2.1 DC Output Voltage of the LCC

The value of DC output voltage in the LCC can be calculated as (6.1). This is mainly determined by the firing angle (α) and the secondary side voltage of transformer (V_{2t}), while the influence of leakage inductance (L_l) in transformer has to be considered as well.

$$V_{dcr} = kNV_{2t}\cos(\alpha) - \frac{3N\omega L_l}{\pi}I_{dcr},$$
(6.1)

where k is $3\sqrt{2}/\pi$ for a 6-pulse thyristor bridge, N represents the number of 6-pulse converters (N = 2 in 12-pulse converter) and I_{dcr} the DC-link current of the LCC.

6.2.2.2 Power Calculation of the LCC

The power (P_{dcr}, Q_{dcr}) of DC-link and related power factor (ϕ) can be expressed as:

$$\begin{cases} P_{dcr} = V_{dcr}I_{dcr}, \\ Q_{dcr} = P_{dcr}\tan(\phi), \end{cases}$$
(6.2)

$$\phi = \cos^{-1}\left(\frac{\cos(\alpha) + \cos(\alpha + \mu)}{2}\right),\tag{6.3}$$

where μ represents the commutation overlap angle.

6.2.2.3 Operating Point and Overlap Period of the AAC

Considering the $\pm 10\%$ voltage variation under normal operation, the operating point of both AACs adopted in the hybrid system is expressed as:

$$m_{a,AACx} = \frac{M_a}{k_a} = 1.157,$$
 (6.4)

where x = 1, 2 and $k_a = 1.1$. Additionally, the short-overlap period is adopted in both AACs and can be calculated as:

$$\hat{t}_{ov,AACx} = \frac{\pi - 2\cos^{-1}(1 - m_{a,AACx})}{\omega}.$$
(6.5)

6.2.2.4 Number of SMs and Switching Devices of the AAC

The required number of SMs per arm and the switching devices per DS can be derived from:

$$\begin{cases} N_{SM,AACx} = \frac{V_{ac,AACx}}{V_{c,AACx}}, \\ N_{DS,AACx} = \frac{\hat{V}_{DS,AACx}}{V_{c,AACx}} = \frac{N_{SM}}{M_a}, \end{cases}$$
(6.6)

where $\hat{V}_{ac,AACx}$ and and $V_{c,AACx}$ represent the peak ac output voltage and the nominal SM capacitor voltage of both AACs, respectively.

6.2.2.5 SM Capacitance of the AAC

The SM capacitance of borh AACs follows the calculation in CIGRE benchmark model of MMC-HVDC [101], as:

$$C_{AACx} = \frac{S_{AACx} E_{AACx}}{3N_{SM,AACx} V_{c,AACx}^2},\tag{6.7}$$

where S_{AACx} is the rated power of both AACs and E_{AACx} is the required energy storage for AAC1 and AAC2.

6.2.2.6 Filters and DC Transmission Line

Based on the previous discussion of filters in Chapter 3.2.3 and Chapter 5.4.4. The AC filters of the LCC in the hybrid MTDC system adopt the combination of double-tuned damped filter and shunt capacitor for filtering $12k\pm1$ and low order harmonics, while the tuning points are set at 12/24 and 6/30 for DC filters [128]. The smoothing reactance in the LCC side can play the role of smoothing ripples of DC current. In addition, the smoothing reactance is also significant for both AACs which can be coordinated with DC capacitance for smoothing the 6-pulse ripple. The DC-link inductance and capacitance for the AAC can follow the top equation of (6.8). Moreover, considering the energy exchange between AC and DC, the energy storage of DC capacitor can be calculated as $C_{dc}V_{dc}^2/2 = S\Delta E_{AAC}E_{AAC}/3$ [200], then the required capacitance value at the DC-link can follow (6.8).

$$\begin{cases}
\frac{L_{dcx}C_{dcx}}{2} > \frac{1}{\omega_c^2}, \\
C_{dcx} > \frac{2Sk_eE_{AACx}}{3V_{dc}^2},
\end{cases}$$
(6.8)

where ω_c is the cut-off frequency of DC-link filters in both AACs, $k_e \in (0.9, 1.1)$ that is the energy deviation of SMs. The parameters of DC inductance and capacitance for the LCC and AACs are shown in Table 6.1.

An overhead line (OHL) is employed in the hybrid MTDC system and the equivalent lumped parameters of the OHL are shown in Table 6.3. Besides, Table 6.4 lists the related control parameters of the LCC and AACs.

transmission system.			
	Parameters of Transmission Line (Overhead Line) [101, 203		
	DC Line Voltage (kV)	± 500	
	Length (LCC/AAC1/AAC2) (km)	700/300/100	
	Resistance (Ω /km)	0.0127	
	Inductance (mH/km)	0.88	
	Capacitance (μ F/km)	0.013	

TABLE 6.3: Parameters of DC transmission line in the LCC and AAC-based hybrid MTDC transmission system.

TABLE 6.4: Control parameters in the LCC and AAC-based hybrid MTDC transmission system.

Control Parameters		
DC Current Control - LCC	$K_P = 0.3, K_I = 80$	
Active/Reactive power control - AACs	$K_P = 1.3, K_I = 33$	
Droop Constant - AACs	$k_{droop} = 0.2$	
Energy Balancing - AACs [30]	$K_P = 2.9, K_I = 75$	
PLL - LCC	$K_P = 5, K_I = 10$	
PLL - AACs	$K_P = 0.084, K_I = 4.69$	

6.3 P/V Droop Control and DC Fault-Ride-Through (FRT) Scheme

6.3.1 P/V Droop Control

The conventional P/V droop control is adopted in the hybrid MTDC system. The control of DC voltage is shared by AAC1 and AAC2 with P/V droop control because the DC voltage is determined by the power variation in both AACs via actual droop charateristic. The droop control can be considered as an extension of constant power or DC voltage control, which can improve the system reliability and reduce the telecommunication dependence among the three converters as well. The expression of droop control follows (6.9), and the droop constant is defined as (6.10) where R is the slope of droop characteristic curve.

The station will run in constant power control or DC voltage control if $k_p = 1, k_u = 0$ or $k_p = 0, k_u = 1$, respectively. The droop constant in both AAC terminals is set as 0.2 [101].

error =
$$k_p(P^* - P) + k_u(V_{dc}^* - V_{dc})$$
 (6.9)

$$k_{droop} = |R| = \frac{k_p}{k_u} \tag{6.10}$$

6.3.2 DC FRT Scheme

The LCC and AACs-based hybrid MTDC system has capability for riding through DC fault, and Fig. 6.3 shows the potential pole to pole fault current path and the proposed DC fault ride through scheme for the hybrid MTDC system.



FIGURE 6.3: DC fault ride through scheme in the LCC and AAC-based hybrid MTDC transmission system.

When the DC fault signal reaches the three converters, the LCC in the rectifier side reverts to the inverter operating mode via firing angle force retard, all SMs in both AACs are blocked and DSs are switched off at the same time. The fault is detected via monitoring the line current difference in three converters [204, 205], and the detection time is set as 0.5 ms in this hybrid HVDC system. After the line deionization, the both AACs are first de-blocked, then power and voltage references are ramped to pre-fault values. In addition, the LCC will switch into constant DC current control when the firing angle decreases to

around the steady-state value, at which point the system will recover from DC fault. It is noteworthy that it is not necessary to open the AC breakers for LCC and both AACs, since the fault current path can be blocked by the thyristors of the LCC, and both AACs can block the fault current from AC side to DC-link. The related results are shown in Chapter 6.4.4.

6.4 Simulation Results

The results of four operating scenarios are presented including i) the steady-state operation, ii) power adjustment of the LCC, iii) active power reference change of the AAC, and iv) reactive power reference change of the AAC. Besides, this section will also discuss the system performance under AC faults at both rectifier side (LCC) and inverter side (AAC1) as well as DC fault ride through capability. The detailed set of results verify the functionalities of the hybrid MTDC system and validate the system performance complying with widely accepted HVDC operating standards.

6.4.1 Steady-State Operation



FIGURE 6.4: Capacitor voltage and arm current of AAC1 in the LCC and AAC-based hybrid MTDC transmission system: (a) capacitor voltage and arm current of AAC1 and (b) capacitor voltage and arm current of AAC1.

In steady state, the LCC in rectifier side regulates the DC current (2 kA) and both AACs that operate at the predefined operating point ($M_a = 1.157$) sharing the control of DC voltage via P/V droop control. The ripples of DC voltages and currents are limited within 5% with the appropriate selection of DC-link filters. In addition, the AC voltages and currents of three terminals also comply with network operating standards, limiting the THDs below 2% [202]. The results of DC voltages and active power flow under initial



FIGURE 6.5: Capacitor voltage and arm current of AAC2 in the LCC and AAC-based hybrid MTDC transmission system: (a) capacitor voltage and arm current of AAC2 and (b) capacitor voltage and arm current of AAC2.

steady-state operation in the hybrid MTDC system are given in the operating scenario 1 of Table 6.5. The capacitor voltage and arm current of both AACs in phase *a* are respectively shown in Fig. 6.4 and Fig. 6.5. In addition, the capacitor voltages are contained within the designed $\pm 10\%$ range where applicable.

It is noteworthy that the waveforms of arm currents and capacitor voltages are different from an MMC due to the arm current interruption in one phase-leg.

Parameter	Scenario 1	Scenario 2	Scenario 3
$V_{dc,LCC}$ (p.u.)	1.029	0.963	1.061
$V_{dc,AAC1}$ (p.u.)	0.994	0.936	1.033
$V_{dc,AAC2}$ (p.u.)	0.996	0.938	1.036
$P_{ac,LCC}$ (MW)	2069.13	1455.06	1599.62
$P_{ac,AAC1}$ (MW)	-773.87	-544.79	-633.07
$P_{ac,AAC2}$ (MW)	-1177.09	-829.19	-913.36

TABLE 6.5: DC Voltages and active power flow for the LCC and AAC-based hybrid MTDC transmission system.

6.4.2 Reference Tracking

6.4.2.1 Power Adjustment of the LCC

Fig. 6.6(1) shows the results of scenario 2 (power adjustment of the LCC). The DC current reference of the LCC is changed from 1.0 p.u. to 0.75 p.u. at 0.2s (Fig. 6.6(1) (d)). The results of power and DC voltage are shown in Fig. 6.6(1) (a)-(c), respectively. In

addition, the corresponding voltages and active power flow are listed in Table 6.5. The DC voltages of all three terminals drop by approximately 6% relative to the steady-state values, and the active powers amongst the three terminals are adjusted based on the droop characteristic and DC current reference change. Moreover, the reactive powers of



FIGURE 6.6: Power adjustment of the LCC and active power reference change of AAC2 in the LCC and AAC-based hybrid MTDC transmission system: (a) active power, (b) DC voltage, (c) reactive power (d) DC current, (e) AC-grid current of LCC, (f) AC-grid current of AAC1, (g) AC-grid current of AAC2, (h) capacitor voltage of AAC1 and (i) capacitor voltage of AAC2.
both AACs keep constant due to the independent reactive power control of the AAC. The reactive power consumption of the LCC deviates with the change of active power and firing angle following with (6.2) and (6.3), which can also explain the reactive power change of the LCC in the following two scenarios.

The AC grid current change of three terminals is shown in Fig. 6.6(1) (e)-(g), which coincides with the decrease of active power. In addition, the average values of capacitor voltage in Fig. 6.6(1) (h)-(i) slightly deviate with the change of DC voltage, while the ripple of SM capacitor voltage declines with the power adjustment of the LCC.

6.4.2.2 Active Power Reference Change of AAC2

The initial DC current is 1.5 kA, regulated by the LCC, and the active power references in the both AACs are 500 MW and 1000 MW, respectively in scenario 3. At 0.2s, the active power reference of AAC2 is changed from 1.0 p.u. to 0.7 p.u (Fig. 6.6(2) (a)). The final steady state operating piont is determined by the droop controller in the both AAC terminals. Table 6.5 and Fig. 6.6(2) (a) show that the final active power of AAC2 is 0.91 p.u, which is caused by the compensation of DC voltage variation (Fig. 6.6(2) (c)). In addition, the DC current in AAC1 and AAC2 follow the change of active power (Fig. 6.6(2) (d)). The change of AC grid currents in the three terminals is shown in Fig. 6.6(2) (e)-(g), respectively. Besides, Fig. 6.6(2) (h)-(i) show that the ripple of SM capacitor voltages in both AACs slightly increases during the change of active power reference.

6.4.2.3 Reactive Power Reference Change of both AACs

Both AACs can independently control reactive power at individual station. In scenario 4, the reactive power reference of AAC1 and AAC2 is changed from 0 p.u. to -0.5 p.u. at 0.2s and 0 p.u. to 0.5 p.u. at 0.5s, respectively. The related results in Fig. 6.7 show the reactive power control capability of both AACs without interfering the LCC station. The AC grid currents of both AACs deviate slightly in the process (Fig. 6.7(f)-(g)), while with larger ripple in SM capacitor voltages (Fig. 6.7(h)-(i)).



FIGURE 6.7: Reactive power reference change of both AACs in the LCC and AAC-based hybrid MTDC transmission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current, (e) AC-grid current of LCC, (f) AC-grid current of AAC1, (g) AC-grid current of AAC2, (h) capacitor voltage of AAC1 and (i) capacitor voltage of AAC2.

6.4.3 System Performance under AC Faults

This part discusses the AC FRT capability of the LCC and AAC-based hybrid MTDC system verifying the robustness of such system. A 200ms single-line to ground (SLG) fault and three-phase short circuit fault are set at both rectifier and inverter side, and the controller of AAC2 is identical with AAC1, thus the AC faults at inverter side are only set at AAC1. In addition, the fault resistance is 0.01 Ω which can represent the worst AC faults case.

6.4.3.1 AC Faults at Rectifier Side (LCC)

Both the SLG and three-phase short circuit faults of the LCC are set at 0.2s shown in Fig. 6.8(1) and Fig. 6.8(2), respectively. During a SLG fault of the LCC, there is still redundant power for the whole system (Fig. 6.8(1) (a)), while the power transmissin is interrupted during a three-phase short circuit fault at the LCC side (Fig. 6.8(2) (a)), which are caused by DC current decrease at the LCC side (Fig. 6.8(1) (d) and Fig. 6.8(2) (d)). The reactive power oscillations of three terminals during SLG and three-phase short circuit fault are respectively shown in Fig. 6.8(1) (b) and Fig. 6.8(2) (b).

In addition, the DC voltages drop by about 0.2 p.u. (Fig. 6.8(1) (c)) and 0.25 p.u. (Fig. 6.8(2) (c)), for SLG fault and three-phase short circuit fault, respectively. Accordingly, the capacitor voltages of both AACs experience the same drop with DC voltages (Fig. 6.8(1) (i) and Fig. 6.8(2) (i)). In general, the performance of AAC1 and AAC2 is similar under AC faults of the LCC, and the AC grid currents of two stations are respectively shown in Fig. 6.8(1) (g), (h) and Fig. 6.8(2) (g), (h). The system can recover from AC faults of the LCC side within 300ms that presents the reliable AC FRT capability of sending terminal in the hybrid MTDC system.

6.4.3.2 AC Faults at Inverter Side (AAC1)

As shown in Fig. 6.9(1) and Fig. 6.9(2), the AC faults occuring at AAC1 at 0.2s lead to the DC voltage increase by approximately 0.08 p.u. (Fig. 6.9(1) (c)) and 0.15 p.u. (Fig. 6.9(2) (c)) for SLG fault and three-phase short circuit fault, respectively. Accordingly, the SM capacitor voltage of AAC1 under such two types of faults respectively increase about by 0.3 p.u. (Fig. 6.9(1) (i)) and 0.5 p.u. (Fig. 6.9(2) (i)). It is noteworthy that the deviation of SM capacitor voltage in AAC2, whether under SLG fault or three-phase short circuit fault, is less than such in AAC1 which is within 0.13 p.u. (Fig. 6.9(1) (j)) and 0.26 p.u. (Fig. 6.9(2) (j)), respectively.

The power oscillations of three terminals for two types of ac faults at AAC1 are respectively shown in Fig. 6.9(1) (a), (b) and Fig. 6.9(2) (a), (b). It can be found that the LCC side can maintain the power transmission capability since the DC current is controlled



(1) Single line to ground fault at the AC-grid of the (2) Three-phase short-circuit fault at the AC-grid of LCC. the LCC.

FIGURE 6.8: Single line to ground and three-phase short-circuit faults of the LCC in the LCC and AAC-based hybrid MTDC transmission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current, (e) AC-grid voltage of LCC, (f) AC-grid current of LCC, (g) AC-grid current of AAC1, (h) AC-grid current of AAC2 and (i) capacitor voltage of AAC1.

by the LCC while slightly drops with the increase of DC voltage. Moreover, there is no serious overcurrent occuring at AAC2 although the decline of DC current at AAC1 leads to the increase of DC current at AAC2 (Fig. 6.9(1) (d) and Fig. 6.9(2) (d)). Besides, the AC grid voltage of AAC1 and ac grid currents of three terminals are respectively shown in Fig. 6.9(1) (e)-(h) and Fig. 6.9(2) (e)-(h). The recovery time is also within 300 ms as AC faults at rectifier side, and the whole process verifies that the AAC as inverter in hybrid MTDC system can ride-through AC faults and the oscillation of current and voltage is within acceptable level.

6.4.4 Temporary Pole-to-Pole DC Fault

A temporary pole to pole DC fault via a 10 Ω resistor is set at 0.2 s, at the end of the first transmission line (700 km) and lasts for 0.1 s. After the DC fault clearing, the whole system recovers at 0.7 s with a 0.4 s line de-ionization [88].

In the hybrid MTDC system, the LCC in the rectifier side can ride through DC fault via firing angle force retard. When the DC fault occurs, the DC current in the LCC side first increases to a value of 4 kA in such hybrid system (Fig. 6.10(d)). After detecting the DC fault, the firing angle of the LCC is increased to 150° enabling the LCC to run in inverter operation, then the fault current path will be blocked by thyristors. The AC filters can provide reactive current to the AC grid of the LCC during DC fault (Fig. 6.10(g)).

The DC fault ride through scheme for the AAC is to force the fault current to zero by utilizing the series connnected capacitors in its bipolar SMs. The fault currents in the both AAC terminals will first increase following the fault current path (from positive pole to negative pole) as the SM capacitors and DC-link capacitors discharge and the fault current flows from AC side. The maximum fault currents for AAC1 and AAC2 are about 4 kA and 8 kA, respectively (Fig. 6.10(e) and (f)). For protecting the IGBTs in SMs, both converters of both AACs are blocked after 4 ms, then the fault currents in both AAC terminals will gradually decrease to zero. The oscillation of DC voltages in all three converters are due to the discharge of the DC-link capacitors and smoothing reactors (Fig. 6.10(c)). In addition, AAC is capable of blocking the fault current (Fig. 6.10(h), (i)), thus the AC breakers are not opened. Besides, the DSs in both AACs should be switched off during the DC fault for employing the antiparallel diodes in IGBTs of DSs to thoroughly cut off the fault current.

In recovery mode, both AACs first deblock and all DSs switch on. In addition, the power and voltage references for both AACs are gradually increased with a rate of 8 MW/ms for AAC1 (12 MW/ms for AAC2) and 40 kV/ms, respectively. At 0.7 s, the firing angle of the LCC is decreased to 20°, then the LCC switches to normal DC current control mode (Fig. 6.10(l)). After about 0.3 s, the whole system totally recovers from DC fault





(1) Single line to ground fault at the AC-grid of AAC1. (2) Three-phase short-circuit fault at the AC-grid of AAC1.

FIGURE 6.9: Single line to ground and three-phase short-circuit faults of AAC1 in the LCC and AAC-based hybrid MTDC transmission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current, (e) AC-grid voltage of AAC1, (f) AC-grid current of LCC, (g) AC-grid current of AAC1, (h) AC-grid current of AAC2, (i) capacitor voltage of AAC1 and (j) capacitor voltage of AAC2.



FIGURE 6.10: Pole to pole temporary DC fault in the LCC and AAC-based hybrid MTDC transmission system: (a) active power, (b) reactive power, (c) DC voltage, (d) DC current of the LCC, (e) DC current of AAC1, (f) DC current of AAC2, (g) AC-grid current of LCC, (h) AC-grid current of AAC1, (i) AC-grid current of AAC2, (j) capacitor voltage of AAC1, (k) capacitor voltage of AAC2 and (l) firing angle of LCC.

6.5 Conclusion

This chapter investigates a novel hybrid MTDC system topology combining one LCC station (two 12-pulse converters) and two AAC stations. The detailed system description is introduced first considering the system topology and parameter selection. P/V droop control is considered for both AACs and a DC FRT scheme is proposed for riding through temporary DC fault in the hybrid MTDC system. For verifying the operation of the hybrid MTDC system, simulation results are demonstrated including the steady-state operation, power adjustment of the LCC, active/reactive power reference change of the AAC, and AC/DC fault performance.

The results of four operating scenarios validate the feasibility of LCC-AAC hybrid MTDC system under P/V droop control. Moreover, the LCC-AAC hybrid MTDC system shows AC FRT capability whether SLG fault or three-phase short circuit fault at both LCC and AAC termimals, and satisfactory temporary DC fault handling performance. In general, the work of this chapter contributes to the further understanding of more complex mixed-converter DC networks in future research on DC-grids.

Chapter 7

Development of Hybrid Multi-Converter DC Grids

7.1 Introduction

Hybrid multi-converter DC grids are extended DC grids formed by combinations of different converter topologies such as the LCC, MMC, AAC and other various topologies. The stable and secure operation of such systems in steady-stage and transient conditions is critical for ensuring the robustness of hybrid multi-converter DC grids. This chapter investigates the AC and DC FRT capability and the generalized expression of DC power flow in a hybrid multi-converter DC grid consisting of LCCs, MMCs and AACs. Fault handling schemes of the LCC, MMC and AAC are combined, extended and coordinated to provide satisfactory transient response of the hybrid DC grid under both AC and DC faults. In order to ensure the static security of developed DC grid, this chapter uses the generalized expression of DC power flow described in Chapter 4 to derive the power flow after converter outage under mixed P/V and I/V droop control considering power limitation, based on the determined initial power flow for normal operation. Simulation results for steady-state and transient operation based on a detailed equivalent model verify the fault handling capability of the hybrid DC grid, and further validate the accuracy of the proposed DC power flow expression in Chapter 4.

7.2 Configuration of Hybrid Multi-Converter DC Grid

For exploring more complex DC grids combining multiple converters, a hybrid multiconverter DC grid is developed as Fig. 7.1 with DC voltage of \pm 500 kV, which functions as power delivery from remote generation zones (e.g. large-scale renewables) to load centers. In the hybrid DC grid, one 12-pulse LCC with power rating of 3000 MVA in the rectifier side, via a combination of OHL and cables, delivers power to both AACs and one MMC with power rating of 900 MVA, 1200 MVA and 1000 MVA, respectively. The detailed parameters of the DC transmission lines are given in Table 7.1.



FIGURE 7.1: Topology of the hybrid multi-converter DC grid.

Parameters	Line 1	Line 2	Line 3	Line 4	Line 5
Туре	OHL	OHL	OHL	Cable	Cable
Distance	600	200	100	100	150
(km)					
Resistance	0.0127	0.0127	0.0127	0.011	0.011
(Ω/km)					
Capacitance	ce 0.013	0.013 (0.013	0.2185	0.2185
(μ F/km)					
Inductance	0.88	0.88	0.88	0.2615	0.2615
(mH/km)					

TABLE 7.1: Parameters of DC transmission lines in the hybrid multi-converter DC grid.

7.2.1 Description of the LCC

The basic high level control modes of an LCC are i) constant DC current or voltage control and ii) constant firing angle or extinction angle control. Other control modes for LCCs can be developed based on the basic control modes as described in Chapter 3.2.2, e.g. constant power, frequency and damping control. For LCCs in the inverter side, VDCOL is usually adopted for reducing the risk of commutation failures [128].

The LCC in the hybrid DC grid is located at rectifier side and operates with constant DC current control. The active and reactive power for LCC is $P_{dc_{LCC}} = V_{dc_{LCC}} I_{dc_{LCC}}$ and $Q_{LCC} = P_{dc_{LCC}} \tan \phi$ (ϕ is the power factor angle), respectively. The detailed parameters of LCC are shown in Table 7.2.

	Parameter	LCC
ſ	Rated Power (MVA)	3000
	DC Voltage (kV)	± 500
	AC Voltage (L-L rms) (kV)	525
	Transformer Ratio	525/220

TABLE 7.2: Parameters of LCC in the hybrid multi-converter DC grid.

7.2.2 Description of the MMC and the AAC

MMCs and AACs all belong to the modular VSC family, thus high level control is similar while with different low level control. The system parameters of MMC and both AACs in the hybrid DC grid are shown in Table 7.3. For the high-level control, P/V or I/V droop control is adopted for three inverters, thus the DC voltage of the hybrid dc grid is balanced by MMC and both AACs. Moreover, the reactive power of MMC and both AACs is set as zero.

Parameter	MMC	AAC1 (AAC2)
Rated Power (MVA)	1000	900 (1200)
DC Voltage (kV)	± 500	± 500
AC Voltage (L-L rms) (l	xV) 525	525
Transformer Ratio	525/551	525/702
Number of SMs per ar	m 1000	640
Nominal SM Voltage (l	xV) 1	1
Arm Inductance (p.u.	.) 0.15	0.02
Nominal Operating Point	(<i>m_a</i>) 0.9	1.15
DC Inductance (mH)) 30	80 (75)
DC Capacitance (μ F)) –	100 (110)

TABLE 7.3: Parameters of MMC and AACs in the hybrid multi-converter DC grid.

7.3 AC and DC Fault Handling Scheme

This section discusses the combined temporary AC and DC fault handling schemes in the hybrid multi-converter DC grid. Special emphasis is given on AC faults at inverter side,

since AC faults at the LCC (rectifier) side will lead to power interruption of the whole proposed system, similarly to [49, 50, 88].

7.3.1 AC Fault Handling Scheme

When temporary AC faults occur at either the MMC or AAC stations, the hybrid DC grid should be capable of riding through these AC faults. In the FRT scheme, the outer control loop is temporarily disabled in order to assist the response of the converter, due to the fast response of the inner current controller compared to the outer loop. The controller will be re-enabled when receiving the clearing signal of the AC faults. Fig. 7.2 shows the AC FRT control scheme when a temporary AC fault occurs at the MMC side of the hybrid DC grid.



FIGURE 7.2: AC fault ride through scheme in the hybrid multi-converter DC grid.

7.3.2 DC Fault Handling Scheme

The hybrid multi-converter DC grid should also be capable of riding-through DC faults. When a DC fault is detected (Fig. 7.3), the firing angle of LCC is increased to above 90° reverting to inverter operating mode, the MMC and AACs are blocked and DSs in AACs are switched off at the same time. Also, the power (or current) and voltage references of MMC and AACs are decreased to zero. Besides, it is necessary to open the AC breakers of the MMC after converter blocking since the MMC with HBSMs cannot self-extinguish the fault current. After the necessary line deionization, the firing angle of LCC will be

gradually decreased, the MMC and both AACs are de-blocked and all DSs are switched on. Meanwhile, the AC breakers of MMC will be re-closed, then power (or current) and voltage references are ramped to the pre-fault values. In addition, LCC will switch into constant DC current control when the firing angle decreases to around the steady-state value, at which point the hybrid multi-converter DC grid will recover from the DC fault.



FIGURE 7.3: DC fault ride through scheme in the hybrid multi-converter DC grid.

7.4 DC Power Flow in the Hybrid Multi-Converter DC Grid

Two control configurations of the four-terminal hybrid DC grid are analyzed in this section. In the first configuration (C1), the LCC in the sending end controls the DC current as constant (2 kA) while are three inverters in the receiving end adopt P/V droop control. In

the second configuration (C2), the control mode of AAC2 changes to I/V droop control. The control modes of C1 and C2 all belong to mixed droop control.

7.4.1 DC Power Flow under Normal Operation

The initial power references for AAC1, AAC2 and MMC are -500 MW, -900 MW (or -0.9 kA) and 1000 kV, respectively (Table 7.4). Moreover, the AC reactive powers for both AACs and MMC are all set as zero.

Parameter		LCC	AAC1	AAC2	MMC
		(T1)	(T2)	(T3)	(T4)
I^* $(l_z \Delta)$	C1	2	-	-	-
I_{dc} (KA)	C2	2	-	-0.9	-
	C1	-	-	-	1000
	C2	-	-	-	1000
	C1	-	-500	-900	-
	C2	-	-500	-	-
δ_{droop}		-	0.05	0.05	0.05
K_{droop}		0	18	24*	20

TABLE 7.4: References for high-level control in the hybrid multi-converter DC grid.

 $^{\ast}24$ MW/kV for P/V droop control and 24 A/kV for I/V droop control.

Following with the initial power flow calculation method in Chapter 4.4.1, the DC current directions in each branch (Fig. 7.4) can be obtained by assuming the initial current references as power references (2000 MW for LCC and 900 MW for AAC2 with I/V droop control), and the coefficient Λ is:

$$\boldsymbol{\Lambda} = \begin{bmatrix} 0 & 1 & 1 & 0 & 0 \\ 0 & -1 & 0 & 1 & 0 \\ 0 & 0 & -1 & 0 & 1 \\ 0 & 0 & 0 & -1 & -1 \end{bmatrix}.$$
 (7.1)

Moreover, the line conductance matrix (G_l) of the DC grid is $[0.1312, 0.3937, 0.7874, 0.9091, 0.6061]^{T}$.

Substituting the known values into (4.42), the initial DC power flow can be derived (Table 7.5) which also includes all DC current values. In addition, the LCC in the rectifier adopts constant DC current control, thus the droop constant is zero. The DC voltage deviation ratio is 0.05, and the droop constants for the three terminals are set as 18, 24 and 20, respectively.



FIGURE 7.4: Detailed current directions of initial state (black) and after MMC outage (red) in the hybrid multi-converter DC grid.

TABLE 7.5: Calculation of initial DC power flow for normal operation in the hybrid multiconverter DC grid.

Parameter		LCC	AAC1	AAC2	MMC
		(T1)	(T2)	(T3)	(T4)
I_{dc}	C1	2.0000	-0.4997	-0.8991	-0.6012
(kA)	C2	2.0000	-0.4997	-0.9000	-0.6003
V_{dc}	C1	1034.5333	1000.6561	1000.9999	1000.0000
(kV)	C2	1034.5319	1000.6555	1000.9979	1000.0000
P_{dc}	C1	2069.0666	-500.0000	-900.0000	-601.2269
(MW)	C2	2069.0638	-500.0000	-900.8981	-600.3275
Iline	I.	T.	I	I	I
(kA)	1/1	112	113	114	115
C1	2.0000	0.7979	1.2021	0.2982	0.3030
C2	2.0000	0.7976	1.2024	0.2979	0.3024

7.4.2 DC Power Flow after MMC Outage

When the MMC in the 4-terminal hybrid DC grid is isolated due to a permanent AC fault, the DC power is re-distributed in the remaining three converters (LCC and both AACs). The current topology configuration is shown in Fig. 7.4, following with the current directions in each branch (red lines) obtained by single P/V droop control assumption. The current coefficient Λ' is:

$$\boldsymbol{\Lambda}' = \begin{bmatrix} 0 & 1 & 1 & 0 \\ 0 & -1 & 0 & -1 \\ 0 & 0 & -1 & 1 \end{bmatrix},$$
(7.2)

and the line conductance matrix (G'_l) is $[0.1312, 0.3937, 0.7874, 0.3636]^T$.

It is necessary to determine the power limitation of other converters after MMC outage. Combining (4.38), (4.60) and (4.39)-(4.62), the DC power deviations for C1 in LCC, AAC1 and AAC2 are 30.5150 MW, -271.2725 MW and -360.5483 MW, respectively, which shows AAC2 is overloaded. For calculating the power flow accurately, AAC2 should be in constant DC power control ($K_{droop,AAC2} = 0$) and the final DC power flow can be obtained by further substituting $\Delta P_{dc_{AAC2}}$ ($\Delta P_{dc_{AAC2}} = -300$ MW) into (4.61). For C2, the DC power of AAC2 is -1268.0533 MW (eq. (4.58)) which also exceeds the maximum power limitation, thus AAC2 switches into constant DC power control. The obtained power flows for the two control configurations after the MMC outage are given in Table 7.6.

	Dono	motor	LCC	AAC1	AAC2
Farameter		(T1) (T2)		(T3)	
	I_{dc}	C1	2.0000	-0.8236	-1.1764
	(kA)	C2	2.0000	-0.8236	-1.1764
	V_{dc}	C1	1053.7257	1019.5249	1020.0261
	(kV)	C2	1053.7250	1019.5243	1020.0254
	P_{dc}	C1	2107.4514	-839.6395	-1200.0000
	(MW)	C2	2107.4501	-839.6381	-1200.0000
	K_d	roop	0	18	0
	I _{line} (kA)	I_{l1}	I_{l2}	I_{l3}	I_{l4}/I_{l5}
	C1	2.0000	0.7324	1.2676	0.0911
	C2	2.0000	0.7324	1.2676	0.0911

TABLE 7.6: Calculation of DC power flow after MMC outage in the hybrid multi-converter DC grid.

7.5 Simulation Results

This section provides verification of the power flow generalized calculation results for the hybrid DC grid via simulation in PLECS-Blockset and Simulink. The temporary AC and DC FRT capabilities of the hybrid DC grid are also demonstrated with control configuration 1 based on the combined fault handling schemes.

7.5.1 Power Flow Verification

Table 7.7 shows the simulation results of the initial DC power flow for two control configurations (C1 and C2) under mixed droop control. The results validate the precise initial DC power flow calculation.

The DC power and current of MMC drops to zero due to the converter outage, thus the DC power flow is re-distributed within LCC, AAC1 and AAC2. The power flow simulation results of the remaining three converter are summarized in Table 7.8. The simulation results are in accordance with the calculation results of Table 7.6 validating the accuracy of the power flow expression after the MMC outage.

Parameter		LCC	AAC1	AAC2	MMC	
		(T1)	(T2)	(T3)	(T4)	
I_{dc}	C1	2.0002	-0.4997	-0.8992	-0.6013	
(kA)	C2	2.0000	-0.4996	-0.8999	-0.6003	
V_{dc}	C1	1034.5333	1000.6561	1001.0000	999.9999	
(kV)	C2	1034.5319	1000.6557	1000.9979	1000.0000	
P_{dc}	C1	2069.0667	-500.0000	-900.0000	-601.2270	
(MW)	C2	2069.0640	-500.0002	-900.8980	-600.3275	
Iline	т	T	T	T	Т	
(kA)	1/1	112	1/3	114	115	
C1	2.0002	0.7980	1.2022	0.2983	0.3030	
C2	2.0000	0.7975	1.2025	0.2979	0.3024	

TABLE 7.7: Simulation results of initial DC power flow for normal operation in the hybrid multi-converter DC grid.

TABLE 7.8: Simulation results of DC power flow after MMC outage in the hybrid multiconverter DC grid.

Parameter		LCC	AAC1	AAC2
		(T1)	(T2)	(T3)
I_{dc} C1		2.0000	-0.8236	-1.1764
(kA)	C2	1.9998	-0.8236	-1.1765
V_{dc}	C1	1053.7257	1019.5250	1020.0260
(kV)	C2	1053.7250	1019.5243	1020.0254
P_{dc}	C1	2107.4514	-839.6395	-1200.0001
(MW) C2		2107.4501	-839.6380	-1199.9999
K _{droop}		0	18	0
Iline	Т	T	Т	τ / τ
(kA) ¹ 11		112	113	I_{l4}/I_{l5}
C1	2.0000	0.7324	1.2676	0.0911
C2	1.9998	0.7324	1.2674	0.0910

Fig. 7.5 shows the waveforms of DC powers, voltages, currents and five line currents for normal operation (0 \sim 1 s) and after the MMC outage (1 s \sim 5 s) with C1 in the hybrid DC grid. The waveforms of C2 are not shown as the control modes are identical for both

control configurations after an MMC outage. It is noted that the sudden increase of DC voltages in the remaining converters is due to the instantaneous power shortage at the inverter side.



FIGURE 7.5: Steady-state and transient waveforms before and after MMC outage with C1 in the hybrid multi-converter DC grid: (a) DC power, (b) DC voltage, (c) DC current and (d) line currents.

7.5.2 AC FRT Capability

Based on the temporary AC fault handling scheme in Chapter 7.3.1, Fig. 7.6 shows the results of a 0.2 s three-phase short-circuit fault of MMC occurring at 0.2 s. Active and reactive power variations in the four terminals of the AC-grid are shown in Fig. 7.6(a) and (b), respectively. The active power and DC current of MMC decreases to zero during AC fault, thus leading to the oscillation of the active powers and DC currents (Fig. 7.6(f)) in LCC, AAC1 and AAC2. Accordingly, the line currents in the five branches (Fig. 7.6(g)) also oscillate with the DC current oscillation that i_{l1} (or DC current in LCC) keeps almost

constant, i_{l2} decreases to below 0.5 kA, i_{l3} increases to above 1.5 kA, i_{l4} decreases and i_{l5} increases but the current direction is changed.

In addition, the DC voltages of the four terminals and SM capacitors of both AACs and MMC all increase on different levels during the AC fault of MMC. The increase of DC voltages in four terminals is within 0.07 p.u. (Fig. 7.6(e)), while the deviation of AAC1, AAC2 and MMC is within 0.13 p.u., 0.2 p.u. and 0.15 p.u., respectively (Fig. 7.6(h)-(j)). In general, the hybrid DC grid shows satisfactory AC FRT capability with the proposed AC fault control scheme.

7.5.3 DC FRT Capability

At 0.2 s, a 0.1 s pole to pole DC fault with resistance of 10 Ω occurring at the end of OHL 1 as Fig. 7.3. The system recovers from DC fault after fault clearance and 0.4 s line de-ionization. Following the DC fault handling scheme of Chapter 7.3.2 (Fig. 7.7(a), (b)), the firing angle of the LCC is increased to 150° after 4 ms, extinguishing the fault current in the rectifier side (Fig. 7.7(l)). The SMs at the inverters, meanwhile, are blocked, DSs of AACs are switched off and DC voltage and power references decrease to zero (Fig. 7.7(c), (d)). In addition, the AC breakers of MMC open after 60 ms for protecting the SMs of MMC (Fig. 7.7(e)). It is noteworthy that thyristors are required in HBSMs for protecting antiparallel diodes during the 60 ms. At 0.7 s, the firing angle of LCC is gradually decreased to the pre-fault value, all SMs and DSs of AACs and MMC are debloked and switched on, respectively. Then, the AC breakers of MMC are closed, while the references of powers and DC voltages of AACs and MMC are ramped from zero to pre-fault values.

During the DC fault, the DC voltages in four terminals are gradually decreased to zero due to the energy reduction in the DC-link (Fig. 7.7(f)), and the oscillation of DC currents is caused by the discharging of DC-link capacitors (Fig. 7.7(g), (h)). The SM capacitor voltages of both AACs are slightly higher after converter blocking due to the energy absorption from AC-side to DC-link, while the SM capacitor voltages of MMC are not affected. In the whole DC fault handling process, the deviation of SM capacitor voltages is regulated within the range of 0.3 p.u. (Fig. 7.7(i)-(k)), and the DC grid returns to normal operation within 0.5 s. The results of DC fault also demonstrate the difference between AAC and MMC that the AAC shows the potential of DC fault self-clearing capability while MMC with HBSMs has to rely on breakers handling DC fault.



FIGURE 7.6: Temporary three-phase short-circuit fault at MMC in the hybrid multiconverter DC grid: (a) AC active power, (b) AC reactive power, (c) AC voltages of MMC, (d) AC currents of MMC, (e) DC voltages, (f) DC currents, (g) Line currents, (h) SM capacitor voltages of AAC1, (i) SM capacitor voltages of AAC2 and (j) SM capacitor voltages of MMC.



FIGURE 7.7: Temporary pole to pole DC fault in the hybrid multi-converter DC grid: (a) AC active power, (b) AC reactive power, (c) AC currents of AAC1, (d) AC currents of AAC2, (e) AC currents of MMC, (f) DC voltages, (g) DC currents, (h) line currents, (i) SM capacitor voltages of AAC1, (j) SM capacitor voltages of AAC2, (k) SM capacitor voltages of MMC and (l) firing angle of LCC.

7.5.4 Discussion

Excellent AC and DC FRT capability is of paramount importance for future complex multiconverter DC grids. The simulation results of AC and DC faults in the developed hybrid multi-converter DC grid show the combined AC and DC fault handling schemes can ensure the safe operation of the hybrid DC grid. The oscillations of corresponding powers, voltages, currents during faults are all within acceptable range, and the fault recovery stage is stable and rapid.

Additionally, the power flow analysis can be used to assess the static security of multiconverter DC grids. The proposed generalized expression of DC power flow in Chapter 4.4 first determines the initial power flow for normal operation at steady-state operating point under mixed P/V and I/V droop control (Fig. 4.11). Then, the power flow of post contingency (converter outage) operation is derived considering the power limitation (Fig. 4.12), which assesses the power flow in the remaining converters precisely. Since the power flow of post fault (temporary AC and DC fault) operation should restore to the initial power flow, the fault recovery level can be estimated by the determined initial DC power flow.

7.6 Conclusion

The AC and DC FRT capability combined with proposed DC power flow generalized expression under mixed P/V and I/V droop control are investigated in the LCC, MMC and AACs-based hybrid DC grid. The correlation between simulation-based DC power flow results and the theoretical power flow calculations for initial and post contingency scenarios verify and validate the performance of the hybrid DC grid. The results of temporary AC and DC fault demonstrate the robustness of the hybrid system under the combined fault handling schemes. Moreover, the power transmission can rapidly recover from faults, and the power flow is restored to the initial state. In general, hybrid multi-converter DC grid and its operation are valid steps towards investigation of more complex DC-grids, and the DC power flow analysis can be utilized for effective reconfiguration of complex DC grids under post fault and contingency operation.

Chapter 8

Conclusions and Future Work

8.1 Conclusions

The future trend for HVDC market would be the combination of multiple converters to constitute complex MTDC systems and DC grids. This thesis focuses on the development of hybrid HVDC system, MTDC system, DC grid combining multiple technologies (LCC, VSC) and topologies (MMC, AAC and other various topologies). The work of this thesis is summarized as below:

- LCC is the most mature technology and many HVDC projects have been commissioned in the world, while the majority are PTP systems. VSC overcomes commutation failures and is more suitable for the formation of MTDC systems. The modular VSC topologies, including the MMC with various SMs, AAC and other hybrid converter configurations, provide improved performances than conventional VSC-based HVDC. The emerging AAC possesses DC fault self-clearing capability and reduces the energy storage in SM capacitors. Also, the integration of LCCs at the sending ends and different modular VSCs at the receiving ends benefits the remote power delivery and flexible power flow regulation. Three different hybrid HVDC systems are developed in Chapters 5, 6 and 7 for demonstrating the feasibility of combination of multiple technologies and topologies.
- HVDC benchmark models can provide reference basis and allow comparison of system performances for future study of different HVDC systems. From the summarization of the available benchmark models for HVDC converters, systems and DC grid studies, the research gaps, data availability and expansion capability of these benchmark models are addressed in Chapter 2 facilitating the future development of more accurate and relevant HVDC benchmark models.

- The basic topologies, operations and control schemes of the LCC, MMC and AAC are reviewed in Chapter 3 with comprehensive theory analysis for the development of hybrid HVDC systems. Besides, the EMT, electromechanical transient, dynamic frequency capturing modeling approaches and co-simulation studies for HVDC systems are discussed in detail. The detailed equivalent models based on the EMT modeling method are adopted for simulations in Chapters 4 7 of this thesis.
- A generalized expression of DC power flow is proposed in Chapter 4 to ensure the static security of MTDC systems and DC grids under mixed P/V and I/V droop control. The generalized expression is the extension of the DC power flow calculation methods of single P/V and I/V droop control. Results derived from the theoretical analysis of the initial DC power flow for normal operation and the DC power flow after converter outage are consistent with the simulation results in an MMC- and further developed LCC, MMC and AAC-based DC grids.
- For exploring integrations of different HVDC technologies and topologies, an LCC and AAC-based hybrid PTP system is first developed in Chapter 5 with detailed control hierarchy description and parameter selection. The simulation results of steady-state operation, reference tracking and fault handling performance of AC and DC faults show the robustness of the hybrid HVDC system. Also, the AAC shows better DC fault handling capability in the hybrid HVDC system than typical HB-MMC. Although the MMC with FBSMs can also tolerate DC fault, the required number of SMs in the AAC is less than the MMC with HBSMs and FBSMs.
- By extending the hybrid LCC-AAC PTP HVDC system, a hybrid MTDC system with one LCC and two AACs under P/V droop control is further demonstrated in Chapter 6 to verify the feasibility of the AAC in hybrid MTDC systems. Presented simulation results of multiple operating scenarios and fault performance of AC and DC faults validate AACs can be in receiving ends to flexibly absorb the power delivered from the LCC.
- Combining the conventional LCC, state-of-the-art MMC and emerging AAC, a hybrid DC grid with the three converters is finally presented in Chapter 7 investigating hybrid multiple DC grids constituted by different converter topologies. The impact of transient AC and DC faults is examined in the hybrid multi-converter DC grid following with the FRT combination schemes. In general, the different hybrid HVDC systems developed in the thesis lay the preliminary fundation for future more complex MTDC systems and DC super grids combining different converters from multiple vendors.

8.2 Future Work

This section introduces the suggested future work based on the work in this thesis.

- Development of other hybrid HVDC systems considering the connection of the AAC to passive networks and the commutation support of the AAC to the LCC operating in inverter mode.
- Exploration of detailed fault location and protection for MTDC systems and more complex multi-converter DC grids.
- Research on fault analysis and calculations for different modular VSCs and hybrid multi-converter DC grids.
- Implementation of developed HVDC systems in real-time digital simulators focusing on the system of systems integration, protection validation, hardware testing and the path to a virtual digital twin.
- Further study on the application of LCC, MMC and AAC for DC systems at medium voltage level for large-scale renewable energy integration both as distribution networks and as collection grids.

Bibliography

- [1] M. P. Bahrman and B. K. Johnson, "The ABCs of HVDC transmission technologies," *IEEE Power and Energy Mag.*, vol. 5, no. 2, pp. 32–44, Mar. 2007.
- [2] N. Flourentzou, V. G. Agelidis, and G. D. Demetriades, "VSC-based HVDC power transmission systems: An overview," *IEEE Trans. Power Electron.*, vol. 24, no. 3, pp. 592–602, Mar. 2009.
- [3] M. Barnes, D. Van Hertem, S. P. Teeuwsen, and M. Callavik, "HVDC systems in smart grid," *Proc. of the IEEE*, vol. 105, no. 11, pp. 2082–2098, Nov. 2017.
- [4] R. Marquardt, "Modular multilevel converter: An universal concept for HVDCnetworks and extended DC-bus-applications," in *The 2010 Int. Power Electron. Conf.* - *ECCE ASIA* -, Jun. 2010, pp. 502–507.
- [5] S. Allebrod, R. Hamerski, and R. Marquardt, "New transformerless, scalable modular multilevel converters for HVDC transmission," in 2008 IEEE Power Electron. Specialists Conf., Jun. 2008, pp. 174–179.
- [6] G. Li, J. Liang, T. Joseph, T. An, J. Lu, M. Szechtman, B. R. Andersen, and Q. Zhuang, "Feasibility and reliability analysis of LCC DC grids and LCC/VSC hybrid DC grids," *IEEE Access*, vol. 7, pp. 22445–22456, Feb. 2019.
- [7] RTS@UNSW, "LCC & VSC-HVDC projects," https://hvdc.shinyapps.io/scatterplot/, Last Updated Feb., 2019.
- [8] H. Huang, M. Uder, R. Barthelmess, and J. Dorn, "Application of high power thyristors in HVDC and FACTS systems," in 17th Conf. of Elect. Power Supply Ind. (CEPSI), 2008, pp. 1–8.
- [9] ABB, "Evolution of HVDC light," https://new.abb.com/news/detail/4224/ evolution-of-hvdc-lightr, accessed Apr. 16, 2018.
- [10] M. O. M. E. Abdelsalam, "Capacitor voltage balancing, fault detection, and fault tolerant control techniques of modular multilevel converters," Ph.D. dissertation, Staffordshire University, 2016.

- [11] Z. Liu, F. Zhang, J. Yu, K. Gao, and W. Ma, "Research on key technologies in ± 1100 kV ultra-high voltage DC transmission," *High Voltage*, vol. 3, no. 4, pp. 279–288, Dec. 2018.
- [12] A. Lesnicar and R. Marquardt, "An innovative modular multilevel converter topology suitable for a wide power range," in 2003 IEEE Bologna Power Tech. Conf. Proc.,, vol. 3, Jun. 2003, pp. 6 pp. Vol.3–.
- [13] H. Knaak, "Modular multilevel converters and HVDC/FACTS: a success story," in Proc. of the 2011 14th European Conf. on Power Electron. and Appl., 2011, pp. 1–6.
- [14] C. Chen, G. P. Adam, S. Finney, J. Fletcher, and B. Williams, "H-bridge modular multi-level converter: control strategy for improved DC fault ride-through capability without converter blocking," *IET Power Electron.*, vol. 8, no. 10, pp. 1996–2008, Oct. 2015.
- [15] G. P. Adam, S. J. Finney, and B. W. Williams, "Enhanced control strategy of fullbridge modular multilevel converter," in 2015 Int. Conf. on Renewable Energy Res. and Appl. (ICRERA), Nov. 2015, pp. 1432–1436.
- [16] M. B. Ghat, A. Shukla, and E. C. Mathew, "A new hybrid modular multilevel converter with increased output voltage levels," in 2017 IEEE Energy Convers. Cong. and Expo. (ECCE), Oct. 2017, pp. 1634–1641.
- [17] W. Lin, D. Jovcic, S. Nguefeu, and H. Saad, "Full bridge MMC converter optimal design to HVDC operational requirements," in 2016 IEEE Power and Energy Soc. General Meeting (PESGM), Jul. 2016, pp. 1–1.
- [18] R. Li, G. P. Adam, D. Holliday, J. E. Fletcher, and B. W. Williams, "Hybrid cascaded modular multilevel converter with DC fault ride-through capability for the HVDC transmission system," *IEEE Trans. Power Del.*, vol. 30, no. 4, pp. 1853–1862, Aug. 2015.
- [19] R. Li, Z. X. Wang, D. Holliday, L. Z. Yao, and B. W. Williams, "DC transformer based on hybrid cascaded MMC in multi-terminal HVDC transmission system," in 8th IET Int. Conf. on Power Electron., Mach. and Drives (PEMD 2016), Apr. 2016, pp. 1–6.
- [20] C. Wang, J. Meng, and Y. Wang, "Modified carrier phase-shifted SPWM for a hybrid modular multilevel converter with HBSMs and FBSMs," in 2016 IEEE 8th Int. Power Electron. and Motion Control Conf. (IPEMC-ECCE Asia), May 2016, pp. 1123–1127.
- [21] J. Jung, S. Cui, and S. Sul, "A new topology of multilevel VSC converter for hybrid HVDC transmission system," in 2016 IEEE Applied Power Electron. Conf. and Expo. (APEC), Mar. 2016, pp. 2620–2628.

- M. M. C. Merlin, T. C. Green, P. D. Mitcheson, D. R. Trainer, R. Critchley, W. Crookes, and F. Hassan, "The alternate arm converter: A new hybrid multi-level converter with DC-fault blocking capability," *IEEE Tran. Power Del.*, vol. 29, no. 1, pp. 310–317, Feb. 2014.
- [23] H. R. Wickramasinghe, G. Konstantinou, J. Pou, and V. G. Agelidis, "Asymmetric overlap and hysteresis current control of zero-current switched alternate arm converter," in *IECON 2016 - 42nd Annual Conf. of the IEEE Ind. Electron. Society*, Oct. 2016, pp. 2526–2531.
- [24] J. Li, H. R. Wickramasinghe, G. Konstantinou, J. Pou, and X. Jin, "Limitations of overlap control for energy balancing of the alternate arm converter under imbalanced AC grid conditions," in 2017 Asian Conf. on Energy, Power and Transp. Electrific. (ACEPT), Oct. 2017, pp. 1–6.
- [25] H. R. Wickramasinghe, G. Konstantinou, Z. Li, and J. Pou, "Development of an alternate arm converter benchmark model for HVDC applications," in *IECON 2017* 43rd Annual Conf. of the IEEE Ind. Electron. Society, Oct. 2017, pp. 4506–4511.
- [26] H. R. Wickramasinghe, G. Konstantinou, and J. Pou, "Comparison of bipolar submodules for the alternate arm converter," *Elect. Power Syst. Res.*, vol. 146, pp. 115– 123, May 2017.
- [27] G. Konstantinou, H. R. Wickramasinghe, S. Ceballos, and J. Pou, "Submodule voltage balancing and loss equalisation in alternate arm converters based on virtual voltages," in 2018 Int. Power Electron. Conf. (IPEC-Niigata 2018 -ECCE Asia), May 2018, pp. 3117–3122.
- [28] H. R. Wickramasinghe, G. Konstantinou, and J. Pou, "Zero-current switching for the alternate arm converter through on-load tap changers," in 2018 IEEE Energy Convers. Congr. and Expo. (ECCE), Sep. 2018, pp. 1342–1347.
- [29] H. R. Wickramasinghe, G. Konstantinou, S. Ceballos, and J. Pou, "Alternate arm converter energy balancing under parameter variation," *IEEE Trans. Power Electron.*, vol. 34, no. 4, pp. 2996–3000, Apr. 2019.
- [30] H. R. Wickramasinghe, G. Konstantinou, and J. Pou, "Gradient-based energy balancing and current control for alternate arm converters," *IEEE Trans. Power Del.*, vol. 33, no. 3, pp. 1459–1468, Jun. 2018.
- [31] H. R. Wickramasinghe, G. Konstantinou, J. Pou, R. Picas, S. Ceballos, and V. G. Agelidis, "Comparison of bipolar sub-modules for the alternate arm converter," in *IECON 2016 - 42nd Annual Conf. of the IEEE Ind. Electron. Society*, Oct. 2016, pp. 6482–6487.

- [32] H. R. Wickramasinghe, W. Wang, K. Ma, and G. Konstantinou, "Thermal characterization of the alternate arm converter for HVDC applications," in 2019 IEEE Int. Conf. on Ind. Technol. (ICIT), Feb. 2019, pp. 556–561.
- [33] H. R. Wickramasinghe, G. Konstantinou, and J. Pou, "Extended ZCS region of AAC-HVDC systems through on-load tap changer coordination," *IEEE Trans. Ind. Electron.*, vol. 68, no. 7, pp. 5497–5506, Jul. 2021.
- [34] O. Jasim, F. Moreno, D. Trainer, R. Feldman, E. Farr, and J. Clare, "Hybrid experimental setup for alternate arm converter and modular multilevel converter," in 13th IET Int. Conf.on AC and DC Power Transmission (ACDC 2017), May 2017, pp. 1–6.
- [35] E. M. Farr, R. Feldman, J. C. Clare, A. J. Watson, and P. W. Wheeler, "The alternate arm converter (AAC)—"short-overlap" mode operation—analysis and design parameter selection," *IEEE Trans. Power Electron.*, vol. 33, no. 7, pp. 5641–5659, Jul. 2018.
- [36] V. Najmi, R. Burgos, and D. Boroyevich, "Design and control of modular multilevel alternate arm converter (AAC) with zero current switching of director switches," in 2015 IEEE Energy Convers. Congr. and Expo. (ECCE), Oct. 2015, pp. 6790–6797.
- [37] F. J. Moreno, M. M. C. Merlin, D. R. Trainer, K. J. Dyke, and T. C. Green, "Control of an alternate arm converter connected to a star transformer," in 2014 16th Eur. Conf. on Power Electron. and Appl., Sep. 2014, pp. 1–10.
- [38] B. Fan, K. Wang, Y. Li, L. Xu, and Z. Zheng, "Module-capacitor voltage fluctuation optimization control for an alternate arm converter," in 2015 IEEE Energy Convers. Congr. and Expo. (ECCE), Oct. 2015, pp. 3326–3330.
- [39] S. Debnath, P. R. Marthi, and J. Sun, "Advanced modeling fast simulation algorithms for alternate arm converters," in 2018 IEEE Electron. Power Grid (eGrid), Jan. 2018, pp. 1–6.
- [40] H. Wang, "Real-time and non-real-time EMT simulation of an alternate arm converter," Master's thesis, University of Manitoba, 2017.
- [41] H. R. Wickramasinghe, "Modeling, control, and benchmarking of multiple modular vsc topologies in multiterminal dc-grids," Ph.D. dissertation, University of New South Wales, 2018.
- [42] Y. Y. Leow and C. A. Ooi, "Arm energy balancing control in AACs: a comparative analysis," *IET Power Electron.*, vol. 13, no. 6, pp. 1113–1128, May 2020.

- [43] F. Gruson, P. Vermeersch, X. Guillaud, and P. Egrot, "Energy control for the alternate arm converter," in *2017 IEEE Manchester PowerTech*, Jul. 2017, pp. 1–6.
- [44] S. Liu, M. Saeedifard, and X. Wang, "Zero-current switching control of the alternate arm HVDC converter station with an extended overlap period," *IEEE Trans. Ind. Electron.*, vol. 66, no. 3, pp. 2355–2365, Mar. 2019.
- [45] M. M. C. Merlin, D. Soto-Sanchez, P. D. Judge, G. Chaffey, P. Clemow, T. C. Green, D. R. Trainer, and K. J. Dyke, "The extended overlap alternate arm converter: A voltage-source converter with DC fault ride-through capability and a compact design," *IEEE Trans. Power Electron*, vol. 33, no. 5, pp. 3898–3910, May 2018.
- [46] P. Vermeersch, F. Gruson, M. M. Merlin, X. Guillaud, and P. Egrot, "A stored energy control based active dc filter for the alternate arm converter with an extended overlap period," in 2019 21st European Conference on Power Electronics and Applications (EPE '19 ECCE Europe), Nov. 2019, pp. 1–10.
- [47] H. Yang, S. Fan, Y. Dong, H. Yang, W. Li, and X. He, "Arm phase-shift conducting modulation for alternate arm multilevel converter with half-bridge submodules," *IEEE Trans. Power Electron.*, vol. 36, no. 5, pp. 5223–5235, May 2021.
- [48] P. Vermeersch, F. Gruson, M. M. C. Merlin, X. Guillaud, and P. Egrot, "Full energy management of EO-AAC: Toward a dynamic equivalence with MMC," *IEEE Trans. Power Del.*, pp. 1–1, 2021.
- [49] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "Hybrid LCC-AAC HVDC transmission system," *Elect. Power Syst. Res.*, vol. 192, p. 106910, Mar. 2021.
- [50] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "Hybrid multiterminal HVDC system based on line-commutated and alternate arm converters," *IEEE Trans. Power Del.*, pp. 1–1, 2021.
- [51] H. R. Wickramasinghe, P. Sun, and G. Konstantinou, "Interoperability of modular multilevel and alternate arm converters in hybrid HVDC systems," *Energies*, vol. 14, no. 5, 2021.
- [52] P. Sun, F. Arrano-Vargas, H. R. Wickramasinghe, and G. Konstantinou, "Benchmark models for HVDC systems and DC-grid studies," in *ICPES 2019, Perth, WA*, Dec. 2019.
- [53] P. Sun, H. R. Wickramasinghe, and G. Konstantinou, "An LCC-AAC hybrid highvoltage DC transmission system," in 2020 IEEE 9th International Power Electronics and Motion Control Conference (IPEMC2020-ECCE Asia), Nov. 2020.

- [54] H. R. Wickramasinghe, P. Sun, and G. Konstantinou, "A hybrid VSC-HVDC system based on modular multilevel converter and alternate arm converter," in *IECON* 2020 The 46th Annual Conference of the IEEE Industrial Electronics Society, 2020, pp. 4141–4146.
- [55] G. P. Adam, S. J. Finney, B. W. Williams, D. R. Trainer, C. D. M. Oates, and D. R. Critchley, "Network fault tolerant voltage-source-converters for high-voltage applications," in 9th IET Int. Conf. on AC and DC Power Transmission (ACDC 2010), 2010, pp. 1–5.
- [56] J. Liu, D. Dong, and D. Zhang, "A hybrid modular multilevel converter family with higher power density and efficiency," *IEEE Trans. Power Electron.*, vol. 36, no. 8, pp. 9001–9014, Aug. 2021.
- [57] H. Pang and X. Wei, "Research on key technology and equipment for zhangbei 500kv DC grid," in 2018 Int. Power Electron. Conf., May 2018, pp. 2343–2351.
- [58] Z. Zhao and M. Iravani, "Application of GTO voltage source inverter in a hybrid HVDC link," *IEEE Trans. on Power Del.*, vol. 9, no. 1, pp. 369–377, Jan. 1994.
- [59] Y. Iwatta, S. Tanaka, K. Sakamoto, H. Konishi, and H. Kawazoe, "Simulation study of a hybrid hvdc system composed of a self-commutated converter and a linecommutated converter," in *Sixth Int. Conf. on AC and DC Power Transmission*, Apr. 1996, pp. 381–386.
- [60] J. Hongbo and A. Ekstrom, "Multiterminal HVDC systems in urban areas of large cities," *IEEE Trans. Power Del.*, vol. 13, no. 4, pp. 1278–1284, 1998.
- [61] N. Nosaka, Y. Tsubota, K. Matsukawa, K. Sakamoto, H. Nakamura, M. Takasaki, and H. Kawazoe, "Simulation studies on a control and protection scheme for hybrid multi-terminal HVDC systems," in *IEEE Power Engineering Society*. 1999 Winter Meeting (Cat. No. 99CH36233), vol. 2. IEEE, Conference Proceedings, pp. 1079– 1084.
- [62] G. Li, G. Li, H. Liang, and M. Yin, "Research on hybrid HVDC," in 2004 Int. Conf. on Power System Technol., 2004. PowerCon 2004., vol. 2, Conference Proceedings, pp. 1607–1612 Vol.2.
- [63] R. E. Torres-Olguin, M. Molinas, and T. Undeland, "Offshore wind farm grid integration by VSC technology with LCC-based HVDC transmission," *IEEE Trans. Sustain. Energy*, vol. 3, no. 4, pp. 899–907, 2012.
- [64] C. Guo, W. Liu, and C. J. S. C. T. S. Zhao, "Research on the control method for voltage-current source hybrid-HVDC system," vol. 56, no. 11, pp. 2771–2777, 2013.

- [65] W. Liu, C. Zhao, and C. Guo, "The control strategy for hybrid HVDC using voltage margin control and voltage dependent current order limiter control," in 2nd IET Renewable Power Generation Conf. (RPG 2013), Conference Proceedings, pp. 1–4.
- [66] R. E. Torres-Olguin, A. Garces, M. Molinas, and T. Undeland, "Integration of offshore wind farm using a hybrid HVDC transmission composed by the PWM currentsource converter and line-commutated converter," *IEEE Trans. Energy Convers.*, vol. 28, no. 1, pp. 125–134, 2013.
- [67] O. Kotb and V. K. Sood, "A hybrid HVDC transmission system supplying a passive load," in 2010 IEEE Electr. Power & Energy Conf. IEEE, 2010, pp. 1–5.
- [68] G. Li, G. Li, H. Liang, M. Yin, and C. Zhao, "Operational mechanism and characteristic analysis of novel hybrid HVDC system," in 2006 Int. Conf. on Power System Technol., Conference Proceedings, pp. 1–6.
- [69] J. Kjærgaard, K. Sogaard, S. Mikkelsen, T. Pande-Rolfson, A. Strandem, B. Bergdahl *et al.*, "Bipolar operation of an HVDC VSC converter with an LCC converter," in *CIGRE Colloquium on HVDC and Power Electron. Systems, San Fansisco, CA, USA*, 2012.
- [70] P. Manohar, V. Kelamane, D. Kaushik, and W. Ahmed, "Improved controls for LCC-VSC hybrid HVDC system," in 2013 Int. conf. on Circuits, Controls and Commun. (CCUBE). IEEE, 2013, pp. 1–5.
- [71] R. E. Torres-Olguin, M. Molinas, and T. M. Undeland, "A model-based controller in rotating reference frame for hybrid HVDC," in 2010 IEEE Energy Convers. Congr. and Expo., Conference Proceedings, pp. 1578–1584.
- [72] R. E. Torres-Olguin, M. Molinas, and T. M. Undeland, "A controller in d-q synchronous reference frame for hybrid HVDC transmission system," in *The 2010 Int. Power Electron. Conf. - ECCE ASIA -*, Conference Proceedings, pp. 376–383.
- [73] R. E. Torres-Olguin, M. Molinas, and T. M. Undeland, "A direct power control for hybrid HVDC transmission systems," in 2011 IEEE Trondheim PowerTech, Conference Proceedings, pp. 1–6.
- [74] C. Guo, W. Liu, and C. Zhao, "Research on the control method for voltage-current source hybrid-HVDC system," *Science China Technological Sciences*, vol. 56, no. 11, pp. 2771–2777, 2013.
- [75] X. Fuan and S. Cheng, "Performance analysis of a hybrid multi-terminal HVDC system," in 2005 Int. Conf. on Electr. Mach. and Systems, vol. 3. IEEE, 2005, pp. 2169–2174.

- [76] X. Chen, H. Sun, J. Wen, W.-J. Lee, X. Yuan, N. Li, and L. Yao, "Integrating wind farm to the grid using hybrid multiterminal HVDC technology," *IEEE Trans. Ind. Appl.*, vol. 47, no. 2, pp. 965–972, 2010.
- [77] J. Xu, C. Zhao, T. Li, J. Xu, H. Pang, and C. Lin, "The hybrid HVDC transmission using line commutated converter and full bridge modular multilevel converter," in 2nd IET Renewable Power Generation Conf. (RPG 2013), Sep. 2013, pp. 1–4.
- [78] L. Yu, C. Guo, C. Zhao, J. Xu, N. An, and X. Hu, "Power reversal of hybrid HVDC system," in 11th IET Int. Conf. on AC and DC Power Transmission, Feb. 2015, pp. 1–6.
- [79] X. Feng and X. Zheng, "Hybrid HVDC system based on LCC and FHMMC," *High Voltage Engineering*, vol. 40, no. 8, pp. 2520–2530, 2014.
- [80] Y. Lee, S. Cui, S. Kim, and S. Sul, "Control of hybrid HVDC transmission system with LCC and FB-MMC," in 2014 IEEE Energy Convers. Congr. and Expo. (ECCE), Sep. 2014, pp. 475–482.
- [81] R. Zeng, L. Xu, L. Yao, S. J. Finney, and Y. Wang, "Hybrid HVDC for integrating wind farms with special consideration on commutation failure," *IEEE Trans. Power Del.*, vol. 31, no. 2, pp. 789–797, Apr. 2016.
- [82] C. Guo, Z. Yin, C. Zhao, and R. Iravani, "Small-signal dynamics of hybrid LCC-VSC HVDC systems," *Int. J.l of Elect. Power & Energy Syst.*, vol. 98, pp. 362–372, 2018.
- [83] C. Guo, W. Liu, C. Zhao, and X. Ni, "Small-signal dynamics and control parameters optimization of hybrid multi-infeed HVDC system," *Int. J.l of Elect. Power & Energy Syst.*, vol. 98, pp. 409 – 418, 2018.
- [84] L. Chen, G. Cheng, C. Shao, Y. Kang, X. Pan, Z. Zhang, and Z. Xu, "Research on control strategy for a 3-terminal LCC-MMC HVDC transmission system," *Gaoya Dianqi/High Voltage Apparatus*, vol. 54, pp. 146–152, 2018.
- [85] L. Xiao, Y. Li, H. Xiao, Z. Zhang, and Z. Xu, "Electromechanical transient modeling of line commutated converter-modular multilevel converter-based hybrid multiterminal high voltage direct current transmission systems," *Energies*, vol. 11, no. 8, p. 2102, 2018.
- [86] D.A.N. Jacobson, P. Wang, S. Howell, C. Karawita, H. Suriyaarachchi, M. Mohaddes, "Hybrid multi-terminal HVDC–LCC with VSC converter taps: A manitoba case study," in *CIGRE*, 2018.

- [87] S. X. Z. Z. H. Rou, Y. Zhou, "Research and development of ultra-high-voltage VSC for the multi-terminal hybrid ±800kV HVDC project in china southern power grid," *CIGRE*, 2018.
- [88] G. Tang and Z. Xu, "A LCC and MMC hybrid HVDC topology with DC line fault clearance capability," *Int. J.l of Elect. Power & Energy Syst.*, vol. 62, pp. 419–428, Nov. 2014.
- [89] G. Tang, Z. Xu, and Y. Zhou, "Impacts of three MMC-HVDC configurations on AC system stability under DC line faults," *IEEE Trans. Power Syst.*, vol. 29, no. 6, pp. 3030–3040, Nov. 2014.
- [90] Yong Lin, Zheng Xu, Liang Xiao, Zheren Zhang, and Huangqing Xiao, "Analysis of coupling effect on LCC-MMC hybrid hvdc from parallel AC lines in close proximity," in 2015 IEEE Power Energy Society General Meeting, Jul. 2015, pp. 1–5.
- [91] Z. Zhang, Z. Xu, Y. Xue, and G. Tang, "DC-side harmonic currents calculation and DC-loop resonance analysis for an LCC-MMC hybrid HVDC transmission system," *IEEE Trans. Power Del.*, vol. 30, no. 2, pp. 642–651, Apr. 2015.
- [92] M. Cheah-Mane, J. Song, R. Ferrer-San-Jose, E. Prieto-Araujo, and O. Gomis-Bellmunt, "Analysis of hybrid LCC-VSC HVDC transmission system configurations," in 15th IET Int. Conf. on AC and DC Power Transmission (ACDC 2019), Feb. 2019, pp. 1–6.
- [93] G. Li, W. Liu, T. Joseph, J. Liang, T. An, J. Lu, M. Szechtman, B. Andersen, and Q. Zhuang, "Control strategies of full-voltage to half-voltage operation for LCC and hybrid LCC/MMC based UHVDC systems," *Energies*, vol. 12, no. 4, p. 742, 2019.
- [94] S. Gao, H. Zhu, B. Zhang, and G. Song, "Modeling and simulation analysis of hybrid bipolar HVDC system based on LCC-HVDC and VSC-HVDC," in 2018 IEEE 3rd Advanced Inform. Technol., Electron. and Autom. Control Conf. (IAEAC), Oct. 2018, pp. 1448–1452.
- [95] ABB, "ABB sets world record in HVDC Light voltage level," ABB, Zurich, Switzerland, Press release, Jan. 2015.
- [96] H. Kai, P. Junchurr, and W. Zhimin, "Research on the interaction between luxi back-to-back and yongfu HVDC," in 2018 China Int. Conf. on Elect. Distribution, Sep. 2018, pp. 1127–1131.
- [97] Z. Li, Z. Wei, R. Zhan, Y. Li, and X.-P. Zhang, "System operational dispatching and scheduling strategy for hybrid cascaded multi-terminal HVDC," *Int. J. of Elect. Power* & Energy Syst., vol. 122, p. 106195, 2020.

- [98] Z. Xu, S. Wang, and H. Xiao, "Hybrid high-voltage direct current topology with line commutated converter and modular multilevel converter in series connection suitable for bulk power overhead line transmission," *IET Power Electron.*, vol. 9, no. 12, pp. 2307–2317, 2016.
- [99] J. Pou, S. Ceballos, G. Konstantinou, G. J. Capella, and V. G. Agelidis, "Control strategy to balance operation of parallel connected legs of modular multilevel converters," in 2013 IEEE Int. Symp. on Ind. Electron., May 2013, pp. 1–7.
- [100] H. Ying, H. Weihuang, L. Ming, and L. Tao, "Steady-state control strategy of multiterminal hybrid UHVDC," in 2017 19th European Conference on Power Electronics and Applications (EPE'17 ECCE Europe), 2017, pp. pp. 1–10.
- [101] W. Cigré, "B4. 57."Guide for the Development of Models for HVDC Converters in a HVDC Grid"," *CIGRE Tech. Brochure*, vol. 604, 2014.
- [102] Plexim GmbH, "Plecs User Manual Version 4.5," Zurich, Switzerland, 2020.
- [103] MATLAB-Simulink version R2020a, R2021a. [Online]. Available: https://www. mathworks.com
- [104] R. Gou, "Research on 1100kV/5500A ultra-high voltage thyristor valve key technology and its application," *IEEE Trans. Power Electron.*, pp. 1–1, Feb. 2019.
- [105] T. Yoshino, "History and trends of converter technology for DC and AC transmission in japan," in 2014 Int. Power Electron. Conf (IPEC-Hiroshima 2014 - ECCE ASIA), May 2014, pp. 3834–3841.
- [106] M. Szechtman, T. Wess and C.V. Thio, "A benchmark model for HVDC system studies," in Int. Conf. on AC and DC Power Transmission, London, UK, Sep. 1991, pp. 374–378.
- [107] M. Szechtman, T. Wess and C.V. Thio, "First benchmark model for HVDC control studies," *Electra*, no. 135, pp. 56–73, 1991.
- [108] CIGRE Working Group 14.02, "The CIGRE HVDC benchmark model–a new proposal with revised parameters," *Electra*, no. 157, pp. 61–66, Dec. 1994.
- [109] M.M. Babu Narayanan and S. Parameswaran, "Study of control strategies for new CIGRE HVDC benchmark model," in *Session 1998*, 1998, pp. 1–6.
- [110] J. Schwartz, E. Rahimi, and H. Zareipour, "A modified CIGRE HVDC benchmark model for 60 hz applications," in *2014 North Amer. Power Symp. (NAPS)*. Pullman, WA, USA: IEEE, Sep. 2014, pp. 1–6.

- [111] P. Lehn, J. Rittiger, and B. Kulicke, "Comparison of the ATP version of the EMTP and the NETOMAC program for simulation of HVDC systems," *IEEE Trans. Power Del.*, vol. 10, no. 4, pp. 2048–2053, Oct. 1995.
- [112] CIGRE Working Group B4.34, "Capacitor commutated converters (CCC) HVDC interconnections - digital modeling and benchmark circuit," CIGRE, Tech. Brochure 352, Jun. 2008.
- [113] K. Rudion, A. Orths, P. B. Eriksen, and Z. A. Styczynski, "Toward a benchmark test system for the offshore grid in the north sea," in *IEEE PES General Meeting*. Providence, RI, USA: IEEE, Jul. 2010, pp. 1–8.
- [114] J. Peralta, H. Saad, S. Dennetiere, J. Mahseredjian, and S. Nguefeu, "Detailed and averaged models for a 401-level MMC–HVDC system," *IEEE Trans. Power Del.*, vol. 27, no. 3, pp. 1501–1508, Jul. 2012.
- [115] T. K. Vrana, Y. Yang, D. Jovcic, S. Dennetière, J. Jardini, and H. Saad, "The CIGRE B4 DC grid test system," *Electra*, no. 270, pp. 10–19, Oct. 2013.
- [116] S. Dennetiere, B. Bruned, H. Saad, and E. Lemieux, "Task separation for real-time simulation of the cigre DC grid benchmark," in 2018 Power Syst. Comput. Conf. (PSCC), Jun. 2018, pp. 1–7.
- [117] T. An *et al.*, "A DC grid benchmark model for studies of interconnection of power systems," *CSEE J. Power and Energy Syst.*, vol. 1, no. 4, pp. 101–109, Dec. 2015.
- [118] H. R. Wickramasinghe, G. Konstantinou, Z. Li, and J. Pou, "Alternate arm converters-based HVDC model compatible with the CIGRE B4 DC grid test system," *IEEE Trans. Power Del.*, vol. 34, no. 1, pp. 149–159, Feb. 2019.
- [119] P. Rault *et al.*, "Final recommendations for interoperability of multivendor HVDC systems," Best Paths, Tech. Rep. 9.3, Dec. 2018.
- [120] G. N. Bathurst, B. C. Smith, N. R. Watson, and J. Arrillaga, "Modelling of HVDC transmission systems in the harmonic domain," *IEEE Trans. Power Del.*, vol. 14, no. 3, pp. 1075–1080, Jul. 1999.
- [121] A. Daneshpooy, A. M. Gole, D. G. Chapman, and J. B. Davies, "Fuzzy logic control for HVDC transmission," *IEEE Trans. Power Del.*, vol. 12, no. 4, pp. 1690–1697, Oct. 1997.
- [122] M. O. Faruque, Yuyan Zhang, and V. Dinavahi, "Detailed modeling of CIGRE HVDC benchmark system using PSCAD/EMTDC and PSB/SIMULINK," *IEEE Trans. Power Del.*, vol. 21, no. 1, pp. 378–387, Jan. 2006.
- [123] Y. Li, L. Luo, Kazuo Nakamura, C. Rehtanz, J. Zhang, J. Xu, and F. Liu, "Transient response characteristics of new HVDC transmission system based on new converter transformer," in 2008 Third Int. Conf. on Elect. Utility Deregulation and Restructuring and Power Technol., Apr. 2008, pp. 1873–1877.
- [124] V. K. Sood, V. Khatri, and H. Jin, "EMTP modelling of CIGRE benchmark based HVDC transmission system operating with weak AC systems," in *Proc. of Int. Conf. on Power Electron., Drives and Energy Syst. for Ind. Growth*, vol. 1, Jan. 1996, pp. 426–432 vol.1.
- [125] S. Todd, A. R. Wood, and P. S. Bodger, "An s-domain model of an HVDC converter," *IEEE Trans. Power Del.*, vol. 12, no. 4, pp. 1723–1729, Oct. 1997.
- [126] X. Yang and C. Chen, "HVDC dynamic modelling for small signal analysis," *IEE Proc. Generation, Transmission and Distribution*, vol. 151, no. 6, pp. 740–746, 2004.
- [127] X. Zhang, J. Bai, G. Cao, and C. Chen, "Optimizing HVDC control parameters in multi-infeed HVDC system based on electromagnetic transient analysis," *Int. J. of Elect. Power & Energy Syst.*, vol. 49, pp. 449–454, Mar. 2013.
- [128] W. Zhao et al., "HVDC transmission engineering technology," 2004.
- [129] A. Alassi, S. Bañales, O. Ellabban, G. Adam, and C. MacIver, "HVDC transmission: Technology review, market trends and future outlook," *Renewable and Sustainable Energy Reviews*, vol. 112, pp. 530 – 554, 2019.
- [130] G. Li, C. Li, and D. Van Hertem, "HVDC technology overview," pp. 45–78, Feb. 2016.
- [131] Y. Xue and X.-P. Zhang, "Reactive power and AC voltage control of LCC HVDC system with controllable capacitors," *IEEE Trans. Power Syst.*, vol. 32, no. 1, pp. 753–764, Jan. 2017.
- [132] R. Blasco-Gimenez, N. Aparicio, S. Ano-Villalba, and S. Bernal-Perez, "LCC-HVDC connection of offshore wind farms with reduced filter banks," *IEEE Trans. Ind. Electron.*, vol. 60, no. 6, pp. 2372–2380, Jun. 2013.
- [133] Y. Xue, X.-P. Zhang, and C. Yang, "Elimination of commutation failures of LCC HVDC system with controllable capacitors," *IEEE Trans. Power Syst.*, vol. 31, no. 4, pp. 3289–3299, Jul. 2016.
- [134] Y. Xue, X.-P. Zhang, and C. Yang, "Commutation failure elimination of LCC HVDC systems using thyristor-based controllable capacitors," *IEEE Trans. Power Del.*, vol. 33, no. 3, pp. 1448–1458, Jun. 2018.

- [135] R. Li, S. Bozhko, and G. Asher, "Frequency control design for offshore wind farm grid with LCC-HVDC link connection," *IEEE Trans. Power Electron.*, vol. 23, no. 3, pp. 1085–1092, May 2008.
- [136] M. Daryabak, S. Filizadeh, J. Jatskevich, A. Davoudi, M. Saeedifard, V. K. Sood, J. A. Martinez, D. Aliprantis, J. Cano, and A. Mehrizi-Sani, "Modeling of LCC-HVDC systems using dynamic phasors," *IEEE Trans. Power Del.*, vol. 29, no. 4, pp. 1989– 1998, Aug. 2014.
- [137] O. E. Oni, I. E. Davidson, and K. N. I. Mbangula, "A review of LCC-HVDC and VSC-HVDC technologies and applications," in 2016 IEEE 16th Int. Conf. on Environment and Elect. Eng. (EEEIC), Jun. 2016, pp. 1–7.
- [138] M. Andersson, X. Yang, and Y. J. Häfner, "A cost effective hybrid HVDC transmission system with high performance in DC line fault handling," 2018.
- [139] X. Shi, Z. Wang, L. M. Tolbert, and F. Wang, "A comparison of phase disposition and phase shift PWM strategies for modular multilevel converters," in 2013 IEEE Energy Convers. Congr. and Expo., Sep. 2013, pp. 4089–4096.
- [140] B. Li, R. Yang, D. Xu, G. Wang, W. Wang, and D. Xu, "Analysis of the phase-shifted carrier modulation for modular multilevel converters," *IEEE Trans. Power Electron.*, vol. 30, no. 1, pp. 297–310, Jan. 2015.
- [141] H. Peng, R. Xie, K. Wang, Y. Deng, X. He, and R. Zhao, "A capacitor voltage balancing method with fundamental sorting frequency for modular multilevel converters under staircase modulation," *IEEE Trans. Power Electron.*, vol. 31, no. 11, pp. 7809– 7822, Nov. 2016.
- [142] M. S. A. Dahidah, G. Konstantinou, and V. G. Agelidis, "A review of multilevel selective harmonic elimination PWM: Formulations, solving algorithms, implementation and applications," *IEEE Trans. Power Electron.*, vol. 30, no. 8, pp. 4091–4106, Aug. 2015.
- [143] R. Darus, J. Pou, G. Konstantinou, S. Ceballos, R. Picas, and V. G. Agelidis, "A modified voltage balancing algorithm for the modular multilevel converter: Evaluation for staircase and phase-disposition PWM," *IEEE Trans. Power Electron.*, vol. 30, no. 8, pp. 4119–4127, Aug. 2015.
- [144] M. Saeedifard and R. Iravani, "Dynamic performance of a modular multilevel back-to-back HVDC system," *IEEE Trans. Power Del.*, vol. 25, no. 4, pp. 2903–2912, Oct. 2010.

- [145] H. R. Wickramasinghe, G. Konstantinou, J. Pou, and V. G. Agelidis, "Interactions between indirect DC-voltage estimation and circulating current controllers of MMCbased HVDC transmission systems," *IEEE Trans. Power Syst.*, vol. 33, no. 1, pp. 829–838, Jan. 2018.
- [146] Q. Tu, Z. Xu, and L. Xu, "Reduced switching-frequency modulation and circulating current suppression for modular multilevel converters," *IEEE Trans. Power Del.*, vol. 26, no. 3, pp. 2009–2017, Jul. 2011.
- [147] J. Pou, S. Ceballos, G. Konstantinou, V. G. Agelidis, R. Picas, and J. Zaragoza, "Circulating current injection methods based on instantaneous information for the modular multilevel converter," *IEEE Trans. Ind. Electron.*, vol. 62, no. 2, pp. 777– 788, Feb. 2015.
- [148] Y. Tian, H. R. Wickramasinghe, Z. Li, and G. Konstantinou, "Modular multilevel converter sub-modules for HVDC applications," in 2020 IEEE 9th Int. Power Electron. and Motion Control Conf. (IPEMC2020-ECCE Asia), 2020, pp. 1510–1515.
- [149] G. S. Konstantinou and V. G. Agelidis, "Performance evaluation of half-bridge cascaded multilevel converters operated with multicarrier sinusoidal PWM techniques," in 2009 4th IEEE Conf. on Ind. Electron. and Appl., May 2009, pp. 3399– 3404.
- [150] J. Qin and M. Saeedifard, "Reduced switching-frequency voltage-balancing strategies for modular multilevel HVDC converters," *IEEE Trans. Power Del.*, vol. 28, no. 4, pp. 2403–2410, Oct. 2013.
- [151] M. Guan, Z. Xu, and H. Chen, "Control and modulation strategies for modular multilevel converter based HVDC system," in *IECON 2011 - 37th Annual Conf. of the IEEE Ind. Electron. Soc.*, Nov. 2011, pp. 849–854.
- [152] F. Therrien, "Numerically efficient modeling of saturable AC machines for power systems electromagnetic transients simulation programs," Ph.D. dissertation, University of British Columbia, 2015.
- [153] J. Mahseredjian, V. Dinavahi, and J. A. Martinez, "Simulation tools for electromagnetic transients in power systems: Overview and challenges," *IEEE Trans. Power Del.*, vol. 24, no. 3, pp. 1657–1669, Jul. 2009.
- [154] U. N. Gnanarathna, A. M. Gole, and R. P. Jayasinghe, "Efficient modeling of modular multilevel HVDC converters (MMC) on electromagnetic transient simulation programs," *IEEE Trans. Power Del.*, vol. 26, no. 1, pp. 316–324, Jan. 2011.

- [155] S. Cole, J. Beerten, and R. Belmans, "Generalized dynamic VSC MTDC model for power system stability studies," *IEEE Trans. Power Syst.*, vol. 25, no. 3, pp. 1655– 1662, Aug. 2010.
- [156] S. Liu, Z. Xu, W. Hua, G. Tang, and Y. Xue, "Electromechanical transient modeling of modular multilevel converter based multi-terminal HVDC systems," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 72–83, Jan. 2014.
- [157] S. R. Sanders, J. M. Noworolski, X. Z. Liu, and G. C. Verghese, "Generalized averaging method for power conversion circuits," *IEEE Trans. Power Electron.*, vol. 6, no. 2, pp. 251–259, Apr. 1991.
- [158] L. Wang, X. Deng, P. Han, X. Qi, X. Wu, M. Li, and H. Xu, "Electromagnetic transient modeling and simulation of power converters based on a piecewise generalized state space averaging method," *IEEE Access*, vol. 7, pp. 12241–12251, Jan. 2019.
- [159] A. M. Stankovic, B. C. Lesieutre, and T. Aydin, "Modeling and analysis of singlephase induction machines with dynamic phasors," *IEEE Trans. Power Syst.*, vol. 14, no. 1, pp. 9–14, 1999.
- [160] S. R. Deore, P. B. Darji, and A. M. Kulkarni, "Dynamic phasor modeling of modular multi-level converters," in 2012 IEEE 7th Int. Conf. on Ind. and Inform. Syst. (ICIIS), Aug. 2012, pp. 1–6.
- [161] K. Strunz, R. Shintaku, and F. Gao, "Frequency-adaptive network modeling for integrative simulation of natural and envelope waveforms in power systems and circuits," *IEEE Trans. Circuits and Syst.I*, vol. 53, no. 12, pp. 2788–2803, Dec. 2006.
- [162] P. Zhang, J. R. Marti, and H. W. Dommel, "Synchronous machine modeling based on shifted frequency analysis," *IEEE Trans. Power Syst.*, vol. 22, no. 3, pp. 1139– 1147, Aug. 2007.
- [163] S. Fan and H. Ding, "Time domain transformation method for accelerating EMTP simulation of power system dynamics," *IEEE Trans. Power Syst.*, vol. 27, no. 4, pp. 1778–1787, Nov. 2012.
- [164] D. Shu, V. Dinavahi, X. Xie, and Q. Jiang, "Shifted frequency modeling of hybrid modular multilevel converters for simulation of MTDC grid," *IEEE Trans. Power Del.*, vol. 33, no. 3, pp. 1288–1298, Jun. 2018.
- [165] Y. Wei, D. Shu, X. Xie, V. Dinavahi, and Z. Yan, "Real-time simulation of hybrid modular multilevel converters using shifted phasor models," *IEEE Access*, vol. 7, pp. 2376–2386, Dec. 2019.

- [166] S. Gao, Y. Chen, Y. Song, Y. Zhang, Z. Tan, and S. Huang, "Shifted-frequency based electromagnetic transient simulation for power system dynamics using two-stage singly diagonally implicit runge–kutta method," *Energy Reports*, vol. 6, pp. 701– 708, Dec. 2020, the 7th International Conference on Power and Energy Systems Engineering.
- [167] J. M. Zavahir, J. Arrillaga, and N. R. Watson, "Hybrid electromagnetic transient simulation with the state variable representation of HVDC converter plant," *IEEE Trans. Power Del.*, vol. 8, no. 3, pp. 1591–1598, Jul. 1993.
- [168] V. Jalili-Marandi, V. Dinavahi, K. Strunz, J. A. Martinez, and A. Ramirez, "Interfacing techniques for transient stability and electromagnetic transient programs IEEE task force on interfacing techniques for simulation tools," *IEEE Trans. Power Del.*, vol. 24, no. 4, pp. 2385–2395, 2009.
- [169] K. Mudunkotuwa and S. Filizadeh, "Co-simulation of electrical networks by interfacing EMT and dynamic-phasor simulators," *Elect. Power Syst. Res.*, vol. 163, pp. 423–429, Oct. 2018.
- [170] D. Shu, Y. Wei, V. Dinavahi, K. Wang, Z. Yan, and X. Li, "Cosimulation of shiftedfrequency/dynamic phasor and electromagnetic transient models of hybrid LCC-MMC DC grids on integrated CPU–GPUs," *IEEE Trans. Ind. Electron.*, vol. 67, no. 8, pp. 6517–6530, Aug. 2020.
- [171] D. Shu, X. Xie, Z. Yan, V. Dinavahi, and K. Strunz, "A multi-domain co-simulation method for comprehensive shifted-frequency phasor DC-grid models and EMT ACgrid models," *IEEE Trans. Power Electron.*, vol. 34, no. 11, pp. 10557–10574, 2019.
- [172] Manitoba HVDC Research Centre, "USER'S GUIDE on the use of PSCAD," Canada, 2018.
- [173] RTDS Technologies Inc., "RTDS simulator software," Winnipeg, MB, Canada.
- [174] OPAL-RT TECHNOLOGIES, "Hypersim Brochure," Montreal, QC, Canada, 2020.
- [175] The MathWorks, Inc., "Simulink 7–User's Guide," Natick, MA, USA, 2009.
- [176] OPAL-RT TECHNOLOGIES, "RT-LAB," Montreal, QC, Canada.
- [177] H. W. Dommel, "Digital computer solution of electromagnetic transients in singleand multiphase networks," *IEEE Trans. Power App. and Syst.*, vol. PAS-88, no. 4, pp. 388–399, 1969.
- [178] C. Dufour, J. Mahseredjian, and J. Bélanger, "A combined state-space nodal method for the simulation of power system transients," *IEEE Trans. Power Del.*, vol. 26, no. 2, pp. 928–935, 2011.

- [179] O. Wasynczuk and S. Sudhoff, "Automated state model generation algorithm for power circuits and systems," *IEEE Trans. Power Syst.*, vol. 11, no. 4, pp. 1951–1956, 1996.
- [180] P. J. Davis and P. Rabinowitz, "Methods of numerical integration," 2007.
- [181] K. Strunz and E. Carlson, "Nested fast and simultaneous solution for time-domain simulation of integrative power-electric and electronic systems," *IEEE Trans. Power Del.*, vol. 22, no. 1, pp. 277–287, Jan. 2007.
- [182] H. Saad, J. Peralta, S. Dennetière, J. Mahseredjian, J. Jatskevich, J. A. Martinez, A. Davoudi, M. Saeedifard, V. Sood, X. Wang, J. Cano, and A. Mehrizi-Sani, "Dynamic averaged and simplified models for MMC-based HVDC transmission systems," *IEEE Trans. Power Del.*, vol. 28, no. 3, pp. 1723–1730, Apr. 2013.
- [183] H. Saad, S. Dennetière, J. Mahseredjian, P. Delarue, X. Guillaud, J. Peralta, and S. Nguefeu, "Modular multilevel converter models for electromagnetic transients," *IEEE Trans. Power Del.*, vol. 29, no. 3, pp. 1481–1489, Jun. 2014.
- [184] J. Xu, C. Zhao, W. Liu, and C. Guo, "Accelerated model of modular multilevel converters in PSCAD/EMTDC," *IEEE Trans. Power Del.*, vol. 28, no. 1, pp. 129–136, Jan. 2013.
- [185] H. Zhang, D. Jovcic, W. Lin, and A. J. Far, "Average value MMC model with accurate blocked state and cell charging/discharging dynamics," in 2016 4th Int. Symp. on Environmental Friendly Energies and Appl. (EFEA), Nov. 2016, pp. 1–6.
- [186] L. Zhao, G. Xu, C. Hong, G. Huang, D. Shi, T. Zhou, H. Zhen, X. Wu, and C. Jing, "Fast simulation of MMC-HVDC systems using a generalized equivalent circuit model of MMC," in 2020 IEEE 16th Int. Conf. on Control Autom. (ICCA), Nov. 2020, pp. 1605–1611.
- [187] S. Wang, X. Zhao, F. Xue, W. Li, H. Peng, D. Shi, S. Wang, and Z. Wang, "Electromechanical transient modeling of modular multilevel converter based HVDC network," in 2019 IEEE Sustain. Power and Energy Conf. (iSPEC), Nov. 2019, pp. 1845–1850.
- [188] J. Beerten, S. Cole, and R. Belmans, "Generalized steady-state VSC MTDC model for sequential AC/DC power flow algorithms," *IEEE Trans. Power Syst.*, vol. 27, no. 2, pp. 821–829, 2012.
- [189] J. Dou, B. Zhang, R. Chai, Z. Guan, and C. Liu, "Unified iterative power flow algorithm of AC/DC networks incorporating hybrid HVDC," in 2016 IEEE 16th Int. Conf. on Environment and Elect. Eng. (EEEIC), 2016, pp. 1–6.

- [190] S.Yao *et al.*, "A fast electromagnetic transient simulation of large-scale power system," *Elect. Power Construction*, vol. 36, no. 12, pp. 16–21, Dec. 2015.
- [191] L. Xinyang, Y. Weiyang, J. Hongjie, and W. Ling, "Engineering application architecture study on an LCC parallel MTDC transmission system," *The J. of Eng.*, vol. 2019, no. 16, pp. 3276–3279, 2019.
- [192] T. M. Haileselassie and K. Uhlen, "Impact of DC line voltage drops on power flow of MTDC using droop control," *IEEE Trans. Power Syst.*, vol. 27, no. 3, pp. 1441–1449, 2012.
- [193] T. Nakajima and S. Irokawa, "A control system for HVDC transmission by voltage sourced converters," in 1999 IEEE Power Eng. Soc. Summer Meeting. Conf. Proc. (Cat. No.99CH36364), vol. 2, 1999, pp. 1113–1119 vol.2.
- [194] K. Rouzbehi, A. Miranian, J. I. Candela, A. Luna, and P. Rodriguez, "A generalized voltage droop strategy for control of multiterminal DC grids," *IEEE Trans. Ind. Appl.*, vol. 51, no. 1, pp. 607–618, 2015.
- [195] W. Wang and M. Barnes, "Power flow algorithms for multi-terminal VSC-HVDC with droop control," *IEEE Trans. Power Syst.*, vol. 29, no. 4, pp. 1721–1730, 2014.
- [196] C. Angeles-Camacho, O. Tortelli, E. Acha, and C. Fuerte-Esquivel, "Inclusion of a high voltage DC-voltage source converter model in a Newton-Raphson power flow algorithm," *IEE Proc.-Generation, Transmission and Distribution*, vol. 150, no. 6, pp. 691–696, 2003.
- [197] G. Konstantinou, J. Pou, S. Ceballos, R. Darus, and V. G. Agelidis, "Switching frequency analysis of staircase-modulated modular multilevel converters and equivalent PWM techniques," *IEEE Trans. Power Del.*, vol. 31, no. 1, pp. 28–36, Feb. 2016.
- [198] Y. Liu and Y. Zhu, "The harmonic characteristics of HVDC system and reduction," in 2013 Third Int. Conf. on Intell. Sys. Design and Eng. Appl., Jan. 2013, pp. 1452– 1455.
- [199] J. Liao, N. Zhou, and Q. Wang, "DC-side harmonic analysis and DC filter design in hybrid HVDC transmission systems," *Int. J. of Elect. Power & Energy Syst.*, vol. 113, pp. 861–873, 2019.
- [200] M. M. C. Merlin and T. C. Green, "Cell capacitor sizing in multilevel converters: cases of the modular multilevel converter and alternate arm converter," *IET Power Electron.*, vol. 8, no. 3, pp. 350–360, Mar. 2015.

- [201] A. Kumar, S. Jhampati, and R. Suri, "HVDC converter stations design for LCC based HVDC transmission system-key consideration," in 2017 14th IEEE India Council Int. Conf. (INDICON). IEEE, 2017, pp. 1–6.
- [202] "IEEE recommended practice and requirements for harmonic control in electric power syst." *IEEE Std 519-2014 (Revision of IEEE Std 519-1992)*, pp. 1–29, Jun. 2014.
- [203] C. Guo, Z. Yin, Y. Wang, and C. Zhao, "Small-signal dynamic model of LCC-MMC hybrid HVDC system," Proc. of the Chinese Soc. of Elect. Eng., vol. 38, pp. 4705– 4714, Aug. 2018.
- [204] L. Tang and B.-T. Ooi, "Locating and isolating DC faults in multi-terminal DC systems," *IEEE Trans. Power Del.*, vol. 22, no. 3, pp. 1877–1884, Jul. 2007.
- [205] E. Kontos, R. T. Pinto, S. Rodrigues, and P. Bauer, "Impact of HVDC transmission system topology on multiterminal DC network faults," *IEEE Trans. Power Del.*, vol. 30, no. 2, pp. 844–852, Apr. 2015.